

| Title | Exploring circadian blood pressure patterns |
|----------------------|--|
| Author(s) | Madden, Jamie M. |
| Publication date | 2017 |
| Original citation | Madden, J. 2017. Exploring circadian blood pressure patterns. PhD Thesis, University College Cork. |
| Type of publication | Doctoral thesis |
| Rights | © 2017, Jamie Madden. http://creativecommons.org/licenses/by-nc-nd/3.0/ |
| Embargo information | No embargo required |
| Item downloaded from | http://hdl.handle.net/10468/3888 |

Downloaded on 2017-09-04T23:45:59Z



University College Cork, Ireland Coláiste na hOllscoile Corcaigh

Exploring Circadian Blood Pressure

Patterns

A thesis submitted to the National University of Ireland, Cork for the degree of Doctor of Philosophy in the Department of Epidemiology & Public Health

February 2017

Jamie Madden (B.Eng., MSc)

Head of Department

Prof. Ivan J. Perry

Supervisors

Dr. Anthony P. Fitzgerald

Prof. Patricia M. Kearney

Dr. Xia Li

CONTENTS

| LIST OF TABLES |
|---|
| LIST OF FIGURES |
| LIST OF ABBREVIATIONS10 |
| DECLARATION12 |
| ACKNOWLEDGMENTS |
| THESIS ABSTRACT14 |
| 1. INTRODUCTION16 |
| 1.1 Introduction17 |
| 1.2 Study Design – Motivating Dataset26 |
| 1.3 Aims and Objectives29 |
| 1.4 Thesis Outline |
| 2. LONGITUDINAL DATA ANALYSIS & APPLICATION TO ABPM |
| 2.1 Chapter Overview |
| 2.2 Overview of Longitudinal Data33 |
| 2.3 Statistical Approaches for Modelling Longitudinal Data35 |
| 2.4 Statistical Approaches for Modelling Individual ABPM Profiles45 |
| 3. CORRELATION BETWEEN SHORT-TERM BLOOD PRESSURE VARIABILITY AND |
| LEFT VENTRICULAR MASS INDEX: A META-ANALYSIS (PAPER 1)67 |
| 3.1 Abstract |
| 3.2 Introduction69 |
| 3.3 Methods70 |
| 3.4 Results72 |
| 3.5 Discussion74 |
| 4. SHORT-TERM BLOOD PRESSURE VARIABILITY OVER 24 HOURS AND TARGET |
| ORGAN DAMAGE IN MIDDLE-AGED MEN AND WOMEN (PAPER 2)85 |

| 4.1 | Abstract | 86 |
|---|---|---|
| 4.2 | Introduction | 87 |
| 4.3 | Methods | 88 |
| 4.4 | Results | 91 |
| 4.5 | Discussion | 95 |
| 4.6 | Conclusion | 98 |
| 5. EXP | PLORING DIURNAL VARIATION USING PIECEWISE LINEAR SPLINES: | AN |
| EXAMPL | E USING BLOOD PRESSURE (PAPER 3) | 105 |
| 5.1 | Abstract | 106 |
| 5.2 | Introduction | 107 |
| 5.3 | Methods | 109 |
| 5.4 | Results | 115 |
| 5.5 | Discussion | 118 |
| 5.6 | Conclusion | 121 |
| | | |
| 6. MO | RNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N | AULTIPLE- |
| 6. MO COMPO | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) | /ULTIPLE- 129 |
| 6. MO COMPORE 6.1 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract | ИULTIPLE- 129 130 |
| 6. MO COMPORE 6.1 6.2 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract Introduction | MULTIPLE- 129 130 131 |
| 6. MO COMPORE 6.1 6.2 6.3 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract Introduction Methods | ИULTIPLE- 129 130 131 133 |
| 6. MO COMPOR 6.1 6.2 6.3 6.4 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract Introduction Methods Statistical Analysis | AULTIPLE- 129 130 131 133 134 |
| 6. MO COMPORE 6.1 6.2 6.3 6.4 6.5 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract Introduction Methods Statistical Analysis Results | AULTIPLE- 129 130 131 133 134 138 |
| 6. MO COMPOR 6.1 6.2 6.3 6.4 6.5 6.6 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) Abstract Introduction Methods Statistical Analysis Results Discussion | AULTIPLE- 129 130 131 133 134 138 142 |
| 6. MO COMPORE 6.1 6.2 6.3 6.4 6.5 6.6 7. SHC | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) | AULTIPLE- 129 130 131 133 134 138 142 142 155 |
| 6. MO COMPORE 6.1 6.2 6.3 6.4 6.5 6.6 7. SHC 7.1 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) | AULTIPLE- 129 130 131 133 133 134 138 142 155 156 |
| MO COMPORE 6.1 6.2 6.3 6.4 6.5 6.6 7. SHC 7.1 7.2 | ORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS N NENT COSINOR MODEL (PAPER 4) | AULTIPLE- 129 130 131 133 134 138 142 155 156 157 |

| 7.4 I | Results |
|-------------|--|
| 7.5 I | Discussion161 |
| 8. DISCU | ISSION |
| 8.1 9 | Summary of Main Findings175 |
| 8.2 I | mplications of Findings178 |
| 8.3 | Strengths and Limitations182 |
| 8.4 F | Future Recommendations for Research187 |
| 8.5 (| Conclusion |
| 9. REFER | RENCES: |
| Appendix / | A Supplementary material (Chapter 2)209 |
| Appendix | B Supplementary material (Chapter 3) - search terms and search |
| strategy fo | or systematic review214 |
| Appendix | C Supplementary material (Chapter 4)220 |
| Appendix | D Supplementary material (Chapter 5)227 |
| Appendix | E Supplementary material (Missing Data)229 |
| Appendix | F Supplementary material (Chapter 6)233 |
| Appendix | G Supplementary Material (Chapter 7)242 |
| Appendix | H Sample R Code250 |
| Appendix | Research output, dissemination and training257 |
| Appendix . | J Published Papers (PDFs)260 |

LIST OF TABLES

| Table 1-1 Short-term BPV Summary Indices 22 |
|---|
| Table 2-1 Comparison of different modeling approaches for the analysis of ABPM |
| data66 |
| Table 3-1 Study characteristics 80 |
| Table 3-2 BPV definitions81 |
| Table 3-3 Extracted Results 82 |
| Table 3-4 Quality Assessment |
| Table 4-1 Baseline Characteristics by LVH and Microalbuminuria101 |
| Table 4-2 Variability Indices and Nocturnal Dip by LVH102 |
| Table 4-3 Variability Indices and Nocturnal Dip by Microalbuminuria102 |
| Table 4-4 Association between SBP BPV and LVH103 |
| Table 4-5 Association between SBP BPV and Microalbuminuria104 |
| Table 5-1 Baseline Characteristics 124 |
| Table 5-2 Various models with parameter estimates for slopes at each segment |
| along with corresponding correlations and variances |
| Table 6-1 Baseline Characteristics 147 |
| Table 6-2 Model parameter estimates (SBP) along with corresponding correlations |
| and variances149 |
| Table 6-3 Maximum Morning Surge (mmHg/30mins)151 |
| Table 7-1 BP levels 2010 v 2014165 |
| Table 7-2 Dipping status 2010 v 2014166 |
| Table 7-3 BP categories 2010 v 2014166 |
| Table 7-4 Variability Indices 2010 v 2014167 |
| Table 7-5 Comparison of SBP parameters from two separate two-component |
| random-effects cosinor models (2010 vs 2014)169 |
| Table 7-6 Comparison of SBP parameters from two separate two-component |
| random-effects cosinor models (2010 vs 2014)170 |
| Table 9-1 Association between parameters extracted from polynomial regression |
| model and microalbuminuria (logistic regression) |
| Table 9-2 Baseline Characteristics by anti-hypertensive treatment |
| Table 9-3 Variability Indices and Dip by White-Coat Hypertension221 |

| Table 9-4 Table of demographics comparing full sample to ABPM sample | 231 |
|--|-----|
| Table 9-5 Peer reviewed publication from PhD | 257 |
| Table 9-6 Other research output during PhD | 257 |
| Table 9-7 Conference Presentations during PhD | 258 |
| Table 9-8 Training & workshops attend during PhD | 259 |
| Table 9-9 Placements Completed | 259 |
| Table 9-10 Awards | 259 |
| | |

LIST OF FIGURES

| Figure 1-1 Overview of thesis including aims and objectives |
|--|
| Figure 2-1 Example of four individual ABPM readings |
| Figure 2-2 Illustration of simple linear regression model and simple linear mixed |
| effects model where the data are clustered38 |
| Figure 2-3 Raw data for four individuals along with predicted 6 th order polynomial |
| model47 |
| Figure 2-4 Illustration of cubic and restricted splines50 |
| Figure 2-5 Illustration of piecewise linear splines52 |
| Figure 2-6 A simple cosinor fit to observed SBP readings of an individual |
| (period=24h)54 |
| Figure 2-7 Fourier series – a linear combination of sines and cosines can produce a |
| complex curve [127]56 |
| Figure 2-8 Illustration of the SITAR model for SBP over 24h. The black line indicates |
| the mean BP curve63 |
| Figure 3-1 Flow diagram of study selection78 |
| Figure 3-2 Pooled meta-analysis of z-scores by BPV index79 |
| Figure 4-1 Four study participants and their 24h BP profiles. Panels (a) & (b) |
| illustrate different patterns yet have similar 24h SD but different values based on |
| the other indices. Both (c) & (d) demonstrate a similar issue (values rounded)100 |
| Figure 5-1 Plot of average SBP over 24h which helped identify 6pm and 4am as |
| common knot points for all participants where there was a notable change in |
| trajectory of BP. Also highlighted are the periods where individuals woke and went |
| to sleep. In addition to the two common points, we were able to obtain additional |
| (two) subject-specific knot points at wake and sleep times123 |
| Figure 5-2 Individual SBP readings along with predicted subject-specific trajectories |
| from a linear mixed effects model as a function of time only using two different |
| approaches; polynomials (red line) and piecewise linear splines (blue lines)125 |
| Figure 5-3 Predicted average (95% CI) piecewise linear trajectory of those |
| with/without presence of microalbuminuria adjusted for age, sex and BMI using a |
| linear mixed-effects model (Model 2). Each linear spline represents the rate of SBP |
| increase or decrease (slope) for that segment and has been given a corresponding |

number which is referred to in Table 5-2. For the purposes of this plot we have set Figure 6-1 ABPM readings (circles, thin black line) of four individuals along with predicted subject-specific trajectories from a random-effects model as a function of time using (i) single cosinor (thick black line) (ii) two-component cosinor (thick red Figure 6-2 ABPM readings of three individuals with fitted subject-specific trajectories from a two-component cosinor random-effects model (left panels). Their corresponding rate of change curves (first derivatives) are also plotted on Figure 6-4 FPCA: Each of the first three FPCs as variations about the mean along with the percentage of total variation explained by the component. The solid black line represents the mean SBP over the day and the functions obtained by adding and subtracting \pm SD of the eigenfunctions to the mean. Plus signs indicate addition and minus signs indicate subtraction.152 Figure 6-5 Scatter plots and the corresponding correlations between the twocomponent random-effects cosinor model parameters and the first three FPC

Figure 7-3 Bland-Altman scatter-plots of agreement of measurements of 24h SBP SD and 24h SBP ARV four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines Figure 7-4 Bland-Altman scatter-plots of agreement of measurements of the first and second amplitudes four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines Figure 7-5 Bland-Altman scatter-plots of agreement of measurements of the first and second phase shifts four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines Figure 7-6 Boxplots of differences between each of the cosinor parameters between the two time-points (2010 minus 2014)......172 Figure 7-7 Comparison of fits, 2010 vs 2014, for six individuals using two separate two-component random-effects cosinor models......173 Figure 9-2 Population Level Group Average (95 % CI) Linear Mixed Effects Model (Age, Sex, BMI Adjusted) (Chapter 2).....211 Figure 9-4 VPC (visual predictive check) plot. Black lines represent median of observed data with 90% interguantile range of observations. Red line is predicted mean along with 90% prediction interval......227 Figure 9-5 Predicted average (95% CI) piecewise linear trajectory of those with/without presence of microalbuminuria adjusted for age, sex, BMI and including an interaction of the spline terms with microalbuminuria using a linear mixed-effects model (Model 3). Each linear spline represents the rate of BP increase

Figure 9-6 ABPM readings of three individuals with fitted subject-specific trajectories from a two-component cosinor mixed-effects model (five parameters) and spline model (six parameters - cubic spline with four knots at 18:00, 24:00, 04:00 and 08:00) (left panels). Their corresponding rate of change curves (first derivatives) are also plotted (right panels). The plots indicated an extremely similar pattern thus giving further justification of the two-component cosinor model.....233 Figure 9-8 Scree plot with eigenvalues and the eight principle components which make up 99% of the variation......235 Figure 9-9 FPCA: Each of the six three FPC as variations about the mean along with the percentage of total variation explained by the component. The solid black line represents the mean SBP over the day. The "+" curves illustrate what happens when a small amount of the component is added to the mean and the "-" curves Figure 9-10 Scatter plots and the corresponding correlations between all the twocomponent cosinor random-effects model parameters and the principle component scores from FPCA. Eight FPCs contribute to 99% of the total variance in the data.237 Figure 9-11 Simulated ARV values taken every 10mins and every 30mins......239 Figure 9-12 Simulated data for one individual with readings taken 1) every 30min and 2) every 10mins......240

LIST OF ABBREVIATIONS

Abbreviations Term

| ARP | Ambulatory Blood Pressure |
|--------|--|
| ABPM | Ambulatory Blood Pressure Monitoring |
| ACFI | Angiotensin-Converting Enzyme Inhibitors |
| ACF | Autocorrelation Function |
| ACR | Albumin:Creatinine Ratio |
| AR | Autoregressive |
| ARB | Angiotensin Recentor Blockers |
| ARV | Average Real Variability |
| BMI | Body Mass Index |
| RP | Blood Pressure |
| BDV | Blood Pressure Variability |
| CCB | Calcium Channel Blockers |
| | Confidence Intervals |
| CV | Coefficient of Variation |
| CVD | Cardiovascular Disease |
| | Diastolic Blood Prossure |
| | Empirical Post Linear Unbiased Predictors |
| ECG | Electrocardiogram |
| ECG | |
| | Functional Data Analysis |
| | Functional Principle Components Analysis |
| | Concretized Estimating Equation |
| GEE | |
| GP | General Practitioner |
| HR | |
| HSE | Health Services Executive |
| LHC | |
| | |
| LVH | Left Ventricular Hypertrophy |
| | Left Ventricular Mass Index |
| MESOR | Midline Estimating Statistic Of Rhythm |
| ML | Maximum Likelihood |
| MLE | Maximum Likelihood Estimates |
| NICE | National Institute For Health And Care Excellence |
| OR | Odds Ratio |
| PACE | Principal Component Analysis through Conditional Expectation |
| PCA | Principle Component Analysis |
| PRISMA | Preferred Reporting Items for Systematic reviews And Meta- |
| | Analyses |
| REML | Restricted (or Residual) Maximum Likelihood Estimation |
| ROC | Receiver Operating Characteristic |
| SBP | Systolic Blood Pressure |
| SD | Standard Deviation |
| SITAR | SuperImposition by Translation And Rotation |
| TILDA | The Irish Longitudinal Study On Ageing |

| TOD | Target Organ Damage |
|-----|-------------------------|
| VPC | Visual Predictive Check |
| wSD | Weighted 24-h SD |

DECLARATION

I declare that this thesis has not been submitted for another degree at this or at any other University. The work, upon which this thesis is based, was carried out in collaboration with a team of researchers and supervisors who are duly acknowledged in the text of the thesis.

| Signed: | Da | te: |
|---------|----|-----|
| 0 | | |

ACKNOWLEDGMENTS

I would like to thank my extraordinary supportive supervisors, Dr. Tony Fitzgerald and Prof Patricia Kearney who both encouraged me to join the Department of Epidemiology and then later to pursue a PhD. Thank you for never refusing to give advice or support and for always being so accessible to me over the past number of years. I specifically thank Tony for all his statistical advice and guidance, humour but above all his patience. I would not have been able to complete the thesis without his help. I thank Patricia for all her medical input and practical thinking which always reminded me not to forget the bigger picture and the purpose of the work. I would also like to thank Dr. Xia Li who joined my supervisory team midway through the thesis and who was always very helpful. I also thank Dr. John Newell and Prof Kate Tilling for their help along the way.

Thank you to the Health Research Board who awarded me a place on the PhD Scholars Programme in Health Services Research and who funded my PhD throughout. Thank you to Sheena and Martin for proof reading the thesis.

Finally, I want to thank my family (especially my granny), friends, classmates and everyone in the Department of Epidemiology who were always extremely helpful and supportive.

I would like to dedicate this PhD to my parents, for everything.

THESIS ABSTRACT

Blood pressure (BP) is one of the most extensively researched topics. It is well established that elevated BP is the most prevalent treatable risk factor for cardiovascular disease. Despite our comprehensive knowledge of the importance of reducing mean levels of BP, we are less informed about the benefits of reducing other parameters of BP, specifically BP variability (BPV), which refers to the amount of variation over a period of time. Recent evidence has suggested that BPV may be an additional clinical target to mean level. However, its full prognostic significance and definition remains in doubt. BP does not remain stationary throughout the day but is constantly fluctuating and follows a circadian rhythm. Short-term BPV refers to fluctuations over this circadian rhythm. Ambulatory blood pressure monitoring (ABPM), which can be used to obtain estimates of BP usually every 30mins over a 24h period, offers a powerful tool in the analysis of circadian patterns and shortterm BPV. Longitudinal circadian data with such a cyclical structure consisting of multiple repeated readings provides an opportunity to analysse BP in many different ways and an overview of possible approaches is outlined in this thesis. The main aims of this thesis were to explore and identify circadian BP patterns between individuals and groups, and extract meaningful measures that describe these patterns while appropriately accounting for the inherent cyclical structure of ABPM data. Specifically, the thesis includes a systematic review which identifies summary measures of BPV, such as standard deviation, that can easily be obtained from the observed data without the need for more advanced modelling. A meta-analysis exploring the correlation between short-term BPV and subclinical target organ damage (TOD), specifically left ventricular mass index, is included. The association between the identified summary measures and subclinical TOD is then explored in a group of middle aged adults. In an attempt to maximise the power of the repeated cyclical readings in ABPM and incorporate the data together in one model, different random-effects models were explored which allowed us to obtain estimates of both

within and between-individual variation of model parameters. A piece-wise linear mixed-effects model was considered as a simple but suitable approach to capture BP trajectory throughout the day. We attempt to relate factors such as the morning slope and within person variability (allowing this to be group-specific) to TOD (microalbuminuria). Finally, a two-component cosinor random-effects model is outlined where derivatives of the model fit presents a novel alternative method to locate and quantify the magnitude of slopes at critical points along the trajectory. This is used to obtain a measure of morning BP surge. We compare the random-effects from this model to principle component scores obtained through functional principle component analysis. Our motivating data comes from the Mitchelstown Study, a population based study of Irish adults where a subsample underwent 24h ABPM.

1. INTRODUCTION

1.1 Introduction

Hypertension

Cardiovascular disease (CVD) is now endemic worldwide. It is the leading cause of death, not only in Ireland but globally, accounting for 17.3 million deaths per year (30% of global deaths), a number that is expected to grow to greater than 23.6 million by 2030 [1]. Despite our comprehensive knowledge of prevention and treatment for hypertension, it is still the most prevalent treatable risk factor for CVD, affecting one billion people globally [2-4]. It has been estimated that 7.6 million deaths worldwide, 54% of strokes, and 47% of cases of ischaemic heart disease can be attributed to high blood pressure (BP) alone [1]. The most recent estimates for the Republic of Ireland (ROI) conducted by the Institute of Public Health using data from the SLÁN 2007 study estimated that more than 850,000 (25.1%) adults in ROI have hypertension with the figure set to rise to almost 1,220,000 adults (28.3%) by 2020 [5].

In order to accurately ascertain a "normal BP" level it is not simply enough to arbitrarily take one BP measurement and based on this single reading conclude whether an individual is above or below the generally accepted value of 140/90 mmHg. It is well known that BP does not remain stationary and is a haemodynamic variable that fluctuates from second to second making the diagnosis of hypertension challenging. In addition it has long been known that each individual has their own diurnal BP trajectory that varies over a 24h period, known as a circadian rhythm. Early evidence illustrated that BP tended to be higher during the day and much lower during the night [6]. This led to the term "dippers" being coined during the late 1980's referring to individuals whose BP fell or dipped at night [7]. Modelling and extracting features of longitudinal data, specifically BP data that is circadian in nature offers a unique challenge for researchers.

Awareness, treatment and control levels are problematic worldwide [4]. In Ireland, The Irish Longitudinal Study on Ageing (TILDA) reported that among those over 50 with hypertension, only 55% were aware of their hypertension status [8]. In addition, only 59% of those were on treatment, of whom only 52% were controlled

to suitable levels. Furthermore, there is on-going controversy that what we have always considered a normal hypertension cut-off (above 140/90mmHg) may still be too high. Results from the internationally acclaimed Systolic Blood Pressure Intervention Trial (SPRINT) suggest target levels of 120/70 mmHg are desirable [9]. SPRINT randomly assigned 9361 people without diabetes with systolic BP (SBP) above 130 mmHg and an increased cardiovascular risk to a target of less than 120 mmHg (intensive treatment) or a target of less than 140 mmHg (standard treatment). After 3 years the trial was stopped early due to a significant difference between treatment arms. Those assigned to the intensive group had a 25% lower relative risk of major cardiovascular events and 27% lower relative risk of all-cause mortality. Potential weaknesses of the study have been highlighted [10, 11]. Cushman et al. pointed out that BPs were measured with patients seated in a quiet room with an automated device without any observer present and argue this method can present BP values up to 20 mmHg lower than clinic readings [10]. The SPRINT trial is not alone in suggesting the threshold should be lower however and a recent meta-analysis examining the effect of intensive BP reduction found a 14% reduction in major cardiovascular events and 13% reduction in myocardial infraction when BP was reduced to 133/76 mmHg compared to standard treatment (140/90 mmHg) [12].

Measurement

Accurately measuring something that is fluctuating so frequently can be problematic. However, advancements in technology and wearable devices make the task of obtaining accurate BP readings throughout a 24h period much more achievable [13]. Ambulatory blood pressure monitoring (ABPM) is recognised as the gold standard of BP measurement [13] and involves wearing a small digital BP device that is attached to a belt around your body and which is connected to a cuff strapped around your upper arm. The devices are usually set to record BP values over a 24h period every 30mins, but can be set to 15mins or 20min [13]. After the device has been worn it can be removed and connected to a computer where the data are extracted, giving a print out of the average BP values and a visual representation of the circadian pattern. ABPM offers a unique insight into an

individual's underlying circadian rhythm and has many advantages over a single clinic reading; measurements can be obtained as individuals go about their daily lives giving a more accurate measure of their real BP values, it allows the detection of white-coat and masked hypertension, but also offers estimates of the night BP and dip parameters. Detecting white-coat is important as it may prevent the unnecessary commencement of antihypertensive medication. We now know that the dip at night is associated with more favourable outcomes, and mean night BP is established as a stronger predictor of outcomes than mean day values [14-16]. For example, in the Dublin Outcome Study, for each 10-mmHg increase in mean night SBP, the mortality risk increased 21% while adjusting for clinic BP [14].

In primary care according to NICE guidelines [17], ABPM is offered if the clinic BP (average of two single readings) is above 140/90 mmHg. Hypertension diagnosis is then based on thresholds that vary slightly based on average day/night values [13, 18]. Recently there have been strong calls advocating for the mandatory use of ABPM and even suggesting that failure to provide it amounts to medical ineptitude [13, 19, 20]. In Ireland the recent national cardiovascular health policy recommended general practises (GP) be encouraged to invest in BP technology (ABPM) that will improve BP management [21]. As a result the Health Service Executive (HSE) has acknowledged the importance of ABPM having recently approved the reimbursement of ABPM in primary care for those with a medical card [22]. Kario argues that in order to obtain perfect 24h BP control it is not enough to focus on the reduction of mean 24h BP, we must also reduce exaggerated blood pressure variability (BPV) and obtain an undisrupted or smooth circadian pattern [23]. Thus, one of the potential benefits of ABPM is that it offers us an opportunity to obtain measures beyond average day/night/24h BP values. More specifically, it allows us to quantify and explore the additional prognostic significance of variability in BP.

Blood Pressure Variability (BPV)

It is important to note there are primarily two types of BPV- long-term variability which refers to variation in readings often taken over months or years and shortterm variability which refers to variation in readings taken over minutes or hours. There has been a recent surge in interest in the prognostic value of long-term BPV since the publication of work by Rothwell et al. [24, 25]. In treated hypertensive patients enrolled in ASCOT-BPLA (Anglo-Scandinavian Cardiac Outcomes Trial-Blood Pressure Lowering Arm), higher visit-to-visit variability in SBP was associated with stroke and coronary events. Rothwell also found that SBP variation between visits was a strong predictor of stroke and coronary events, independent of mean clinic BP. To synthesize findings from subsequent studies, Tai et al. recently conducted a meta-analysis of 13 studies to evaluate the prognostic value of visit-tovisit SBP variability by different parameters in 77,299 patients [26]. They found a pooled age and mean SBP-adjusted hazard ratio (HR) for all-cause mortality of 1.03 (95% CI, 1.02-1.04; p<0.001) per 1-mmHg increase in SBP standard deviation (SD) and 1.04 (1.02-1.06, p<.001) per one precent increase in SBP coefficient of variation (CV), with the corresponding values for cardiovascular mortality being 1.10 (1.02-1.17, p<.001) and 1.01 (0.99-1.03, p=.32), respectively. In addition, a 1-mm Hg increase in SD was significantly associated with an increased stroke risk, with an HR of 1.02 (1.01-1.03, P<.001).

The prognostic value of short-term BPV obtained from ABPM has also been examined, but with inconsistent results. Parati *et al.* were the first to demonstrate high BPV was associated with an increased risk of target organ damage (TOD) specifically left ventricular hypertrophy (LVH) [27]. There have been a number of other studies also reporting associations between short-term BPV and both TOD [28-31] and cardiovascular events [25, 32]. However, other studies have shown no or only weak associations after adjustment for the mean BP [32-35]. The majority of these studies however, have used SD as a measure of BPV. The appropriateness of such an index has been disputed because it only reflects the dispersion of measurements around a single value (mean) and does not account for the order in which BP measurements were obtained and the longitudinal variation in the

circadian data [36, 37]. Alternatively, studies have explored other indices of variability including:

- coefficient of variation which attempts to adjust for the tendency of those with a higher average BP to also have a higher SD [25, 38, 39];
- average real variability (ARV) which is the average absolute difference between successive readings, and is thought to give a true reflection of real variability [28, 33, 36] and;
- a weighted 24h SD (wSD) which attempts to remove the influence of the day-night BP difference from the estimate of BPV [28, 30, 33, 40].

Mule *et al.* demonstrated an association between BPV using ARV, SD and microalbuminuria among a group of 328 hypertensives after adjustment including mean BP [41]. In a separate study, Mule *et al.* found that ARV SBP was associated with microalbuminuria in 315 untreated essential hypertensives after adjustment for covariates including mean SBP [42]. However wSD, and SD of day and night periods were not found to be independently associated with microalbuminuria.

Moreover, there has also been extensive work examining variability during the morning period alone where BP rises rapidly to its peak, known as morning surge, before falling again throughout the day [23]. There is substantial evidence indicating that the morning is the most important period and is when cardiovascular events most frequently occur [23, 43, 44]. It has been shown that the morning surge (calculated as mean SBP during the 2 hours after awakening minus mean SBP during the hour that included the lowest sleep BP) is independently related to organ damage and risk of cardiovascular events [45-47]. Different indices of short-term BPV and various morning surge parameters can be seen in Table 1-1. The short-term summary measures are outlined in more detail in Chapter 3.

| Measure of BPV | Description |
|--------------------|--|
| 24h SD | Standard deviation over 24 hour period |
| Day SD | Standard deviation over day period usually 9am-9pm |
| Night SD | Standard deviation over night period usually 1am-6am |
| wSD | Weighted standard deviation (of day and night standard deviation) |
| CV | Coefficient of variation |
| ARV | Average real variability |
| Morning Surge | |
| Sleep-trough surge | 2h morning BP minus 1h lowest night BP |
| Prewaking surge | 2h morning BP minus 2h prewaking BP |
| Rising surge | BP on rising minus last ABPM in a supine position <30m before rising |
| *Kania [45] | |

Table 1-1 Short-term BPV Summary Indices

*Kario [45]

Despite increased interest and research conducted on BPV we are still unclear as to what the best method or index is to quantify variability, and its clinical relevance. In addition, all the above studies have used direct summary measures of BPV. That is, measures that can be directly obtained from the raw data without the need for advanced statistical methods. This thesis aims to explore other possible novel measures that describe BPV. Considering the clinical importance of BP it is surprising to learn of so few studies that fully utilise the benefits of the longitudinal nature of ABPM. Lambert *et al.* [48] who explored the use of cubic splines to model the trajectory of ABPM emphasised the lack of research on the longitudinal analysis of 24h ABPM while Edwards *et al.* [49] who utilised orthonormal polynomials stressed there is no current "standard" method for analysing ABPM. These and other longitudinal models of ABPM variation are described in more detail in Chapter 2 (Longitudinal Data Analysis and Application to ABPM).

Biological Plausibility of BPV

Although the precise mechanisms responsible for BPV are not entirely understood [50], there are a number of acknowledged factors that influence variations in BP. Parati *et al.* [51, 52] argue it is primarily modulated by neural (increased central sympathetic drive and reduced arterial and cardiopulmonary reflexes), humoral (angiotensin II, insulin, bradykinin, nitric oxide) and vascular effects (elastic properties of arteries). The night-time dipping phenomenon that occurs during sleep is associated with a marked drop in sympathetic drive, while the morning

surge is associated with a remarkable sympatho-activation [53, 54]. In addition, behavioural influences such as sleep, physical activity and postural changes can induce significant changes to BP [52, 55]. Emotional or psychological stress can also sporadically increase BP values [50]. An emotional condition that is often seen in clinical practice is that of the "white-coat effect" [50]. This well-known phenomenon is characterised by a sharp increase in BP values during the course of a doctor's visit which then disappears once the doctor leaves. Even high altitude [56] and environmental (seasonal effects) [57] factors are known to have an effect on BP values.

BPV and Antihypertensive Medication

The interest in BPV and the importance of how it is quantified comes to the fore when we begin to examine therapeutic approaches to treat hypertension. In 2005, the ASCOT trial reported that antihypertensive medications (amlodipine and perindopril combination) significantly improved outcome compared to older dated alternatives [58]. However, of particular interest was that the difference in mean BP between treatments was only 1.6 mmHg at the end of the trial. This small difference suggested that the improvement in outcome between arms was unlikely to be as a result of reducing mean BP alone. In the subsequent years more analyses published from the study showed that the newer medication had significantly reduced central BP (pressure exerted on the heart and brain), night-time BP and BPV [25, 59, 60], suggesting we should not be solely focusing on mean BP. It highlighted the potential benefit of obtaining new indices of BP and the advantage of exploring ABPM data. There is growing evidence to suggest that although different antihypertensive classes have the same effect in terms of reducing mean BP, they have significantly different effects on BPV, particularly calcium channel blockers (CCB) or calcium antagonists [61, 62]. The X-CELLENT trial examined the efficacy of calcium antagonists in reducing BPV (measured as wSD and ARV) among 577 patients before and after anti-hypertensive treatment [63]. Larger reductions were obtained with amlodipine (CCB) compared to indapamide (diuretic), candesartan (angiotensin receptor blocker (ARB)) and placebo where the differences even remained significant after adjustment for mean BP. Further

evidence was found in a large cohort of 2780 hypertensive patients, where those receiving either CCBs or diuretics, alone or in addition to other drugs had significantly lower 24h SBP SD compared with angiotensin-converting enzyme inhibitors (ACEIs), ARBs or β -blockers alone or in combination [64]. CCB-based combinations again performed better than other combinations in post hoc analysis of the COPE (Combination Therapy of Hypertension to Prevent Cardiovascular Events) trial in reducing visit-to-visit BPV (SD & CV) [65]. The cope trial was the first clinical trial to examine the treatment of hypertension with combination therapies. This raises the question of whether BPV could be an additional therapeutic target of antihypertensive treatment to improve cardiovascular protection. And highlights potential benefits of obtaining new BP measures which can provide new information and bring about greater insights into the mode of action of new antihypertensive medication [52, 66].

These variability indices may be particularly useful for the analysis of chronotherapy effects which refers to the treatment of an illness or disease by administering a drug at a time of day believed to be in harmony with the body's natural rhythms. There are no current recommendations for when BP-lowering medications should be ingested but most people are instructed by their physicians and pharmacists to take the medication during the morning period. As explained previously, the morning period is a crucial time coinciding with a surge in BP and perhaps the administration of medication before this period may be more beneficial in reducing BP rather than during it. The MAPEC study was the first to examine the benefits of bedtime chronotherapy with one or more conventional hypertension medications on BP control and CVD risk reduction versus conventional morning therapy [67, 68]. They found subjects ingesting antihypertensive medication at bedtime showed significantly lower night BP, reduced prevalence of non-dipping and lower relative risk of both total CVD and major CVD events after a median follow-up of 5.6 years. In a similar study conducted among 448 hypertensive patients with type 2 diabetes, comparable results were found [69]. After a median follow up of 5.4 years patients with diabetes ingesting at least one hypertension medication at bedtime showed significantly lower night BP, higher prevalence of controlled ambulatory BP (ABP)

and a significantly lower cardiovascular risk than those taking medication upon awaking. Improved BP control was also shown in a separate study of 250 patients with resistant hypertension [70]. Indeed a systematic review [71] and a Cochrane review [72] have suggested that patients with evening dosing of antihypertensive drugs had better 24h BP control than those with a morning dosing regimen. Although the clinical significance of night-time administration needs to be explored further it does emphasize the need to be able to quantify the immediate effect of ingesting an antihypertensive drug so that we have the ability to compare different drug classes.

Summary

The primary focus of the management of hypertension to date has centred on lowering mean BP with little consideration for other factors, such as BPV, maximum BP reached, episodic hypertension or its circadian pattern. Hypertension guidelines have only focused on mean BP, which is clearly important, but fail to mention BPV [17]. As ABPM is more widely available and we begin to collect more and more data, it begs the question of whether we are maximising the rich data available. Recently published ABPM guidelines have highlighted that most ABPM studies have reported results derived from investigating mainly one unique ABPM characteristic, e.g., either a BP mean value, or a variability measure, or morning surge, without comparison or appropriate adjustment for the prognostic value of additional ABPMderived characteristics [73]. It is clear that the full potential available from ABPM measurements is not currently being explored. This thesis explores current approaches and suggests novel alternatives to capture circadian BP patterns.

1.2 Study Design – Motivating Dataset

The motivating ABPM data for this thesis comes from a study examining the prevalence of major CVD risk factors in a middle-aged population in Ireland. The analysis utilises existing data obtained from the Mitchelstown Cohort Study, a population based study of middle-aged men and women, recruited in Ireland in 2010-2011. A detailed description of the study design is available elsewhere [74] but a summary is provided here.

The primary aim of the study was to provide a profile of cardiovascular health and their related factors in an Irish adult general population sample. The study comprised of 2047 adults aged 47-73 years (response rate: 67%) recruited from patients attending a single large primary care centre, the Livinghealth Clinic (LHC) in Mitchelstown, County Cork. The clinic is a GP developed independent advanced Primary Care Centre. The clinic serves a catchment area of 20,000 people, with a mix of urban and rural residents. At baseline, participants completed a detailed health and lifestyle questionnaire and were invited to attend the LHC for a physical examination to be carried out by a nurse trained in the study research protocols. Clinic measurements included height, weight, BP and in addition, fasting blood samples (minimum of 8-h fast) and urine samples were collected. Participants also underwent standard 12-lead electrocardiogram (ECG). ABPM was offered to all 2047 participants, and it was completed by 1207 (response rate: 58%). All participants provided written informed consent and ethical approval was obtained from the Clinical Research Ethics Committee of the Cork Teaching Hospitals.

BP measurement

With the participant in a seated position, study BP was measured using an OMRON M7. Three readings of BP were obtained and the average of the second and third measurements was used. ABP was measured using dabl ABPM system (dabl ltd., Ireland) with the Meditech ABOM-05 Monitor (Meditech LTD., Hungary). The monitors were programmed to obtain readings every 30mins and remained in place

for 24h. Participants kept diaries of wake and sleep periods, which were used to calculate day and night BP respectively. If no diary was kept, the period from 1am to 6am was used as the night period and from 9am to 9pm as the day period.

Subclinical Target Organ Damage (TOD)

To explore the prognostic value of BPV and for purposes of illustration of some of the methods, subclinical TOD is used as a surrogate marker of cardiovascular events where the association between BPV and TOD is investigated. Subclinical TOD, which is the development of asymptomatic functional and structural abnormalities in the human body, often precedes the occurrence of major cardiovascular events [75]. It usually refers to damage that occurs in major organs fed by the circulatory system e.g. heart, kidneys, brain and eyes. As they are subclinical, they are nearly or completely asymptomatic i.e. showing no signs or symptoms. Two examples of TOD are microalbuminuria (kidney) and LVH (heart).

Microalbuminuria is the persistent elevation of albumin in the urine [76]. It occurs when the kidney leaks small amounts of albumin into the urine. Specifically, it signals increased permeability (capacity of a blood vessel wall to allow for the flow of small molecules or whole cells in and out of the vessel) of the endothelial cells and signifies that some level of injury is present and vascular responsiveness is comprised [76]. It is a marker of increased risk for cardiovascular morbidity and mortality especially, but not exclusively, in high risk populations such as diabetes and hypertensives [77]. Microalbuminuria is measured in spot morning urine obtained from the patient and sent for measurement of both albumin and creatinine. A meta-analysis in 2010 demonstrated increased risk of mortality with urine - albumin:creatinine ratio (ACR) \geq 1.1 mg/mmol and as a result microalbuminuria was defined using this cut-point [78].

LVH is enlargement and thickening (hypertrophy) of the walls of the heart's main pumping chamber (left ventricle). LVH can develop in response to factors such as increased BP or a heart condition that causes the left ventricle to work harder. Complications of LVH include atrial fibrillation, diastolic heart failure, systolic heart failure, and sudden death [79]. Both earlier recognition and improved understanding of cardiac hypertrophy may lead to more effective therapeutic strategies for this cardiovascular risk factor [79]. An electrocardiogram (ECG) is usually used to detect the presence of LVH. In this study, a 12-lead ECG was obtained in the standard manner using a Siemens – Eclipse 850i machine. ECGs were reviewed and coded for LVH voltage criteria by a cardiologist. The presence of LVH was defined by Cornell Product ECG voltage criteria i.e. SV3 + RaVL (+6 in women) X QRS duration \geq 2440 mm x ms [80].

Reproducibility Sub-study (Follow-up Data)

Based on the initial ABPM results the sample was divided into 4 groups: normotension, isolated nocturnal hypertension, isolated daytime hypertension and day-night hypertension [81]. Twenty participants were randomly selected from each group and invited to attend for follow-up ABPM measurements in 2014 using the Spacelabs 90217 monitor. Data was stored using the Spacelabs 92506 Ambulatory BP Report Management System software. At four years follow-up a subsample of fifty (response rate: 63%) of these participants consented to participation in this study. The mean period of follow up was 3.9 years.

1.3 Aims and Objectives

The aim of this thesis was to explore and identify circadian BP patterns between individuals and groups, and extract meaningful measures that describe these patterns.

Specifically the objectives were:

- To conduct an extensive literature review to determine current summary measures that is used to quantify BPV and explore their association with the presence of TOD, specifically LVH.
- To determine the association between short-term BPV extracted from 24h ABPM and subclinical TOD in middle aged adults based on the measures found in (1).
- 3. To implement models that adequately describe patterns of BP and which can provide novel measures of short-term BPV taking into account the full temporal and circadian nature ABPM.
- To outline an approach that can be used to comprehensively interrogate the reproducibility of ABPM readings taken at two different occasions incorporating some of the approaches outlined in (3).

1.4 Thesis Outline

This thesis is comprised of four papers, Figure 1-1 illustrates each aim and objective and the corresponding chapter.

This thesis focuses on extracting features of longitudinal data, specifically data that is circadian. Although the focus is BP, the methods covered could be easily applied to other physiological processes that follow this circadian cycle. Chapter 2 first outlines approaches suitable for modelling longitudinal data in general but then focuses on specific techniques that may be appropriate for BP.

Chapter 3 is a systematic review that examines the prognostic significance of shortterm BPV on the presence of TOD, specifically LVH. The review identifies variability summary measures that are currently used to quantify BPV. A meta-analysis exploring the correlation between short-term BPV and left ventricular mass index (LVMI) is included.

Chapter 4 explores the association between short-term BPV over 24h and subclinical TOD in middle aged adults using data from the Mitchelstown study.

Chapter 5 illustrates a method of describing and quantifying circadian BP patterns using piecewise linear mixed-effects models while ensuring periodicity.

Chapter 6 examines the traditional ABPM model (cosinor) using random-effects and illustrates how it can be extended and used as a method to determine morning BP surge. The model is compared to a functional principle component analysis.

Chapter 7 explores approaches to determine the reproducibility of ABPM trajectories taken at two different occasions.

Chapter 8 summarises the findings from the analysis, discussed the strengths and limitations of the thesis, and makes some suggestions for future research.

Aim

To explore and identify circadian BP patterns between individuals and groups, and extract meaningful measures that describe these patterns.



Figure 1-1 Overview of thesis including aims and objectives.

2. LONGITUDINAL DATA ANALYSIS & APPLICATION TO ABPM

2.1 Chapter Overview

This chapter introduces hierarchical and specifically longitudinal data structures. This is followed by a detailed description of statistical techniques available that are suitable for the analysis of ABPM both in terms of accounting for the longitudinal nature of the data but also methods that are capable of capturing non-linear circadian BP curves. A graphical illustration of each technique is presented using the motivating dataset as an example. This is followed by a summary of its application in the literature, paying particular attention to the use of the technique to analyse BP data. Particular attention is drawn to the techniques used in this thesis; however alternative methods are described briefly. A comparison of the different approaches is given at the end the chapter.

2.2 Overview of Longitudinal Data

Many epidemiological studies have a natural hierarchical data structure associated with them where there are different levels of information. For example, data from a number of different hospitals would comprise of hospital level information but within each hospital (or cluster) we could have patient level information. The term hierarchical (or nested/multilevel) data is given to data with this structure. Longitudinal data (panel data or repeated measures) can be considered a special case of hierarchical data where a sample is followed up over time with information collected at several occasions or time points. This type of data can be considered as a two level cluster where there are repeated measurements (level 1 units) with a natural ordering nested within individuals (level 2 units). It is important to highlight the natural ordering of the measurements which are not exchangeable, unlike for example patients nested within hospitals. Recognising both the hierarchical structure of the data points and the natural ordering in time of the data points is critical. Measurements within clusters or subjects will tend to be more similar than measurements from different subjects. In addition, the ordering of time points means that within-subject readings close together may be more correlated than points further away. Both issues of correlation must be accounted for in the analysis and renders traditional regression analysis techniques, which assume

independence of observations for non-hierarchical data to be inappropriate. ABPM is a perfect example of such hierarchical data where there are many repeated BP readings on an individual taken over a 24h period with a natural order (in time) to the observations. An example of four individual ABPM readings from the Mitchelstown ABPM dataset is provided in Figure 2-1.



Figure 2-1 Example of four individual ABPM readings
2.3 Statistical Approaches for Modelling Longitudinal Data

Before the specific analysis of ABPM is introduced, a general description of statistical approaches for the modelling of longitudinal data is outlined.

There have been substantial developments in statistical methodology for the analysis of longitudinal data over the last 40 years. One of the key drivers has been improvements in technology which means computationally complex problems can now be solved in a fraction of the time taken previously. This, coupled with increased widespread availability of software programs and packages which are becoming more and more accessible to users, means researchers have access to a variety of more rigorous approaches for the analysis of longitudinal data [82]. Two of the most widely used approaches are generalised estimating equations (GEE) models [83-85] and mixed-effects models [86, 87]. As both methods make an adjustment for the dependency of the observations within an individual, both are suitable for the analysis of longitudinal data. However, there are differences in how both methods adjust for this correlated data leading to different model assumptions. In addition, the two models can be implemented to answer different questions relating to longitudinal data.

Generalised Estimating Equations

GEE models (or marginal models) are often referred to as population-average models where the target of inference is the population [88]. The term marginal is used as the focus is on the mean response, which depends only on the covariates of interest and does not specify the joint distribution of the individual's observations. An important point in the analysis of marginal models is that the mean response and the within-subject association are modelled separately. The latter is referred to as a nuisance characteristic that must be accounted for in the analysis so that correct inferences about the changes in the population mean response (primary goal) can be made [88, 89]. Separately modelling the mean response and the within-subject association has the important implication that the regression coefficients derived from GEE models have population-average interpretations, where they describe features of the mean response and how these relate to covariates. GEE uses a partial-likelihood approach to estimate the parameter of interest.

A GEE model can be defined as the following three-part specification [88]:

The marginal expectation of the response, $E(Y_{ij}|X_{ij}) = \mu_{ij}$, depends on the covariates, X_{ij}, through a known link function

$$g(\mu_{ij}) = \eta_{ij} = X'_{ij}\beta \tag{1}$$

The variance of each Y_{ij} , given the covariates, depends on the mean

$$\operatorname{Var}(Y_{ij}|X_{ij}) = \phi v\left(\mu_{ij}\right)$$
⁽²⁾

where $v(\mu_{ij})$ is a known "variance function" (known function of the mean, μ_{ij}) and ϕ is a scale parameter that needs to be estimated when the response is continuous [88].

The pairwise (or two-way) within-subject association among the vector of repeated responses, given the covariates, is assumed to be a function of the means, μ_{ij} , and an additional set of within-subject association parameters, α . Given a model for the pairwise correlations, the corresponding covariance matrix can be constructed as the product of standard deviations and correlations

$$V_i = A_i^{1/2} \text{Corr}(Y_i) A_i^{1/2},$$
(3)

where A_i, is a diagonal matrix with $Var(Y_{ij}|X_{ij}) = \phi v(\mu_{ij})$ along the diagonal (and $A_i^{1/2}$ is a diagonal matrix with standard deviations, $\sqrt{\phi v(\mu_{ij})}$, along the diagonal), and Corr(Y_i) is the correlation matrix (here a function of α). To distinguish it from the true underlying covariance among the Y_i , V_i is known as a working matrix which acknowledges the uncertainty about the assumed model for the variances and within-subject associations. Unless they have been modelled correctly, our model for the covariance matrix may not be correct [88].

The GEE estimator of β for marginal models (or generalised linear models for longitudinal data) can be obtained by minimizing the following function (where *N* is the sample size):

$$\sum_{i=1}^{N} \{ y_i - \mu_i(\beta) \}' V_i^{-1} \{ y_i - \mu_i(\beta) \}$$
(4)

It can be shown that if a minimum of the function given by equation (4) exists then, it must solve the following GEE:

$$\sum_{i=1}^{N} D'_{i} V_{i}^{-1} (y_{i} - \mu_{i}) = 0$$
(5)

where $D_i = \partial \mu_{ij} / \partial \beta$ is the derivative matrix containing the derivative of μ_{ij} with respect to the components of β [88].

Mixed-Effects Regression Models

While GEE models are focused on inference for the population, mixed-effects (subject-specific) models target inference at the individual level. The full-likelihood approach associated with mixed-effects models provides estimates of subject-specific effects (e.g. subject-specific trajectories) that are useful for understanding between individual variability in the longitudinal response [82]. These are known as random-effects which are coefficients that are allowed to vary between individuals (clusters). The estimation of the random-effects determines the subject-specific curves and explains the correlation structure of the longitudinal data [90]. Mixed-effects models allow us to estimate both the degree of variation within a person (within-subject variation) and in the population of individuals (between-subject variation) [82].

The simple linear mixed-effects model [86-88], which is an extension of the simple linear regression model, can be written as:

$$y_{ij} = \beta_0 + b_{0i} + (\beta_1 + b_{1i})t_{ij} + e_{ij}$$
(6)

where y_{ij} is the response value for the j^{th} measurement on the i^{th} subject, at time t_{ij} , the β 's are the fixed effects coefficients associated with the population (average) intercept at β_0 (equivalent to linear regression model) and the population slope (β_1), b's are the random-effects where b_{0i} and b_{1i} are the intercept and slope deviations respectively for the *i*th subject, and e_{ij} represent the individual-level residuals from the model. A comparison between the traditional linear regression and the linear mixed-effects model is presented in Figure 2-2 where we have a response (y) plotted against time (t). In the fixed-effects linear regression plot all observations are considered independent. However, in reality the data may be clustered with repeat readings on different subjects, represented as colour coded dots in the mixed-effects plot. The black line represents the population average which is described by intercept β_0 and slope β_1 in the model. In addition, each individual obtains their own random-effects (b_{0i} and b_{1i}) which can be added to the population coefficients to obtain subject-specific curves (Figure 2-2).



Figure 2-2 Illustration of simple linear regression model and simple linear mixed effects model where the data are clustered

The individual level residuals are assumed to be independent and have mean zero and variance σ^2 . It is assumed the random-effects have zero mean and follow a bivariate normal distribution with a variance-covariance matrix Σ_b :

$$\sum_{b} = \begin{bmatrix} \sigma_{b0}^2 & \sigma_{b0b1} \\ \sigma_{b0b1} & \sigma_{b1}^2 \end{bmatrix}$$
(7)

where σ_{b0}^2 and σ_{b1}^2 represent the between-subject variation of the intercept and slope random-effects respectively. The magnitude and direction of the covariance term (σ_{b0b1}) gives information about the interaction between the random slope and intercept. In Figure 2-2 subjects with a high intercept have a steeper slope, giving a positive correlation between b_{0i} and b_{1i} , resulting in a positive covariance. Accounting for this is an important aspect of longitudinal modelling.

Although the model presented above is quite flexible some of the assumptions may be too restrictive when examining certain data structures. It can however, easily be extended to accommodate many situations. Like linear regression, the model can adequately make adjustment for confounding variables. The model assumes the response changes linearly over time but for many biological processes, the mean profile is a non-linear function of time [91]. One approach is to model the nonlinearity by including non-linear functions of time such as quadratic or cubic terms. It is important to note the model will remain linear in the random-effects. There are however, many other alternative approaches for modelling the mean (see section 2.4).

One of the defining properties of linear mixed-effects models is that it allows for the explicit analysis of between-subject and within-subject sources of variation i.e. the random-effects covariance structure can be explicitly defined [88]. This essentially involves defining two separate variance-covariance structures, one for the random-effects across subjects and one for the within-subject random errors [92]. This can be illustrated by rewriting the variance of *b*'s (random-effects) and e_{ij} (random errors) in matrix form:

$$var \begin{pmatrix} b_i \\ e_i \end{pmatrix} = \begin{pmatrix} G & 0 \\ 0 & R_i \end{pmatrix}$$
(8)

where *G* is the variance-covariance matrix for the random-effects across subjects. This is equivalent to the variance-covariance matrix defined in equation (6). The matrix R_i is the variance-covariance matrix for the within-subject random errors. The b_i (random-effects) and e_i (random errors) are assumed to be independent. As assumed in the mixed-efffects model above, the R_i matrix is often simplified as $\sigma^2 \mathbf{I}$, where \mathbf{I} is the identity matrix, assuming within-subject random errors to be conditionally independent with homogeneous variance, that is, they are independent conditional on the b_i random-effects which accounts for the intraindividual correlation [90, 92]. In order to get the total variance of y_i , we first rewrite the simple linear mixed-effects model (equation (6)) in matrix form where the response is given as:

$$Y_i = X_i \beta + Z_i b_i + e_i \tag{9}$$

where X_i and Z_i are design matrix of covariates. From this, the total variance (V_i) becomes:

$$V_i = Z_i G Z_i' + R_i \tag{10}$$

By constructing the design matrix Z_i and specifying structures for G and R_i matrices, the variance of repeated measurements can be adequately specified in linear mixed models [92]. The subscript i on the V_i matrix indicates that the total variancecovariance matrix depends on the subjects covariates. Under matrix notion the repeated measurements of the response Y_i follows the following multivariate normal distribution:

$$Y_i \sim N(X_i'\beta, V_i) \tag{11}$$

where β is a vector of the population parameters, with the first element representing the intercept.

The structure of *G* is often left unspecified when more than one between-subjects random term is specified in a linear mixed-effects model. When there is no explicit structure assumed, it is referred to as an unstructured pattern. This has the advantage that no assumptions have to be made about the variances and covariances of the random-effects [88, 92].

Regarding the R_i matrix, Fitzmaurice *et al.* argue that in their experience of longitudinal data, variances are rarely constant over time and emphasize the benefits of the absence of restrictions on the variances [88]. Ideally we would like to use a unstructured pattern for all models but the number of parameters that

have to be estimated grows rapidly with the number of measurement occasions. This can affect the computational stability; where a large number of covariance parameters need to be estimated, relative to the sample size (both subjects and repeat readings), estimation is likely to be unstable [88]. There are alternatives that try to simplify the problem such as the compound symmetry or exchangeable structure that specifies that observations on the same subject have homogeneous covariance and variance. Other structures can be specified for the R_i matrix such as Toeplitz or autoregressive residual structure where the correlation between observations on the same subject are not equal, but decrease towards zero with increasing lag. We are not confined to these structures however. For example, the between-subject or within-subject variation could increase or decrease over time in which case the variance could be modelled as a function of time [93]. A similar situation could arise where we allow variation to vary depending on a grouping variable [94]. Selecting the appropriate structure involves finding a compromise between over fitting and under-fitting the model which results in a parsimonious model. Liu suggests the structure can be selected according to theory or empirically [92] while Zuur et al. emphasizes that it is "important to model the correlation structure in a reasonable and meaningful way rather than to model the correlation structure perfectly" [93].

The parameters from a mixed-effects model (fixed coefficients, variances, covariances) can be estimated by a number of methods. Maximum likelihood (ML) is the best known technique of obtaining estimates of unknown parameters. The maximum likelihood estimates (MLEs) of the parameters are the values of the parameters that maximise the likelihood function (i.e. the values of the parameters that make the observed values of the dependent variable most likely, given the distributional assumptions) [95]. For the case of the linear mixed-effect model, it can be shown by obtaining the likelihood of the multivariate normal probability density functions of the model that the log-likelihood function is defined as:

$$l = -\frac{n}{2}ln(2\pi) - \frac{1}{2}\sum_{i=1}^{n}ln|V_i| - \frac{1}{2}\left(\sum_{i=1}^{n}(Y_i - X_i\beta)'V_i'(Y_i - X_i\beta)\right)$$
(12)

The natural log is introduced to simplify the mathematics as it converts the product of the individual density functions to a sum of log-density functions. In the case of linear regression (independent observations), maximising the likelihood function results in estimates that can be solved relatively easily as the equations have closed form solutions. In the case of linear mixed models, ML estimation can involve nonlinear equations for which there is no closed form solution for the parameters [93]. Instead, the ML estimates can be obtained using an iterative technique. There are a number of different algorithms that can be applied to obtain MLEs including iterative generalised least squares and iteratively re-weighted least squares which are discussed in detail elsewhere [94]. One drawback of the ML approach is that the estimates for the parameters of the covariance matrix are biased by a factor of (n-2)/n. The reason for this is because it ignores the fact that the intercept and slope are estimated [93]. Restricted (or residual) maximum likelihood (REML) estimation was developed to address this problem based on a slight modification on the likelihood above where it is defined only in terms of the covariance matrix V_{i} , thus separating it from β (estimated parameter). It is recommended to use REML over ML when possible as it will provide less biased estimates, although it is less of an issue with significantly large sample sizes [88].

The likelihood ratio tests (LRT) can be used for hypothesis testing. The LRT compares the maximised log-likelihood of two models, a full model with a nested model. A formal test is obtained by taking twice the difference in the maximised log-likelihood and comparing the statistic to a chi-squared distribution [88]. Despite the recommendation to use REML, in order to be able to compare models using LRT we first estimated the models using MLE to obtain ML solution.

42

Comparison of GEE and Mixed-Effects Analysis

GEE models use partial-likelihood methods to estimate parameters of interest whereas mixed-effects models are based on full-likelihood methods. The advantages of partial-likelihood is that they are computationally easier than fulllikelihood methods and they do not require distributional assumptions for the observations as there is no specification of the joint distribution, only the mean response [82, 88]. One of the appealing properties of the GEE approach is that when the data is balanced and under the condition that the mean response is modelled correctly, it produces consistent and unbiased estimates of regression parameters and corresponding standard errors even when the covariance structure is misspecified [88, 96]. This robustness can be attributed to the use of the empirical or sandwich estimator [97, 98] which produces valid standard errors for the parameters of the model even when the assumed covariance among the repeated readings is not correct. However, Fitzmaurice et al. argue that correctly specifying the covariance structure makes optimal use of the available data for estimation of the parameters of interest [88]. One of the main drawbacks of the sandwich estimator is that it is best suited to balanced longitudinal designs where there are a large number of individuals with a small number of observations. In situations where the design is severely unbalanced, with each individual having a unique sequence of measurement occasions in addition to having few subjects, sandwich based standard errors can be biased downward and result in underestimation of the variance of the regression coefficient [88]. For linear models with balanced data, GEE coefficients and the fixed-effects coefficients from a mixed-effects model will yield the same values. However, they are not equivalent for non-linear models [88]. By comparing the GEE function in equation (4) and the mixed-effects function in equation (12) similarities between the two can be seen. The larger mixed-effects function is essentially the GEE function with an additional variance component.

One of the key advantages of random-effects models over other techniques (e.g. GEE models, traditional analysis-of-variance approaches) is that they make full use of all available data from each individual and do not require balanced datasets. GEE

assumes that missing data are missing completely at random (MCAR). Fulllikelihood estimation random-effects rely on a weaker assumption that missing data are missing at random (MAR). Missing data are known to be completely at random (MCAR) when their absence is not related to both the observed and unobserved data [99]. MAR however refers to missing data that depends on observed data but not on unobserved data [99]. In the case of random-effects, missing data are ignorable if the missing responses can be explained either by covariates in the model or by the available responses from a given subject [82, 88]. A mixed-effect analysis does not require a balanced dataset because the covariance can be expressed as an explicit function of times of measurement (when times of measurement, or functions of time are included in Z_i [88]. In theory each individual can have their own unique sequence of time points. Additionally as highlighted previously, random-effects can allow the variance and covariance to be modelled as a function of time unlike other techniques which force the variance to be constant over time. Unbalanced data can however, have implications on specifying the autocorrelation residual structure when there are irregular gaps making it necessary to parameterise the autocorrelation function (ACF) explicitly [91]. For example, applying a continuous autoregressive error (CAR) structure is similar to applying a autoregressive structure but uses the actual value of measurement time which helps account for the issue of unbalanced data [94, 100].

It is the ability to estimate both between and within-subject variation that makes the use of mixed-effects models so appealing for the analysis of ABPM in general, but also specifically to help answer the questions posed in this thesis where the focus is on obtaining population and subject-specific measures of BPV and quantifying this variation both between and within individuals. For this reason the use of mixed-effects models is preferred over the GEE approach in this thesis.

2.4 Statistical Approaches for Modelling Individual ABPM Profiles

This section explores different approaches for modelling ABPM using randomeffects models from which variability measures can be obtained. Previous longitudinal modelling approaches to ABPM from the literature are discussed as well as suggesting other novel techniques. A comparison of the suggested techniques is also included at the end.

Summary Measures of BPV

The simplest form of random-effects is in essence subject-specific summary parameters, where the parameter is random. The idea is that multiple measurements on an individual are collapsed into one (sometimes more than one) summary measurement which is an estimate of a subject-specific measure. This can then be related to an outcome measure. The advantage of this method is that data with a longitudinal structure are converted to a cross-sectional problem where traditional regression techniques are valid without worrying about correlation between readings. This can lead to straightforward analysis and does not require statistical knowledge of multilevel modelling.

With the focus of the thesis on patterns of BP, summary measures in this context relate to short-term summary measures of variability, as opposed to say mean BP values. As highlighted in Chapter 1, summary measures have been the focus of much of the literature to date in relation to short-term BPV. This simple approach can also be considered a simple two-stage model where the summary measures of variability such as SD over 24h, are obtained first (first-stage) and then these values are entered into a subsequent second model (second-stage), such as a logistic regression model, exploring its effect on an outcome such as TOD. The main disadvantage of this method is the loss of information as the statistical power associated with longitudinal data is removed by collapsing the data into one value. The other issue is that quantifying variability is difficult as highlighted in Chapter 1 and attempting to do this with one measure makes the task significantly more challenging. Nevertheless, this does not mean that summary measures should be discarded. In fact many ABPM software programs routinely include estimates of

45

variability measures such as SD (dabl ABPM system, dabl LTD, Ireland) in their reports and are an important reference point. Because of the ease with which these can be obtained and included in an ABPM report, the exploration of suitable summary measures that capture variability deserves substantial attention prior to exploring more complex methods. For this reason Chapter 3 is a systematic review identifying summary variability measures while Chapter 4 applies these measures to our dataset and explores their association with TOD.

Polynomials

As is evident from Figure 2-1, like many biological processes, BP is not linear over time. Polynomial regression is a simple, but useful tool in the analysis of longitudinal medical data to cope with non-linear or curvilinear relationships [101]. The general principle of polynomial regression is to use increasing powers of time (t) as separate predictor variables to model the mean of the response variable of BP. For an individual, the model can be expressed as:

$$BP_{ij} = \beta_{0i} + \beta_{1i}t_{ij} + \beta_{2i}t_{ij}^2 + \dots + \beta_{pi}t_{ij}^p + e_{ij}$$
(13)

where BP_{ij} is the BP value for the j^{th} measurement on the i^{th} person, at time t_{ij} , the β 's are the polynomial regression coefficients that comprise of the average fixedeffect and the subject-specific random effect. The e_j represents the individual-level residuals from the model.

This model can be problematic however as the powers of *t* will be correlated, often quite highly so, which may not only lead to multicollinearity issues but also software convergence difficulties. Some of these issues can be overcome by centring the predictor variables, time in our case [102]. However, a better approach is to transform the powers into orthogonal polynomials, an equivalent set of predictor variables which are uncorrelated. These contain the same information, address the issue of collinearity while also helping to reduce the problem of convergence. To illustrate the flexibility that can be obtained from polynomial analysis, raw data were plotted for four individuals along with the fitted values from a 6th order polynomial model, Figure 2-3.



Figure 2-3 Raw data for four individuals along with predicted 6th order polynomial model

Corrao *et al.* implemented a 3rd degree polynomial on ABPM and demonstrated that the prevalence of TOD (LVH or ischemia) was higher in hypertensives with absence or inversion in BP circadian rhythm compared with hypertensives with preserved BP circadian rhythm [103]. Zwinderman *et al.* found that a 4th order polynomial fitted ABPM data quite well and could obtain trough-peak values from the predicted curves which could be used to compare antihypertensive medication but the sample size was a major limitation (n=10) [104]. More recently, Edwards *et al.* utilised a linear mixed model with orthonormal polynomials across time in both the fixed and random-effects in a group of hypertensive subjects [49]. Edwards demonstrated that a 9th order polynomial was the best fit to the data. Although a 9th order is unrealistic, Edwards correctly argues that using high order polynomials

is a valid method to smooth individual trajectories eliminating excess "noise" and providing researchers the ability to construct additional measures of 24h ABPM.

Splines – Cubic and restricted cubic splines

Another approach commonly used for modelling non-linear data involves the use of regression splines. Polynomials and regression splines are intrinsically related. Regression splines involve splitting a continuous covariate (usually time) into separate sections. Within each section a polynomial is fitted, and the fitted values from each section are then connected to form a smooth curve [93]. The points where these meet are known as knots. To ensure that there are smooth connections at these knots, certain conditions are imposed. In the instance when there are no knots, the function is a special case of regression splines and the function reverts to being a simple polynomial model. When a cubic polynomial is fitted within each interval it is known as a cubic regression spline with the constraint that it is has continuous first and second derivatives at each knot point. This is the most common type of spline but there is an array of splines now available in different software packages including B-splines, penalised splines, natural splines, thin-splines and smoothing splines, all of which have the capacity to produce well-fitting curves to non-linear data [105].

In the context of ABPM, Selwyn *et al.* used a hierarchical model incorporating a cubic spline model with eight knots [106] to model mean BP profiles. Lambert *et al.* expanded on this by incorporating restricted cubic splines to model the mean BP profiles [48]. Restricted cubic splines are an extension of regression splines with a constraint that the tails are linear. An unrestricted regression spline will have four more parameters to estimate than that of a restricted spline with the same number of knots. Figure 2-4 illustrates a cubic and restricted cubic curve. The assumption of linear tails is suggested as often there are sparse data at tails and suggesting that a polynomial fits this data best can be dangerous or unrealistic.

Both polynomials and cubic splines, by their nature, have the ability to produce well-fitting curves to the data. Increasing the degrees of freedom/knots that are added to these models inevitably increases the fit but can lead to over fitting of the

data. The main disadvantage of polynomials and splines is that the corresponding coefficients are challenging to interpret directly. Moreover, interpretation becomes more difficult with increasing complexity in the model when the number of parameters increases. Also, when the current sample data is over fitted it can also lead to difficulty in extrapolating to other data. However, using these methods to obtain smoothed curves from which new ABPM measures can be extracted is one practical way of using them. Although this approach was not considered as a separate chapter in the thesis, this method was applied to the dataset using a 6th order polynomial (see Appendix A for details and results). Briefly, the maximum, minimum, number of minutes spent above certain hypertensive guidelines and variation about the curve (residual sums of squares) were calculated as measures of variability from the subject-specific fits from the polynomial random-effects model. The association between the extracted BPV measures and the presence of microalbuminuria was assessed using logistic regression with adjustment for age, sex, smoking status, BMI, diabetes and antihypertensive treatment. Additional models adjusted for mean BP. We found all measures were significantly related to microalbuminuria but the association did not persist after additional adjustment for mean BP.



Figure 2-4 Illustration of cubic and restricted splines

Splines – Piecewise Linear Splines

In addition to the previous approaches a method that allowed direct meaningful interpretation of regression coefficients was a requisite of this thesis. Piecewise linear splines in a mixed-effects model were explored as an alternative approach for modelling ABPM. This approach can be seen as a simplification of the principle idea of smooth regression splines where a BP pattern is broken into different segments of time over the 24h period. Rather than fitting polynomials between knot points however, linear splines (straight lines) are used to connect knots. These simple spline models are known as piecewise linear or "broken-stick" models. The benefit of this simple modification is that coefficients represent something meaningful, in this case the slope of BP at different periods of the day. Applying piecewise linear splines in a mixed-effects model allows us to obtain new measures of short-term BPV; the variation about an individual's trajectory and also the variation in slopes

between individuals. This method also allows slopes at a group level to be easily compared. Although linear splines have been used extensively in growth models [107, 108] and even specifically to model BP change over years and gestational age [109], using them to explore daily patterns of BP represents a novel application for analysing ABPM data.

A full analysis using piecewise linear splines on ABPM curves is presented in Chapter 5 but a brief illustration is given here. For each individual m linear splines can be created, where the k^{th} spline:

$$s_{k}(t) = 0 \text{ if } t \le t_{ki}$$

$$s_{k}(t) = t_{i} - t_{ki} \text{ if } t_{k} < t \le t_{ki+1} \qquad \text{for}$$

$$k=1,...,m$$

$$s_{k}(t) = t_{ki+1} - t_{ki} \text{ if } t > t_{ki+1}$$
(14)

These can easily be incorporated into the linear mixed effects model outlined previously:

$$BP_{ij} = (\beta_0 + b_{0i}) + \sum_{k=1}^{m} (\beta_k + b_{ki})s_k + e_{ij}$$
(15)

Figure 2-5 illustrates an example of piecewise linear splines on individual ABPM curves. As Howe *et al.* explain there are a number of methods to determine the number and position of knot points [91]. One option is to place knot points at centiles of the distribution of the x-axis (time in this case), or implement a stepwise regression to select knots where there is statistical evidence of a difference between slopes either side of the knot point [110]. Obtaining a smooth curve for the data and extracting derivatives of this curve represents another approach to help inform the number and positions of knot points [111].

Crucially Howe *et al.* argue that subject knowledge of the underlying physiological process being examined can help the choice of knot point positioning [91]. Considering the circadian nature of BP, this is especially relevant for the analysis of

ABPM where we try to incorporate prior known characteristics of BP, see Chapter 5. Alternatively, the data itself can help decide knot positions [91]. There is no perfect way to determine the number and position of knots but in order to keep the number of parameters that have to be estimated to a minimum it is important to try and have as few knots as possible. Again, similar to picking the correct covariance structure, it is about achieving a compromise between over fitting and under-fitting the model which results in a parsimonious model.

One issue with these methods is there is no constraint to force the model to be periodic. In Chapter 5, a constraint is introduced and presented for a piecewise linear model that ensures periodicity, so that on the average subject-specific BP is the same 24h later.



Figure 2-5 Illustration of piecewise linear splines

Cosinor Analysis

Cosinor analysis, which was first developed by Halberg [112, 113], has been the most common approach to modelling 24h BP [114-118]. The single-component cosinor uses a cosine function as a model for physiological processes that have a circadian rhythm:

$$f(t) = M + A\cos\left(\frac{2\pi t}{\tau} + \emptyset\right) + e_t$$
(16)

where *M* is the MESOR (Midline Estimating Statistic Of Rhythm, the average value over the period), *A* is the amplitude (half the difference between the highest and lowest values, or the distance between the MESOR and the highest (lowest) value), τ is the period or duration of one cycle, ϕ is the acrophase (a measure of the time of the overall high values recurring in each cycle) and *e* is the error term. Incorporating a 24h period and rewriting the equation in a linear form gives:

$$f(t) = M + \beta_1 \cos\left(\frac{2\pi t}{24}\right) + \gamma_1 \sin\left(\frac{2\pi t}{24}\right) + e_t$$
(17)

where the amplitude and acrophase can be obtained:

$$A = \sqrt{\beta_1^2 + \gamma_1^2} \tag{18}$$

$$\phi = \arctan\left(\frac{-\gamma}{\beta_1}\right) \tag{19}$$

A graphical representation of this model fitted to raw data for an individual can be seen in Figure 2-6. Cosinor can be incorporated into a mixed-effects model. However, the majority of the studies to date exploring the use of sinusoidal functions have used fixed-effects models where the inference is on population effects [114-118]. For example, typical inferences are based on estimated differences in model parameters between particular groups of patients, such as comparing the estimated amplitude or MESOR between-groups (fixed-effects) of individuals on different antihypertensive agents [116]. The single component cosinor has a number of limitations. It has been suggested that this method imposes too many restrictions on the shape of the profile and has been shown to fit real profiles poorly [119]. Wang *et al.* [120] suggest problems with fitting a sinusoidal function to a circadian pattern include (i) that the pattern over 24h may not be symmetric; that is, the peak and nadir may not be separated by 12 hours and/or the amplitude and width of the peak may differ from those of the nadir, (ii) sometimes there are local minimum and maximum points.



Figure 2-6 A simple cosinor fit to observed SBP readings of an individual (period=24h)

Additionally Wang *et al.* [121] suggests that the sinusoidal function is too restrictive and "rhythms with a shape closely approximating a cosine curve are uncommon" [122]. The method has some advantages however; a BP curve can be described with the use of only three parameters that have been shown to be related to clinical markers. Also, inherent in the model is the assumption of periodicity – that BP on the average is the same 24h later. Extension of the simple cosinor model to include multiple sin and cosine terms result in more flexible models and overcome many of the disadvantages stated above. Although not as common as the single-component cosinor model, attempts to extend the model by including multiple cosine terms (Fourier analysis) allow more flexible curves to be obtained which has previously been implemented on BP data [123-125]. This is illustrated in Figure 2-7 where a complex curve (red curve) can be broken up into a linear combination of sines and cosines (blue curves). The main advantage of this method is that many complex curves can be obtained by including more terms. However, similar to the singlecomponent model, the majority of studies to date exploring the use of multiple terms have only focused on fixed-effect models [123-125]. A key feature of ABPM analysis however is the exploration of subject-specific effects (random-effects) where its use in cosinor models has been limited [126]. The fitting of a multiplecomponent cosinor random-effects model is outlined in detail in Chapter 6. We determine how the rate of change or "morning surge" changes over time by using first-order derivatives of the multiple-component cosinor model. To the best of our knowledge this is the first use of the cosinor model to determine a measure of morning surge.



Figure 2-7 Fourier series – a linear combination of sines and cosines can produce a complex curve [127].

Functional Principle Component Analysis (FPCA)

Another possible approach to help identify patterns in BP that builds on the idea of combining multiple curves outlined in the previous section is a method known as functional data analysis (FDA), specifically FPCA [128, 129]. FDA refers to an advanced methodology that consists of a set of techniques designed for the analysis and smoothing of curves or functions which makes it appealing for analysing repeated measures data [129, 130]. In addition one of the major advantages of FDA is it does not make any a priori assumption about the curves. Often FDA is applied to data with high sampling frequency such as accelerometers, which takes readings as regularly as every second of a day. The initial aim is to replace the original observations with curves or functions which are then used for further analysis [131]. This is achieved by smoothing the data using a set of building blocks ϕ_{k} , k=1,...,K called basis functions, which are combined linearly [129]. A function x(t) can be expressed as a base function expansion:

$$x(t) = \sum_{k=1}^{K} c_k \phi_k(t) = \boldsymbol{c}' \phi(t)$$
(20)

where c_1 , c_2 ,..., c_k are the coefficients of the expansion, the matrix expression in the last term uses **c** to stand for the vector *K* coefficients and ϕ to denote a vector of length K containing the basis functions. There are two main types of basis systems; Fourier series and splines. Spline bases are more flexible and complex than Fourier series [129]. There are many types of splines but B-splines are typically used. Although Fourier series are generally used for periodic data, splines are flexible enough to capture a periodic cycle as long as the order of the spline is large enough [129].

To simplify, FPCA can be considered an extension of traditional principle component analysis (PCA) to the case of time series data. For example, rather than having separate variables measured on individuals and performing a PCA on these, as in the case of standard analysis, we instead have one variable (BP) measured multiple times for each individual which creates a time series or "function" over time. Each time-point can be essentially considered a variable similar to multivariate PCA. As in multivariate statistics, eigenvalues of the bivariate variancecovariance function are indicators of the importance of the principle components and plotting them, known as a scree plot, can help illustrate how many are necessary to produce a reasonable summary of the data [129]. Rather than an eigenvector, in FPCA there is an eigenfunction associated with each eigenvalue. The resulting output is also similar to that of regular PCA with the only difference being the focus is on functional data where we can identify the main forms of variability in curves [130]. The eigenfunctions describe the major variational components. Similarly, FPCA allows for each feature to be expressed in terms of its percentage of explained variance. FPCA is the most flexible approach outlined as it tries to estimate the "optimal" functions that explain the variability in the data without pre-specifying a particular pattern. As opposed to saying the pattern follows a certain function e.g. cosine function, it is a data driven process where we do not pick the form of the functions. It can be shown that the random function x(t)is $x(t) = \mu(t) + \sum_{k=1}^{\infty} \xi_k \phi_k(t)$, where $\mu(t) = E\{x(t)\}$ and $\xi_k = \int_0^1 \{x(t) - \xi_k\}$

 $\mu(t)$ $\{ \phi_k(t) dt \ \text{are uncorrelated random variables with mean zero and variance } \lambda_k$ (eigenvalues) [132]. These random variables are the principle component scores or loadings.

As mentioned, FDA has traditionally been implemented on high sampling frequency data (e.g. every second) but when there are fewer data points, several FDA methods become inefficient [133, 134]. In the case of sparse data, such as ABPM, Yao et al. developed a version of FPCA in which functional principle component (FPC) scores are framed as conditional expectations which is referred to as principal component analysis through conditional expectation (PACE) for longitudinal data [134]. In essence this approach is similar to a random-effects model where a mixed model framework is used to estimate curve-specific scores and variances. Unlike traditional FDA where curves are analysed and smoothed separately first, this method first ignores the hierarchical structure of the data and a smooth curve is fitted to the pooled data [133]. This estimate of the mean curve is then used to obtain the covariance matrix of the deviations from the mean for each pair of timepoints. This covariance matrix is smoothed using a bivariate smoother and the main diagonal is removed. This smoothed covariance matrix is then decomposed (summarized) into a linear combination of orthogonal (uncorrelated) eigenfunctions and eigenvalues i.e. into its principal components and scores [133]. In the context of random-effects models these FPCs can be seen as patterns of within-subject variation remaining after the mean fit. The first FPC summarizes the main pattern of deviations from the mean trajectory. The second FPC which is uncorrelated to the first explains the next main pattern of variation from the mean and so on. Therefore, a linear combination of a few efficient functions can account for high proportion of the variance. The weights that define the optimal fit to each function are the principle component scores. These can be used to obtain individual BP curves by multiplying the weights by the functions. Zero scores for an individual would result in their trajectory following the mean pattern. The scores would usually be estimated through numerical integration but with sparse data, the approximation is sometimes deemed inadequate and in this case the scores are estimated by the PACE method [133, 134]. Using this method the scores are

58

estimated for each individual using their repeated measures while borrowing strength from the cohort with sample estimates of the mean function, covariance, eigenvalues and eigenfunctions [133, 134].

Although similar, this is not the same as a traditional random-effects model. FPCA will be inherently more flexible. For example, if we consider a two-component cosinor random-effects model as outlined in the previous section – it would assume a mean and two cosine terms with two different periods, FPCA however will try to estimate the "optimal" functions that explain the variability in the dataset. However, if the mean, a one-period cosine and two-period cosine are close to optimal then the rest of FPCA and the model fitting will be similar: random-effects will be used to estimate the subject-specific coefficients for each of these functions. To account for missing data, smoothing (denoising) and interpolation of the ABPM data is incorporated in the FPCA instead of a pre-processing step. Missing data can be problematic for pre-defined smoothing. To elaborate the process, we first estimate the mean and principle component basis functions (common to all ABPM profiles) by using all the available data across subjects. Secondly, we estimate subject-specific curve loadings by using the data available for that curve, and we combine them with the mean and basis functions to estimate the curve over the full domain. The issue of missing and sparse data is accounted for in this approach.

The implementation of FPCA on ABPM as described above is outlined in Chapter 6 and is compared to the multiple-component cosinor model. The rationale is that by using B-spline basis functions it allowed the FPCA to detect the primary patterns in the data without exactly specifying a model. In comparing the resulting individual FPC scores with the cosinor parameters using the individual random-effects estimates it allowed us to determine the adequacy of our proposed simpler multiple-component cosinor model. If the FPC scores from the more flexible approach correlated well with the cosinor model it would suggest that the model was close to the "optimal" model determined by the FPCA and would provide support for the use of our model. In addition, the use of FPCA enabled us to determine the main patterns of variation in the data and the percentage of the total variation that each principle component contributed. This offers a novel

59

method to describe variation within BP that to our knowledge has not been used before on ABPM data.

Other Possible Approaches

On occasion, there have been other attempts to model ABPM using different approaches than those outlined in the previous sections. Degaute et al. attempted to quantify the overall 24h variation by building a best-fit curve based on periodogram calculations [135, 136]. The method is quite similar to a Fourier analysis approach but is optimized for unevenly time-sampled data where troughto-peak values can be obtained along with an acrophase parameter. Another method that has been purposed is the cumulative sums (cumsums) method [137]. This method draws a reference line across an individual's pattern at their mean value, and successive deviations of the data from that line are then summed and plotted. It is argued changes in the cusums reveal changes in the trend of data from the baseline much more sensitively than do the data themselves [137]. It is argued that the values obtained are not restricted to follow a symmetrical pattern or fixed period of time e.g. sleep or awake times. The resulting parameters include the crest and trough of BP which are similar to the values that can be obtained using the amplitude in cosinor analysis. However, multiple-component cosinor is more flexible and offers more parameters that describe the curve.

Another approach proposed to model ABPM trajectories was a method used in growth curve analysis literature. Beath introduced a method of data reduction to simplify comparisons between individual growth curves by describing a shape invariant model of infant weight [138]. Cole *et al.* extended this to create the SITAR model (SuperImposition by Translation and Rotation) [139, 140]. The idea is that there is an underlying mean curve which can be applied to all subjects and by applying just three subject-specific translations and rotations of the curve, individual trajectories can be obtained [140]. In the context of BP and mixed-effects, the model is:

$$BP_{it} = \alpha_i + h\left(\frac{t - \beta_i}{\exp(-\gamma_i)}\right)$$
(21)

where BP_{it} is BP for subject *i* at time *t*, h(t) is a natural cubic spline curve of BP vs time, and α_i , β_i and γ_i are subject-specific random-effects. Essentially the three parameters shift the curve up or down (α), left or right (β), and stretch or squash the time-axis (γ) [91]. Although the model has primarily been developed for examining data over years (growth curves), it does not mean it is not suitable for examining non-linear circadian data. Cole *et al.* recently applied the model to growth curves of insulin-like growth factor 1 (IGF-1) [141]. Unlike height for example, IGF-1 follows a curvilinear shape that does not stop when it reaches a maximum and is similar in shape to a polynomial (during adolescent years). Despite the non-linear pattern of the data Cole *et al.* found that the model explained 65% of the variance in IGF-1 [141].

A SITAR model was applied to our ABPM dataset. The model summarizes the set of BP curves with a mean BP curve as a cubic regression spline, plus the three fixed and random-effects (α_i , β_i and γ_i) defining how individual BP curves differ from the mean curve. Only these three parameters were random, with the spline terms fixed. The fitted models used regression splines with seven degrees of freedom. Figure 2-8 shows the raw ABPM data plotted against time along with the data replotted after adjustment was made for the subject-specific parameters (α_i , β_i and γ_i).

The plot indicates that although it is difficult to see a pattern in the raw data, after adjustment a somewhat clearer image can be seen. It suggests that adjusting for these three parameters can partially explain some of the variation in BP. These random-effects could then potentially be used as pseudo variability measures and applied in further models to explore their prognostic significance. One disadvantage of the method is that it is not periodic and additional work would need to be done to make it so. Furthermore the interpretation and clinical meaning of the three parameters that are obtained from the model are not instantly clear. However, if we compare the SITAR to the single-component cosinor model and their parameters, similarities can be seen between both models. They both essentially have parameters that represent a mean and phase shift (left to right). The SITAR also has a stretching parameter which is not directly equivalent but is similar to the

61

period in the cosinor which is fixed. Although not immediately obvious, with a little manipulation, the $exp(-\gamma_i)$ term could be considered similar to that of the amplitude parameter. The main difference between the models is that one uses a cubic spline (SITAR) and the other uses a cosine (cosinor) as a function of time. Although the parameters from the SITAR model may not seem that interpretable on first glance, they do have similarities to the cosinor parameters. However, implementing a cosinor analysis over a SITAR model seemed more appropriate for the analysis of BP data as the parameters are directly relevant and easier to understand with the advantage that the model is naturally periodic.



After SITAR adjustment

Figure 2-8 Illustration of the SITAR model for SBP over 24h. The black line indicates the mean BP curve.

Comparison of Potential ABPM Models

A number of different approaches to capture circadian BP patterns have been outlined and choosing a model amongst them is challenging. The ultimate goal remains the development of a physiological model that captures known features of the pattern while not over-fitting the data so that we can obtain clinically relevant parameters [142]. It is widely acknowledged that achieving this is a difficult task and there is no perfect model that captures all the features of the circadian rhythm simultaneously [49, 142-144]. To help compare models, Table 2-1 considers the main features that are deemed important in deciding on a final ABPM model. It reiterates the point that no model is optimal on all of the suggested features and choosing one involves making a compromise based on what the researcher deems most important for their particular research question.

We believed the two most important features that should influence our decision on model choice were flexibility and ease of parameter interpretability (which included clinical relevance). Flexibility referred to the ability of the model to capture the circadian pattern in the data. Although, as highlighted previously, summary measures such as SD are the crudest form of analysis, their ease of interpretation and widespread use (often included in standard ABPM reports) made their inclusion in the thesis mandatory (Chapter 3 and 4). By their nature polynomials and splines are flexible and can provide good fits to the data that can be particularly useful when exploring data. However, they have the problem that their resulting coefficients are not directly interpretable (excluding piecewise splines). For this reason they were not considered as a separate chapter in the thesis. Piecewise linear splines on the other hand are more interpretable and depending on the number of knots included can offer a flexible fit to the data. In addition, to the best of our knowledge the method has not been applied to ABPM data before thus offering novel measures of BPV (Chapter 5).

Considering the traditional single-component cosinor model has been the most commonly used approach for the longitudinal analysis of ABPM, further research in an attempt to extend the model seemed warranted for inclusion in this thesis. The

64

single-component model has the advantage of parameters being easy to understand but the disadvantage that the shapes available are very restricted. Although a multiple-component model is not as interpretable as a singlecomponent model, it is still quite intuitive compared to spline models and their coefficients. The model also had the benefit of being periodic. For these reasons the exploration of a multiple-component cosinor model seemed reasonable (Chapter 6).

As outlined, FPCA is the most flexible approach and tries to estimate the "optimal" functions that explain the variability in the data without pre-specifying a particular pattern. Given a fixed number of functions, it will find the optimal fit. This was used in conjunction with the cosinor method to determine how well the model identified patterns in the data (Chapter 6).

Periodogram which has rarely been used is almost identical to Fourier analysis and was not considered due to the similarity with the resulting output from a cosinor analysis. Similarly, the method of cumulative sums has rarely been used and was not considered in the thesis as the resulting output is too similar to that of cosinor amplitude values. The SITAR model has never been used on BP data but does have similarities to a single-component cosinor model. We felt that the cosinor method gave similar output but with easier interpretation and for this reason the SITAR was not considered in a full analysis.

| Modeling Technique | Flexibility | Implemented on ABPM | Naturally Periodic | Ease of Parameter Interpretability |
|--------------------|-------------|---------------------|-----------------------|---------------------------------------|
| Summary Measures | 1 | Often | No | Excellent |
| Polynomials | 2 | Sometimes | No† | Poor |
| Splines | | | | |
| Cubic | 3 | Rarely | No† | Poor |
| Piecewise | 2 | Never* | No† | Excellent |
| Cosinor | | | | |
| Single-component | 1 | Sometimes | Yes | Excellent |
| Multiple-component | 3 | Rarely | Yes | Good |
| FPCA | 3 | Never* | No | Not Implicitly known‡ |
| SITAR | 2 | Never* | No | Poor |
| Periodogram | 2 | Rarely | Yes | Good |
| Cumulative Sums | 2 | Rarely | No | Good |

Table 2-1 Comparison of different modeling approaches for the analysis of ABPM data.

Flexibility rating: 1,2, 3, where 1 is the least flexible and 3 represents the most flexible; Implemented on ABPM scale: Never, rarely, sometimes, often; Interpretability scale: Poor, good, excellent

*not to the best of our knowledge, †not directly but with manipulation they can be forced to be periodic, ‡Depending on the ease of interpretation of the FPCs – this won't be known until the results are explored

3. CORRELATION BETWEEN SHORT-TERM BLOOD PRESSURE VARIABILITY AND LEFT VENTRICULAR MASS INDEX: A META-ANALYSIS (PAPER 1)

J. M. MADDEN^a

A. M. O'FLYNN^a

A. P. FITZGERALD^{a,b}

P. M. KEARNEY^a

^a Department of Epidemiology & Public Health, University College Cork, Ireland

^{a,b} Department of Statistics, University College Cork, Ireland

This paper was published in Hypertension Research in 2015 (See Appendix J)

Another search was completed with no new papers identified (15th Sept 2016)

3.1 Abstract

Long term BPV has been associated with cardiovascular events but the prognostic significance of short-term BPV remains uncertain, including its influence on the presence of TOD, specifically LVH. A meta-analysis exploring the correlation between short-term BPV and LVMI was performed. Studies were identified by systematic searches in Pubmed and EMBASE. Any summary measure of short-term BPV obtained from ABPM was included. Twelve studies were included. SD, ARV, wSD and CV across 24h/day/night periods were identified as measures of variability. Meta-analysis showed the pooled subgroup correlation coefficients of LVMI with 24h SBP SD, day SBP SD, wSD SBP and 24h ARV SBP were 0.22 (95% CI: 0.12-0.31), 0.19 (95% CI: 0.15-0.25), 0.23 (95% CI: 0.13-0.33), 0.37 (95% CI: 0.01-0.65) respectively. This meta-analysis suggests there is a weak positive correlation, between BPV and LVMI.

3.2 Introduction

Hypertension is a well-established risk factor for CVD [3, 145]. To date guidelines on the management of hypertension have focused on reducing mean BP, which is clearly important, but don't mention BPV [17], for which there is increasing evidence of prognostic value. Evidence from meta-analyses suggest that although different antihypertensive-drug classes have similar effects in terms of reducing BP levels, pronounced differences in their ability to reduce BPV are observed [61, 62]. These differences additionally accounted for effects on stroke risk independent of mean BP. Studies have also shown that SBP variation from one visit to the next may be associated with a poor cardiovascular prognosis. In treated hypertensive patients enrolled in ASCOT-BPLA (Anglo-Scandinavian Cardiac Outcomes Trial-Blood Pressure Lowering Arm), higher visit-to-visit variability in SBP was associated with stroke and coronary events independent of mean BP [25]. In a population based observational study higher visit-to-visit variability in SBP was associated with increased mortality risk over a 14-year follow-up [38]. Importantly visit-to-visit BPV predicted all-cause mortality among those with normal BP, suggesting it may be a prognostic marker before hypertension develops.

Short-term BPV refers to fluctuations of BP across minutes or hours usually taken over a 24h period and can be obtained through the use of ABPM [146]. The predictive value of short-term BPV is less well established than that of visit-to-visit variability. Hansen *et al.* [33] using a large population cohort (8938 subjects) explored the relationship between BPV recorded at base line with cardiovascular events over a median period of 11.3 years and determined that although shortterm reading-to-reading BPV was an independent predictor, it did not contribute significantly to risk stratification over and beyond 24h BP. Evidence from the ASCOT-BPLA trial which included both long and short-term variability suggests that although not as strong a predictor as visit-to-visit BPV, short-term BPV measured by the CV still predicted risk of vascular events independently of average daytime mean SBP [25].

69

The occurrence of major cardiovascular events is usually the result of long-term exposure to hypertension and other risk factors and is often preceded by the development of asymptomatic functional and structural abnormalities known as TOD [75]. Little is known about the influence of short-term BPV on the presence of TOD, specifically LVH. LVH can be determined by ECG or quantified more accurately by measuring left ventricular mass by echocardiography and indexing this to body surface area to give the LVMI [147]. In their seminal paper Parati et al. [27] demonstrated that higher diurnal BPV measured as 24h SD was associated with an increased risk of LVH (determined by ECG) in 108 mild-to-severe essentially hypertensive patients. They also showed that for nearly any level of 24h mean BP, subjects in whom the 24h BPV was low had a lower prevalence and severity of TOD those in whom BPV was high, indicating an independent association. However as highlighted evidence since suggests the predictive value of short-term BPV remains unclear and may not contribute much more than mean levels alone [33]. To advance our knowledge of short-term variability, this review attempts to assess and quantify the correlation between BPV and LVMI. A meta-analysis on the various correlation coefficients will be performed.

3.3 Methods

Types of studies

Cohort, cross-sectional or case-control studies that explored the relationship between 24h BPV and LVMI.

Study populations

Participants recruited to observational studies that underwent 24h non-invasive ABPM and an assessment of LVMI. Studies of pregnant women and children were excluded.
Predictor variables

Any summary measure of short-term BPV, where short-term refers to variations across minutes or hours taken over a 24h period obtained by non-invasive ABPM. Summary measures refer to those that can be obtained without the need for advanced statistical methods.

Outcomes

LVMI determined with echocardiography.

Search methods for identification of studies

Studies were identified by systematic searches in Pubmed and EMBASE (up to June 2015). The following search terms were used as keywords and/or MESH terms: ((ambulatory blood pressure) OR (blood pressure) OR (ambulatory blood pressure monitoring) OR (short-term blood pressure) OR (24 hour blood pressure)) AND (variability) AND ((left ventricular hypertrophy) OR (left ventricular mass index) OR ((end OR target) organ (damage OR disease))). The full search strategy can be seen in the Appendix B which includes different spellings and combinations of words.

Potentially relevant articles were identified and duplicates were removed. Only original research articles were included. We supplemented our electronic search by crosschecking the reference lists of all identified studies. There were no date or language restrictions. Non-English papers were translated with an online translation programme. The full texts of relevant articles were obtained and an independent reviewer reviewed selected papers against the inclusion criteria and assessed their quality using the guidelines recommended by Hayden *et al.* [148] for quality appraisal in systematic reviews of prognostic studies. Our systematic review and meta-analysis was conducted according to the checklist of Meta-analysis of Observational Studies in Epidemiology (MOOSE), and the Preferred Reporting Items for Systematic reviews and Meta-Analyses guidelines (PRISMA), see Appendix B.

Data Extraction

The study characteristics extracted included sampling approach, study design, sample size, mean age, BPV index and value, mean LVMI, correlation coefficients and relative information, such as p-values and if it was indicated that they were statistically significant or not. The data was extracted independently by two researchers (JMM and AMOF).

Statistical Analysis

For the meta-analysis, correlation coefficients were converted into Fisher's z-scores and standard errors which in turn were used to calculate 95% confidence intervals. The overall effect size was the weighted inverse variance of the adjusted individual effect sizes (z-scores). The overall effect sizes from the meta-analyses were then back transformed which corresponded to the overall correlation coefficients. Data from the various studies were pooled using the random-effects model. Heterogeneity between studies was assessed using the l² statistic. The Begg's test and Egger's test were used to assess the extent of publication bias. All analysis was performed using Stata software [149].

3.4 Results

Basic characteristics of studies

After removal of 218 duplicates a total of 440 articles were identified during the search, of which 416 were excluded based on their titles and abstracts alone, Figure 3-1. After reviewing the remaining 24 full-text articles, 12 were eligible for inclusion in the review, Table 3-1. Reasons for exclusion included articles didn't calculate LVMI as an outcome, summary measures of variability weren't calculated and no effect size was reported. Of the 12 studies, 11 were cross-sectional and one had a case-control design. The population sample sizes ranged from 33 to 1822. The

various indexes used in the studies along with their definitions are presented in Table 3-2. The SD of either 24h/day/night BP readings were used as indexes of BPV in all studies with the exception of two: one which only reported CV [150], and another study [151] only reported average real variability (ARV). In addition to SD, two studies also included CV [39, 152] and a further two included wSD [28, 40]. Leoncini *et al.* [28] also explored ARV. The average value of 24h SBP SD, day SD, night SD had range 13.0-19.7, 10.9-19 and 11.5-13.6 mmHg respectively. As there were so few studies exploring the other indices we have not reported their range here but can be found in Table 3-3. The correlation between 24h SBP SD, day SD, night SD and LVMI had range 0.05-0.52, 0.13-0.21 and 0.04-0.21 respectively. These correlations were all statistically significant with the exception of day SD (r=0.19) [153], 24h SBP SD (r=0.05) and night SD (r=0.04) [40]. In the three studies in which it was explored, wSD had a statistically significant correlation of r= 0.15, 0.26 and 0.31 [28, 30, 40]. Similarly in the two studies which examined 24h SBP ARV, a statistically significant correlation of r= 0.15, 0.26 and 0.31 [28, 30, 40]. Similarly in the two studies which examined 24h SBP ARV, a statistically significant correlation of 0.53 and 0.19 with LVMI was observed [28, 151].

Of the studies that adjusted for covariates (including mean BP), findings were mixed. Schillaci *et al.* [34] who considered 1822 untreated subjects with essential hypertension, reported a weak univarite statistically significant correlation between daytime and night-time SD and LVMI but the association didn't persist after adjustment for various confounders. Similar findings were found by Roman *et al.* [39] who found daytime and night-time SD were univariately associated with LVMI but the association did not persist after adjustment for confounders including average BP. Pascual *et al.* [154] also found similar results after adjustment for age, sex and mean BP.

In contrast, Tatasciore *et al.* [30] in a study examining 180 untreated hypertensive patients, found daytime SD and wSD to be statistically significantly associated with LVMI even after adjustment for other covariates, including mean BP. Similarly Bilo *et al.* [40] found wSD was statistically significantly related to LVMI in a study which investigated 339 hypertensive patients. Zhang *et al.* [151] also found 24h ARV to be significantly related to LVMI after adjustment.

Using the guidelines recommended by Hayden *et al.* [148], the quality appraisal of each paper was assessed and is presented in Table 3-4.

Meta-analysis

Figure 3-2 presents converted correlation coefficients (z-scores) with subgroup meta-analysis reported for each BPV index. An overall z-score for all studies was omitted as combining different indexes would not be appropriate. After conversion from z-scores the pooled subgroup correlation coefficients of LVMI with 24h SBP SD, day SBP SD, wSD SBP and 24h ARV SBP were 0.22 (95% CI: 0.12-0.31), 0.19 (95% CI: 0.15-0.25), 0.23 (95% CI: 0.13-0.33), 0.37 (95% CI: 0.01-0.65) respectively. All but one index (wSD) showed heterogeneity (p<0.05) across the studies and as a result random-effects models were used to combine coefficients. Begg's and Egger's tests indicated no evidence of publication bias within each variability index.

3.5 Discussion

Overall our review suggests that there is a weak positive correlation, between BPV and LVMI. We carried out a separate analysis for each measure of variability, resulting in reduced power in the meta-analysis. Our review highlights the lack of good epidemiological studies exploring the relationship between BPV and LVMI. As eleven of the twelve studies were cross-sectional, we cannot assess cause-effect relationships. Although all studies reported univariate coefficients we found just over half of the studies did any further analysis or appropriate adjustment for covariates. Despite these limitations, the results are still worth exploring and the review raised some important issues in relation to BPV in general and also specifically to LVMI.

Veerman *et al.* [153] reported a non-statistically significant correlation with day SD. We cited the small sample size (n=33) as a potential reason for the discrepancy between day SD compared to the other studies. Bilo *et al.* [40] also reported a nonstatistically significant correlation of LVMI with 24h SD but interestingly in the same study found both day SD and wSD were significantly correlated with LVMI even

after adjustment. This finding highlights that results are sensitive to the index chosen and leads to the issue of variability measurement.

Most studies have used SD as a measure of BPV and the appropriateness of such an index has been disputed because it only reflects the dispersion of measurements around a single value (mean) not accounting for the order in which BP measurements were obtained [36, 37]. The discrepancies between day and 24h SD in the study by Bilo et al. [40] may be explained by the fall of BP at night (dip). A large dip which is known to be associated with healthier individuals will lead to a larger 24h SD. The wSD attempts to remove the effect of the dip and was found to be significantly correlated unlike the 24h SD. This suggests that perhaps SD, at least over 24h may not be a good measure of BPV. Mena et al. [36] first explored, and later Pierdomenico et al. [37], ARV in relation to BP which is the average absolute difference between successive readings, and is thought to give a true reflection of real variability. In both studies high ARV was found to be an independent predictor of cardiovascular risk in hypertension patients while high SD was not. The two studies that included ARV in this review both found a statistically significant correlation with LVMI [28, 151]. In one study association remained significant after adjustment [151] while the other study found it to be an independent predictor of multiple TOD where the majority of these had LVH [28]. As ARV is thought to give a true reflection of real variability it may be the most appropriate marker of short term BPV over other indexes and could potentially be used to predict outcome in patients even before BP becomes elevated and ultimately provide a means of identifying at risk patients before they develop hypertension.

Other studies exploring the relationship between BPV and TOD have found varied results. As mentioned Parati *et al.* [27] found an association between 24h BPV and severity of TOD (a score based on presence of LVH, chest x-ray abnormalities, abnormalities of the fundus plus a clinical event and/or a renal abnormality). The same group conducted another follow-up study with a follow-up period of 7 years to assess the prognostic relevance of short-term BPV on 73 hypertensive patients [155]. They found an independent association between 24h BPV at baseline and TOD at follow-up. Similarly, in another study of over 700 hypertensive and

normotensive patients, daytime systolic SD was found to be associated with degree of TOD. However in the same study, after adjustment for mean BP no strong association was found between BPV and LVH [156]. Hansen *et al.* explored the relationship between BPV and cardiovascular events [33]. ARV predicted all fatal and nonfatal outcomes even after adjustment for mean BP but found that it added only 0.1% to the explained risk of an event occurring. They concluded that the main risk factor remained mean BP.

Another point that must not be overlooked when considering the importance of the results from the individual studies is the issue of statistical and clinical significance. All the studies have decided on the importance of the correlation coefficient based solely on a statistical significance threshold level of 0.05. It is critical to remember that the smaller the p-level the more significant the relationship but the larger the correlation, the stronger the relationship. It is for this reason that we must be careful how we interpret findings from such studies. For example Bilo et al. found a statistically significant correlation between wSD and LVMI but the value of the coefficient was only 0.15 which would be considered a weak association [40]. In this case, with a large sample size (n=339), researchers should not solely focus on statistical significance. Here, we should not over emphasise the statistical significance finding as the correction coefficient is in fact quite low. Sterne and Smith argue that a p-value<0.001 provides much stronger evidence against the null hypothesis, in comparison to a p-value <0.05 and suggest that results of medical research should generally not be reported as "significant" or "non-significant" where appropriate [157]. It is important to take a practical but sensible approach where statistical and clinical significance are both given equal thought when deciding on the importance of a relationship.

As the studies are cross-sectional in nature we are not able to determine whether higher BPV initiates increases in LVMI or do increases in LVMI represent a risk factor for increased BPV rather than being a consequence of it. It is however argued that vascular hypertrophy induced by exaggerated and large BPV may lead to an impaired arterial distensibility of the large arteries, resulting in increased cardiac afterload and as a result increases LVMI [158]. Clinical trials have recently shown

that some classes of anti-hypertensive drugs significantly outperform others in terms of lowering BPV, and that this reduction in short-term and long-term BPV contributes to the prevention of cardiovascular events in hypertensive patients [25, 159]. Results indicate that CCB and to a lesser extent thiazide diuretics are superior to other drugs in reducing BPV and preventing stroke and other vascular events compared to the older β -blocker atenolol which increases BPV [61, 160]. Similar findings were reported in a more recent observational study assessing the efficacy of mono and combination therapy on short-term BPV of 2780 hypertensive patients [64]. Again CCB's, followed by diuretics were correlated with lower short-term BPV compared with ARB's, ACEI's and β -blockers. In addition combination of CCB's and diuretics resulted in the lowest BPV compared to others. In those with marked BPV, the prescribing of these drugs may offer a better alternative and could help reduce the risk of LVH especially in individuals where hypertension has not yet developed.

The major limitation of this review is that we have pooled together studies in a meta-analysis in regard to their correlation coefficients which are a very weak marker of association. As a result of using correlation coefficients there is an implicit assumption that the association between BPV and LVMI is linear which in reality may not be the case. The strength of this review is its focus on short-term BPV which has recently been receiving growing attention. It is also the first review to our knowledge that quantifies the correlation between BPV and LVMI. The review identifies a research gap where stronger epidemiological studies are needed to explore the relationship further and understand the prognostic value, if any, of short-term BPV.



Figure 3-1 Flow diagram of study selection



Figure 3-2 Pooled meta-analysis of z-scores by BPV index.

| Study | Population | Sampling Approach | Design | n (women, %) | Mean age (SD or range), yr | BPV index | Outcome measure |
|----------------------|---|----------------------|-----------------|---|-------------------------------|--------------------------------|--------------------|
| Colivicchi (1996) | Elderly untreated HTN males with matched normotensives | Convenience | Case-control | 50 (0%) | 74 (4) | 24h, day, night - SD | LVMI |
| Veerman (1996) | Referred to hypertension clinic due to suspected hypertension | Convenience | Cross-sectional | 33 (48%) | 41 (26-59) | Day- SD | LVMI |
| Schillaci (1998) | Hospital based untreated HTN | Convenience | Cross-sectional | 1822 (47%) | 50 (12) | 24h, day, night- SD | LVMI |
| Pascual (1999) | Untreated HTN | Convenience | Cross-sectional | 149 (33%) | 38 (7) | 24h, day, night – SD, CV | LVMI |
| Kristensen (2001) | Untreated HTN from general practice and subjects drawn at random from Danish national register | Convenience & random | Cross-sectional | 566 (52%) | 48 (20-79) | 24h, day, night – SD,CV | LVMI |
| Roman (2001) | Subjects from a worksite- based study and who were evaluated at a hospital | NR | Cross-sectional | 511 (44%) | 50 (12) | Day, night – SD,CV | LVMI |
| Polonia (2005) | Population sample | NR | Cross-sectional | 743 (56%) 185 (LVMI measured) | 52 (14) | Day SD | LVMI |
| Tatasciore (2007) | Outpatients referred to clinic by GP | Convenience | Cross-sectional | 180 (40%) | 53 (8) | 24h, day, night – SD wSD | LVMI |
| Bilo (2007) | Two hypertension centres | NR | Cross-sectional | 3863 (54%) 339 (50%) (Echo taken) | 54 (12) | 24h, Day, night – SD, wSD | LVMI |
| Zhang (2011) | Elderly hospitalised HTN and normotensive controls | Convenience | Cross-sectional | 197 (35%) | 76.5 (7.8) | 24h, day, night- ARV | LVMI |
| Ajayi (2011) | Nigerian HTN | Convenience | Cross-sectional | 130 (26%) | 54 (12) (31-85) | 24h – CV | LVMI |

Table 3-1 Study characteristics

| Study | Population | Sampling Approach | Design | n (women, %) | Mean age (SD or range), yr | BPV index | Outcome measure |
|---------------------------------------|--|-------------------|-----------------|--------------|-------------------------------|-------------------------------------|--------------------|
| Continued on fo Leoncini (2013) | llowing page Untreated HTN attending outpatient clinic | Convenience | Cross-sectional | 169 (33%) | 47 (10) | 24h, day, night – SD ARV, wSD | LVMI |

Abbreviations: SD: Standard Deviation, CV: coefficient of variation, wSD: weighted standard deviation, LVMI: Left ventricular hypertrophy index, TOD: target organ damage, LVH: left ventricular hypertrophy, ARV: average real variability, HTN: Hypertensives, GP: general practitioner, NR: not reported

Table 3-2 BPV definitions

| Measure of BPV | Description |
|----------------|---|
| 24h SD | Standard deviation over 24h period |
| Day SD | Standard deviation over day period usually 9am-9pm |
| Night SD | Standard deviation over night period usually 1am-6am |
| wSD | Weighted standard deviation which is the mean of day and night standard deviation values corrected for the number of hours included in |
| | each of the two sub-periods which attempts to eliminate the effect of nocturnal fall. |
| | wSD=((day standard deviation x day hours) + ((night standard deviation x night hours)))/total number of hours |
| CV | Coefficient of variation = (Standard deviation over 24h/mean 24h) x 100 |
| ARV | Average real variability which averages the absolute differences between successive readings which is the average absolute difference |
| | between successive readings. $ARV = \frac{1}{N-1} \sum_{k=1}^{N-1} BP_{k+1} - BP_k $, where k ranges from 1 to N-1 and N is the number of BP measurements |

| Study | SBP Variability SD | Mean LVMI (SD) g/m ² | Multivariate analysis adjusted for: | Method of adjustment for mean BP | Correlation Coefficient (r) | Beta coefficients (multivariate analysis) β (se) |
|----------------------|--|------------------------------------|--|--|---|--|
| Colivicchi (1996) | 16.2 (3.5) 24h SD 19 (5.4) day SD 12 (2.9) night SD | 134.9 (27.5) (LVH=16%) | - | - | r=0.52† 24h SD (HTN only) r=NS day & night SD | - |
| Veerman (1996) | 12.7 (7.5-22) day SD | 115 (67-153) (LVH=13%) | Day BP | Regression | r=0.19 day SBP SD r=0.35* day DBP SD | 3.16 (1.18)* Day DBP SD |
| Schillaci (1998) | NR | NR | Age, body height, 24h BP, DBP, BMI, duration of hypertension, alcohol, smoking | Subjects divided into similar levels of BP, then subdivided if below or above median BPV & separately a regression analysis | (r=0.13, p<0.01) day SBP SD (r=0.1, p<0.01) night SBP SD r=NS day & night DBP SD | NS |
| Pascual (1999) | 14.0 (3.4) 24h SD 10.3 (2.4) 24h CV 10.9 (2.8) day SD 13.8 (3.3) day CV 11.7 (3.5) night SD 11.7 (3.7) night CV | 126 (34.6) | Age, sex, BP | Regression | r=0.23 [†] (24h & day SBP SD) r=NS (24h, day, night DBP SD & CV; 24h, day, night SBP CV & night SBP SD) | 0.87 (.83) 24h SBP SD |
| Kristensen (2001) | NR | 102.7 (28.3) | - | - | r=0.24 (p<0.01) | NR |
| Roman (2001) | NR | NR | Age, age ² , sex, basal SBP or DBP, serum cholesterol, smoking, antihypertensive medication | Regression | r=0.29‡both day & night SBP SD r=0.19‡ DBP SD day r=0.26‡ DBP SD night r=0.08 SBP CV day r=0.022 SBP CV night r=0.00 DBP CV day r=0.06 DBP CV night | NS |
| Polonia (2005) | 13.2 (3.4) day SBP SD 14.3 (3.4) 24h SBP SD 11.5 (3.8) night SBP SD | 100 (40) | - | - | r=0.162* day SBP SD | - |

Table 3-3 Extracted Results

| Study | SBP Variability SD | Mean LVMI (SD) g/m ² | Multivariate analysis adjusted | Method of adjustment for mean | Correlation Coefficient (r) | Beta coefficients (multivariate analysis) |
|----------------------|--|------------------------------------|---|-------------------------------|---|---|
| | | | for: | BP | | β (se) |
| Tatasciore (2007) | 13.0 (4.1) 24h SBP 10.9 (4.0) 24h DBP | 96.7 (17.8) | Age, sex, alcohol, triglycerides, SBP, DBP, SBP load, DBP load | Regression | r=0.312‡ 24h SBP SD r=NS – 24h DBP r=0.310‡ wSD SBP r=NS wSD DBP | 0.633 (p=0.028) 24h SBP |
| Bilo (2007) | NR for echocardiographic group | 109.5 (33.8) | 24h BP, sex | Regression | r=0.05 24h SBP SD r=0.15† 24h wSD r=0.16† day SBP SD r=0.04 night SBP SD | 0.15 (p<0.01) 24h wSD SBP 0.16 (p<0.01) day SBP SD 0.04 (p=NS) 24h SD |
| Zhang (2011) | 11.9 (2.6) 24 ARV SBP 8.6 (2.6) 24 ARV DBP 9.3 (2.3) day ARV SBP 8.2 (2.1) day ARV DBP 9.2 (3.3) night ARV SBP 7.8 (2.1) night ARV DBP | 145.1 (43.2) | Age, duration of HTN, total cholesterol, low- density lipoprotein cholesterol, 24h BP | Regression | r=0.525† 24h ARV SBP r=NS NR 24h ARV DBP | 0.593† 24h ARV SBP |
| Ajayi (2011) | NR | 109.65 (38.1) | - | - | r=0.379* 24h CV SBP r=0.124 24h CV DBP | - |
| Leoncini (2013) | 19.7 (5.9) 24h SD 18.5 (6.3) day SD 13.6 (5.5) night 16.8 (5.3) 24h wSD 15.1 (4.8) 24h ARV 15.9 (5.7) day ARV 13.0 (5.3) night ARV 15.1 (5.0) wARV (All SBP, DBP not displayed) | 46 (11) (LVM g m ^{-2.7}) | Office BP, age, BMI, sex, smoking, triglycerides, cholesterol, glucose, duration of HTN, AASI | Logistic regression | r=0.24 ⁺ 24h SD r=0.19 [*] 24h ARV r=0.26 [‡] 24h wSD r=0.21 ⁺ day SD r=0.21 ⁺ night SD r=0.19 [*] day ARV r=0.14 night ARV all for LVMI indexed for BSA | Odds ratio reported but for overall TOD presence (2 or more) 1.103 (1.003-1.212)† 24h SD 1.114 (1.012-1.227)† day SD 1.140 (1.004-1.295)† 24h ARV |

Abbreviations: SD: Standard Deviation, CV: coefficient of variation, wSD: weighted standard deviation, LVMI: Left ventricular hypertrophy index, TOD: target organ damage, LVH: left ventricular hypertrophy, ARV: average real variability, HTN: Hypertensives, NR: not reported, BP: blood pressure, SBP: systolic blood pressure, AASI: Arterial stiffness index based, NS: non-significant and no value was reported, *p<0.05; †p<0.01; ‡p<0.001

| Study | Study Participation | Study Attrition | Prognostic | Factor | Confounding | Measurement | and | Analysis |
|-------------------|---------------------|-----------------|-------------|----------|-------------|-------------|-----|----------|
| | | | Measurement | /Outcome | Account | | | |
| | | | Measurement | | | | | |
| Colivicchi (1996) | No | No | Yes | | No | | | No |
| Veerman (1996) | Partly | No | Partly | | No | | | No |
| Schillaci (1998) | Yes | Partly | Yes | | Yes | | | Yes |
| Pascual (1999) | Partly | Partly | Partly | | Partly | | | Partly |
| Kristensen (2001) | Partly | No | Yes | | No | | | No |
| Roman (2001) | Partly | No | Partly | | Partly | | | Yes |
| Polonia (2005) | Partly | No | Partly | | No | | | No |
| Tatasciore (2007) | Yes | Yes | Yes | | Yes | | | Yes |
| Bilo (2007) | Partly | Partly | Partly | | Partly | | | Partly |
| Zhang (2011) | Partly | No | Yes | | Yes | | | Partly |
| Ajayi (2011) | Partly | No | Yes | | No | | | Yes |
| Leoncini (2013) | Partly | Yes | Yes | | Partly | | | Partly |

Table 3-4 Quality Assessment

4. SHORT-TERM BLOOD PRESSURE VARIABILITY OVER 24 HOURS AND TARGET ORGAN DAMAGE IN MIDDLE-AGED MEN AND WOMEN (PAPER 2)

J. M. MADDEN^a

A. M. O'FLYNN^a

E. DOLAN^c

A. P. FITZGERALD^{a,b}

P. M. KEARNEY^a

^a Department of Epidemiology & Public Health, University College Cork, Ireland

^{a,b} Department of Statistics, University College Cork, Ireland

^c Stroke and Hypertension unit, Connolly Hospital, Blanchardstown, Dublin, Ireland

This paper was published in Journal of Human Hypertension in 2015 (See Appendix J)

4.1Abstract

BPV has been associated with cardiovascular events but the prognostic significance of short-term BPV remains uncertain. As uncertainty also remains as to which measure of variability most accurately describes short-term BPV, this study explores different indices and investigates their relationship with subclinical TOD.

We used data from the Mitchelstown Study, a population based study of Irish adults aged 47-73 years (n=2,047). A subsample (1,207) underwent 24h ABPM. As measures of short-term BPV we estimated the SD, wSD, CV and ARV. TOD was documented by microalbuminuria and ECG LVH.

There was no association found between any measure of BPV and LVH in both unadjusted and fully adjusted logistic regression models. Similar analysis found ARV (24h, day and night), SD (day & night) and wSD were all univariately associated with microalbuminuria and remained associated after adjustment for age, gender, smoking, BMI, diabetes and anti-hypertensive treatment. However, when the models were further adjusted for mean BP the association did not persist for all indices.

Our findings illustrate choosing the appropriate summary measure which accurately captures short-term BPV is difficult. Despite discrepancies in values between the different measures there was no association between any indexes of variability with TOD measures after adjustment for mean BP.

4.2 Introduction

The prognostic value of BPV in addition to mean BP has been receiving growing attention [161, 162]. In addition to showing a strong association with increased mortality risk, visit-to-visit variability in BP has been shown to predict stroke and coronary events independent of mean clinic BP [25, 38]. It has also been linked with TOD [38, 163, 164]. However, the predictive value of short-term BPV, i.e. fluctuations of BP across minutes or hours usually taken over a 24h period, and obtained through the use of ABPM remains contentious. While a number of studies have reported associations between short-term BPV and TOD [27-31] and cardiovascular events [25, 32], others have shown none or only weak associations after adjustment for mean BP [33, 34, 39].

There have been a number of different indices of variability proposed to describe short-term BPV with most studies considering SD [25, 27, 28, 30-34, 38, 39]. The appropriateness however, of such an index as an indicator of BPV has been disputed because it only reflects the dispersion of measurements around a single value (mean) not accounting for the order in which BP measurements were obtained [36, 37]. As a consequence, other indices of variability that have been used include: CV which attempts to adjust for the tendency of those with a higher average BP to also have a higher SD [25, 38, 39], ARV which is the average absolute difference between successive readings, and is thought to give a true reflection of real variability [28, 33, 36] and wSD which attempts to remove the influence of the day-night BP difference from the estimate of BPV [28, 30, 33, 40].

As the prognostic significance of short-term BPV remains uncertain with no consensus as to which measure of variability most accurately describes BPV, this study explores different indices of BPV and investigates their relationship with two parameters of subclinical TOD; LVH and microalbuminuria.

4.3 Methods

Study Population

The analysis utilises data from the Mitchelstown Study, a population based study of middle-aged men and women, recruited in Ireland 2010-2011. A detailed description of the study design is available from a previous publication [74]. In brief, the primary aim of the study was to provide a profile of cardiovascular health and their related factors in an Irish adult general population sample. The study recruited patients attending a single large primary care centre, the LHC, in Mitchelstown. Participants completed a detailed health and lifestyle questionnaire including a question on use of anti-hypertensive medication, and were invited to attend their primary care provider's surgery for a physical examination to be carried out by a nurse trained in the study research protocols. Study measurements included height, weight, BP and in addition, fasting blood samples (minimum of 8-h fast) and urine samples. Participants also underwent standard 12-lead electrocardiogram (ECG) and ABPM was offered to all participants. All participants provided written informed consent and ethical approval was obtained from the Clinical Research Ethics Committee Cork.

BP Measurements

Study BP was measured three times after 5 minutes of rest in a seated position by experienced research nurses using an OMRON M7 BP monitor (OMRON Healthcare, The Netherlands). The average of the second and third measurements was used. ABPM was measured using dabl ABPM system (dabl ltd., Ireland) with the Meditech ABOM-05 Monitor (Meditech LTD., Hungary). The monitors were programmed to obtain readings every 30mins and remained in place for 24h. Participants kept diaries of wake and sleep periods, which were used to calculate day and night BP respectively. If no diary was kept, the period from 1am to 6am was used as the night period and from 9am to 9pm as the day period. Only participants with a

minimum of 14 measurements during the day and a minimum of 7 measurements during the night period were included in the analysis [165].

As indices of short-term reading-to-reading BPV, we estimated the SD over 24h, wSD, CV ((SD 24h BP/mean 24h BP)*100) and ARV. The wSD is the mean of day and night SD values corrected for the number of hours included in each of these two sub-periods [40]. The ARV averages the absolute differences between consecutive measurements. To illustrate the subtle differences between each measure we have included 24h SBP profiles of four participants from the study along with their corresponding 24h variability measures (Figure 4-1.a-d). The two individuals illustrated in Figure 4-1.a and Figure 4-1.b have different overall levels of BP with the individual in Figure 4-1.b having substantially lower BP. In addition to this, the two BP patterns can be seen to fluctuate quite differently throughout the 24h but despite this, they have the same 24h SD, emphasizing the disadvantage of using SD as it only reflects dispersion around the average. Visually it is hard to justify that the two profiles have the same variability when it seems the first participant (Figure 4-1.a) has larger variation between each consecutive measurements compared to the second participant (Figure 4-1.b). In comparison, their 24h ARV differ considerably and highlights the benefit of using this measure as it accounts for the order of the BP readings. wSD and 24h CV values are also presented in Figure 4-1. The bottom two profiles are provided as another illustration where fluctuations and overall BP levels are different yet 24h SD values are similar but 24h ARV values vary (Figure 4-1.c & Figure 4-1.d).

Additional indices SD, CV, and ARV were calculated for both day and night periods. To investigate if BPV was influenced by nocturnal BP fall, a night to day BP fall (nocturnal dip) parameter was calculated as the ratio of mean BP between nightday periods. Additionally, the time rate of variability was calculated which is similar to ARV but is independent of the time intervals between measurements. All indices were calculated for both SBP and DBP.

Target organ damage

Each participant had a blood sample taken after a minimum of 8-h fast and in addition provided an early morning spot urine sample on the day of their appointment. Laboratory analyses included analysis for glycosylated haemoglobin (HbA1C) and ACR. A meta-analysis in 2010 demonstrated increased risk of mortality with urine ACR \geq 1.1 mg/mmol and as a result microalbuminuria is defined using this cut-point [78]. A 12-lead ECG was obtained in the standard manner using a Siemens – Eclipse 850i machine. ECGs were reviewed and coded for LVH voltage criteria by a cardiologist. The presence of LVH was defined by Cornell Product ECG voltage criteria i.e. SV3 + RaVL (+6 in women) X QRS duration \geq 2440 mm x ms [80].

Other Measurements

The questionaires provided data on participants smoking habits and were categorized as never, former and current smoker. Diabetes mellitus was defined as HbA1c level greater than or equal to 6.5% [166] or self-reported doctor diagnosis of diabetes. The classification of hypertension was based on SBP≥140 mmHg and/or DBP≥90 mmHg and/or on anti-hypertensive treatment. Weight was measured using a Tanita weighing scales and height measured with the use of a portable Seca length measure. Body mass index (BMI) was calculated as body weight in kilograms divided by height in meters squared. Participants with a BMI ≥30, 25-30 and <25 kg/m² were classified as obese, overweight and normal/underweight respectively.

Statistical Analysis

All of the data are expressed as arithmetic mean (SD) or as percentages where appropriate. Normally distributed continuous variables were compared using student's t-test. Correlations among variables were assessed by Pearson correlation coefficient (r). The association between BPV and the presence of TOD was assessed using logistic regression with adjustment for age, sex, smoking status, BMI, diabetes and anti-hypertensive treatment. Additional models adjusted for mean 24h BP. Multicollinearity issues between mean BP and variability measures were assessed by checking for inflated standard errors, correlation coefficients and variance inflation factors. Results are presented as odds ratios for a one-SD change with 95% confidence intervals (CI). The association between BPV and microalbuminuria was further explored by excluding those with diabetes from the analysis. Models were also separately run on those classified as hypertensive by ABPM and/or on antihypertensive treatment. In addition we examined just those on anti-hypertensive treatment alone. Receiver operating characteristic (ROC) curves were plotted to evaluate the predictive value of BPV. Additionally ACR was included in a linear regression model as a continuous outcome. To investigate the impact that the white-coat effect may have on variability, the analysis was repeated after the first hour of readings were excluded. Often, it is recommended to allow for multiple testing when examining independent measures and protect against data dredging where we might find a significant finding by chance alone when there is no real association. Typically an adjustment such as the Bonferonni correction is made but when measures are highly correlated as in our case, this method is inappropriate, as it will be highly conservative and may miss real differences [167]. Importantly in our study however, each variability index is in fact capturing different characteristics of variability and as we are interested in each measure separately or in isolation as opposed to seeing them as a collective, multiple testing is not required. If we were testing an overall hypothesis that BPV was related to our outcome then adjusting for multiple testing may have been considered. All of the statistical analyses were performed with the use of Stata version 12 (StataCorp., College Station, TX, USA). A value of P<0.05 was considered statistically significant. Figure 4-1 was plotted with the use of the R programming language.

4.4 Results

Of 3051 individuals invited to participate, 2047 (response rate: 67%) completed the questionnaire and physical examination component. ABPM was offered to all 2047 participants and it was completed by 1207 (response rate: 58%) of which 1134 had satisfactory amount of measurements recorded. The average numbers of day and night readings were 27 and 16 respectively. The main clinical characteristics of the

1134 participants are presented in Table 4-1 separately for those with and without LVH and microalbuminuria. Overall, participants had a mean (SD) age of 60.2 (5.5) and the majority were female (54%). Over half were classified as hypertensive (59%) and almost a tenth had diabetes (9.3%). The overall prevalence of LVH and microalbuminuria was 6.6% and 11.4% respectively.

Table 4-2 and Table 4-3 provides mean (SD) of different indices used to characterise BPV including the nocturnal dip, by 24h/day/night for both LVH and microalbuminuria respectively. With the exception of the nocturnal dip ratio there were no significant differences in BPV indexes between those with and without LVH (Table 4-2). However, night ARV, SD and wSD (both SBP & DBP) were significantly higher in those with microalbuminuria compared to those without. Additionally 24h ARV, day ARV, day SD (all SBP) and 24h ARV (DBP) were also higher in those with microalbuminuria (Table 4-3). Although the variability was higher in those without microalbuminuria, the nocturnal dip was significantly higher in those without microalbuminuria for both SBP and DBP. All four indices taken over the 24h period were significantly correlated with one another (correlation coefficient ranged 0.40-0.91 (p<0.001)).

Presented in Table 4-4 and Table 4-5 are results from regression analysis for both LVH and microalbuminuria respectively. Univariate logistic regression showed that none of the variability indices during any period were associated with LVH. In contrast, mean 24h ABPM was significantly associated with LVH in both univariate and adjusted models (OR=1.37(1.07-1.75)) (Table 4-4). The nocturnal dip was associated with a protective effect on LVH but the association did not persist when additionally adjusted for mean SBP (Table 4-4). In comparison ARV (24h, day and night), SD (day & night) and wSD were all univariately associated with microalbuminuria and the association persisted after adjustment for age, gender, smoking, BMI, diabetes and antihypertensive treatment (Table 4-5). However, when mean SBP was added to the model the association did not persist for all indices (Model 2, Table 4-5). Nocturnal dip was significantly associated with a protective effect on microalbuminuria in both the unadjusted and the fully adjusted models including mean BP (Table 4-5).

When analysis focused on only those classified as hypertensive by ABPM and/or taking antihypertensive treatment no differences to the findings above were observed. When those on antihypertensive treatment were analysed separately, models gave similar results to what we had seen already in relation to LVH and broadly similar in relation to microalbuminuria with the exception of 24h ARV. Despite adjustment for mean 24h BP in addition to age, gender, smoking, BMI and diabetes; 24h ARV was significantly associated with microalbuminuria (OR=1.43 (1.04-1.99). The prevalence of microalbuminuria in this sub-group was 14.7%.

The time rate of variability measure gave similar results to ARV (results not shown). No significant results were observed when mean BP was included in the models including when those on anti-hypertensive treatment were explored separately.

The stability of the models after the adjustment of mean BP was assessed to identify any multicollinearity issues. After adjustment standard errors were not inflated and were similar to the values before the inclusion of mean BP indicating stable models. Additionally; correlation coefficients were obtained between mean BP and all BPV indexes. The values ranged from -0.02-0.47. Finally, variance inflation factors were calculated in each model and all values were under 1.5 indicating that multicollinearity issues did not arise.

To evaluate the additional predictive value of BPV, ROC curves were plotted. The plots compared a model which included mean BP and BPV to a model which just included mean BP while adjusting for age, gender, smoking, BMI, diabetes and anti-hypertensive treatment. There was no significant difference between the areas under the curves regardless of which BPV measure or period of time was used (24h/day/night) (results not shown, plots in Appendix C).

All analyses were repeated for DBP. ARV (24h/day/night), night SD and wSD were found to be significantly related to microalbuminuria but the association did not persist once mean BP was added to the model. No other associations were observed with either LVH or microalbuminuria. Omitting the first hour of readings to explore the white-coat effect had no impact on our findings (data not shown). Results presented include the first hour of readings. The exclusion of participants

with diabetes in the analysis of microalbuminuria did not alter our findings. Similar findings to that of the logistic regression model were obtained when ACR was included in a linear regression model as a continuous outcome.

4.5 Discussion

In this large population based study we have explored the association between short-term BPV and subclinical TOD. Much debate remains in the literature as to the best measure of BPV and as a result four different indices were calculated and each assessed for their prognostic significance. Our work highlights the difficulties in accurately describing short-term variability with one summary measure. To illustrate this we provided cases of individuals who have distinctly different BP profiles and overall mean BP levels throughout the day yet have similar values in terms of SD but quite different values based on the other indices calculated. Despite the discrepancies between different measures our results indicated that there was no association between any index of variability we calculated and LVH even in unadjusted models. The associations between ARV (24h, day and night), SD (day & night), wSD, with microalbuminuria persisted following adjustment for a range of important potential confounders but did not persist after the models were additionally adjusted for mean BP. However, in a sub-analysis of those only taking treatment the association between 24h ARV with anti-hypertensive microalbuminuria persisted following adjustment for all confounders including mean BP.

The relationship between mean BP and TOD is well established and the literature does not question the importance of lowering mean levels of BP as is recommended in guidelines [18]. As a consequence, to identify BPV as an additional independent predictor we adjusted our analysis for mean 24h BP. The intrinsic relationship that can exist between mean BP and BPV can lead to invalid measures of variability and Hansen *et al.* [168] also claims the strong association between the two may cause problems in building stable regression models (although our analysis did not find such problems). Furthermore, SD is widely used as a measure of BPV but the appropriateness of such an index has been disputed because it only reflects the dispersion of measurements around a single value (mean) not accounting for the order in which BP measurements were obtained [36, 37]. This issue was highlighted in the different BP profiles presented in Figure 4-1. The CV has also been used and adjusts for the tendency of those with a higher average BP to also

have a higher SD as was found in this study. However these two measures (SD, CV) are still inherently linked to mean BP and this helps to advocate the use of ARV and wSD as more suitable measures of BPV.

Many studies have proposed ARV as an appropriate measure of variation which averages the absolute difference between successive readings and accounts for the order in which the BP readings are obtained [28, 33, 36], while wSD has also been utilised and is a weighted average of both day and night SDs which specifically attempts to remove the day-night difference in mean BP from the estimate of BPV [28, 30, 33, 40]. As a result 24h ARV and wSD are not strongly influenced by the difference between day-night mean BP levels. This is in contrast to SD and CV in which a large nocturnal dip will increase both values. The protective effect on the presence of microalbuminuria that was observed with nocturnal dip after adjustment for mean BP is not surprising. An absence in BP fall at night has been cited in many studies as a prognostic marker of cardiovascular events both in hypertensive [15, 169, 170] participants and the general population [171, 172].

Although the associations between ARV (24h, day and night), SD (day & night), wSD, with microalbuminuria did not persist after additionally adjusting for mean BP we plotted ROC curves to check if the combination of mean BP and BPV had more predictive power over that of just mean BP. Despite discrepancies between the different measures and the problems outlined above the plots revealed no additional benefit in including BPV over and beyond 24h BP irrespective of the BPV index used.

While there was no association overall between 24h ARV and microalbuminuria once adjusted for mean BP, among those on anti-hypertensive treatment the association persisted. We do not have data on specific treatment classes so we were unable to assess if this effect was mediated by a particular anti-hypertensive class. Some individual drug data was collected but sufficient numbers were not obtained to explore classes separately. Our finding does suggest however that antihypertensive treatment is having an effect on BPV. Evidence from meta-analyses suggest that although different anti-hypertensive-drug classes have similar effects

in terms of reducing BP levels, pronounced differences in their ability to reduce BPV [61, 62].

Considering the strong relationship that exists between mean BP and the presence of LVH, which was also evident in this study, it is interesting that no associations were identified with any measure of BPV even under unadjusted conditions alone. One contributing factor may be that LVH was determined based on the evaluation of a 12-lead ECG which is known to have low sensitivity in the detection of LVH [18]. Although it has its own technical issues, use of echocardiography would have been preferred as it is more sensitive than ECG in diagnosing LVH. Despite this however, guidelines promote the use of ECG as part of routine assessment of all hypertensive patients [18]. In addition to low sensitivity, we also had a low prevalence of LVH (6.6%) in the study. When these two factors are considered in combination, they may have had a large confounding effect on the relationship between BPV and TOD. When prevalence is low, there are few true positives in the sample, and false positives can be large compared to the number of true positives. It is clear that these factors may have masked the true underlying association in our data.

It is important to highlight that other studies have found an association between BPV and TOD even after adjustment for mean BP. Among hypertensive patients, Tatasciore *et al.* [30] found an association between awake SBP SD and LVMI in multivariate analysis even with the inclusion of awake SBP. Similarly, in multivariate analysis including daytime SBP Zakopoulos *et al.* [31] found an association between the daytime rate of SBP variation and left ventricular mass. Perhaps the use of echocardiography in the measurement of the outcome in these studies can, in part, explain discrepancies between these and our findings.

Another argument may be that visit-to-visit variability is a better prognostic predictor. Many large studies have demonstrated a strong relationship between visit-to-visit variability and all-cause mortality [38], cardiovascular events [25, 173] and TOD [163, 174]. However there is still no consensus on the matter and as highlighted in a recent review there is a need to design studies to prospectively determine the causes of visit-to-visit variability and determine if treatments that

reduce variability lead to improved clinical outcome [175]. Correctly determining the prognostic significance of both short-term and visit-to-visit BPV could have important implications for the prescribing of anti-hypertensive medication.

Limitations

The study is cross-sectional in nature and as mentioned above we are not able to determine whether increases in BPV promotes the development of TOD or if TOD represents a risk factor for increased BPV rather than being a consequence of it. Hansen et al. [168] argues that as TOD is a forerunner of cardiovascular complications, BPV will inevitably increase. This leads to the issue of reverse causality with increased BPV being a marker of underlying disease rather than being an independent predictor. The low sensitivity of ECG in the detection of LVH was a concern as previously highlighted. Although there are few guidelines focusing on BPV, O'Brien et al. suggests small intervals (15min) between measurements is best when measuring BPV [13]. Our monitors were set to intervals of length 30min and this is recognized as a limitation although there is still no strong consensus on the optimal interval size to measure BPV. Despite a response rate of only 58% the ABPM subsample was found to be representative of the full sample based on sex, age and education. In addition the prevalence of TOD in both the full sample and ABPM subsample were similar (LVH 6.0% vs 6.6%, ACR 10.6% vs 11.4%). The main strengths of the study lie in the robust community based design and large sample of ABPM data recorded.

4.6 Conclusion

The findings of this study have highlighted that accurately measuring short-term BPV over 24h is not straight forward and different indices can be heavily influenced by factors such as mean BP or nocturnal BP fall. Without consideration for these confounding factors inferences made on the true prognostic value of BPV could be misleading. Based on our findings variability indices such as ARV and wSD that are not influenced by the difference in mean BP between day-night periods best describe short-term BPV but were not associated with LVH or microalbuminuria after adjustment for mean BP. Future research should focus on long-term studies

where short-term BPV is measured at baseline and participants followed to view the development of TOD.



Figure 4-1 Four study participants and their 24h BP profiles. Panels (a) & (b) illustrate different patterns yet have similar 24h SD but different values based on the other indices. Both (c) & (d) demonstrate a similar issue (values rounded).

| | | | LVH | Microal | buminuria |
|---------------------|--------------|--------------|----------------|--------------|----------------|
| Characteristic | Total | No | Yes | No | Yes |
| | (n=1134) | (n=1058) | (n=75) | (n=996) | (n=128) |
| Age, y | 60.2 (5.5) | 60.5 (5.5) | 62.8 (4.7)** | 60.1 (5.5) | 61.1 (5.8) |
| Gender, Male n(%) | 527 (46.5) | 486 (45.9) | 40 (53.3) | 457 (45.8) | 65 (50.8) |
| BMI, n (%) | | | | | |
| Underweight/normal | 230 (20.3) | 216 (20.4) | 13 (17.3) | 207 (20.8) | 22 (17.3)** |
| Overweight | 498 (44.0) | 468 (44.3) | 30 (40.0) | 451 (45.3) | 41 (32.3) |
| Obese | 405 (35.7) | 373 (35.3) | 32 (42.7) | 338 (33.9) | 64 (50.4) |
| Smoking, n (%) | | | | | |
| Never | 564 (51.9) | 529 (52.2) | 35 (48.6) | 495 (52.1) | 67 (52.8) |
| Former | 355 (32.7) | 325 (32.1) | 30 (41.8) | 307 (32.3) | 42 (33.1) |
| Current | 168 (15.4) | 160 (15.8) | 7 (9.7) | 149 (15.7) | 18 (14.2) |
| HbA1C | 5.9 (0.8) | 5.9 (0.7) | 6.1 (1.2) | 5.8 (0.7) | 6.2 (1.2)** |
| Diabetes, n (%) | 103 (9.3) | 94 (9.1) | 9 (12.2) | 82 (8.4) | 21 (16.4)** |
| Office SBP, mm Hg | 134.2 (17.5) | 133.8 (17.3) | 140.4 (19.4)** | 133.4 (17.1) | 140.4 (20.4)** |
| Office DBP, mm Hg | 82.7 (10.1) | 82.5 (10.1) | 84.7 (11.8) | 82.4 (9.9) | 84.8 (12.3)* |
| Hypertension, n (%) | 665 (58.7) | 607 (57.4) | 58 (77.3)** | 563 (56.6) | 96 (75.0)** |
| Sleep time, hours | 8.6 (1.3) | 8.7 (1.3) | 8.3 (1.3) | 8.6 (1.3) | 8.8 (1.3) |
| ABPM measurements | | | | | |
| 24h SBP, mm Hg | | | | | |
| 24h DBP, mm Hg | 124.1 (13.3) | 123.8 (13.1) | 128.9 (15.4)** | 123.1 (12.7) | 131.5 (15.4)** |
| Awake SBP, mm Hg | /1.8 (8.3) | /1.8 (8.2) | /2.0 (9.5) | /1.4 (8.0) | /4.6 (10.0)** |
| Awake DBP, mm Hg | 131.4 (14.1) | 131.1 (14.0) | 135.1 (16.0)* | 130.5 (13.6) | 138.1 (16.1)** |
| Sleep SBP, mm Hg | //.4 (9.0) | //.4 (9.0) | //.0 (9.9) | //.0 (8./) | 80.0 (11.0)** |
| Sleep DBP, mm Hg | 112.3 (14.0) | 111.9 (13.6) | 118.6 (17.1)** | 111.2 (13.3) | 121.2 (16.2)** |
| | 62.8 (8.3) | 62.7 (8.2) | 63.6 (10.2) | 62.3 (8.0) | 66.4 (9.4)** |

Table 4-1 Baseline Characteristics by LVH and Microalbuminuria

Data are mean (SD). SBP: Systolic Blood pressure, DBP: Diastolic blood pressure, BMI:Body mass index, LVH: Left ventricular hypertrophy, ABPM: Ambulatory blood pressure monitor. Microalbuminuria: albumin:creatinine ratio \geq 1.1 mg/mmol. * p<0.05, **p-value<0.01, p-values comparing those with/without target organ damage

| Variability | 24-hour | 24-hour (mm Hg) | | Day (mm Hg) | | mm Hg) |
|-------------|------------|-----------------|------------|-------------|------------|------------|
| Measure | No | Yes | No | Yes | No | Yes |
| ARV SBP | 11.3 (2.5) | 11.7 (2.4) | 11.9 (3.1) | 12.0 (3.1) | 10.6 (3.3) | 11.2 (2.9) |
| ARV DBP | 7.8 (1.7) | 7.5 (2.0) | 8.1 (2.4) | 7.8 (2.8) | 7.3 (2.1) | 7.1 (2.2) |
| SD SBP | 15.8 (4.0) | 16.0 (4.1) | 12.8 (3.5) | 13.4 (3.5) | 11.2 (3.6) | 11.5 (3.2) |
| SD DBP | 11.5 (2.8) | 11.1 (2.7) | 8.9 (2.8) | 8.7 (2.7) | 8.0 (2.4) | 7.7 (1.8) |
| CV SBP | 12.8 (2.9) | 12.4 (2.9) | 9.8 (2.4) | 10.0 (2.4) | 10.0 (3.0) | 9.7 (2.7) |
| CV DBP | 16.1 (3.7) | 15.6 (3.9) | 11.7 (3.6) | 11.5 (3.8) | 12.9 (3.8) | 12.4 (3.2) |
| wSD SBP | 12.3 (2.9) | 12.8 (2.7) | - | - | - | - |
| wSD DBP | 8.7 (2.1) | 8.4 (2.0) | - | - | - | - |
| Dip SBP | 14.5 (6.7) | 12.0 (9.0) * | - | - | - | - |
| Dip DBP | 18.7 (7.5) | 17.2 (9.8) | - | - | - | - |

Table 4-2 Variability Indices and Nocturnal Dip by LVH

Data are mean (SD). ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, SBP: Systolic Blood pressure, DBP: Diastolic blood pressure, *p<0.05; represents significance between those with and without LVH during each period of the day

| Variability | 24-hou | 24-hour (mm Hg) | | Day (mm Hg) | | Night (mm Hg) | |
|-------------|------------|-----------------|------------|---------------|------------|---------------|--|
| Measure | No | Yes | No | Yes | No | Yes | |
| ARV SBP | 11.2 (2.5) | 12.4 (2.9) ** | 11.8 (3.1) | 12.7 (3.5) ** | 10.5 (3.2) | 11.9 (3.5) ** | |
| ARV DBP | 7.7 (1.7) | 8.2 (2.1) ** | 8.1 (2.3) | 8.5 (3.0) | 7.2 (2.1) | 7.9 (2.3) ** | |
| SD SBP | 15.8 (4.0) | 16.2 (4.2) | 12.8 (3.4) | 13.8 (4.1) ** | 11.1 (3.5) | 12.3 (3.7) ** | |
| SD DBP | 11.4 (2.8) | 11.5 (2.9) | 8.9 (2.7) | 9.2 (3.1) | 7.9 (2.3) | 8.7 (2.5) ** | |
| CV SBP | 12.8 (2.9) | 12.4 (2.9) | 9.8 (2.3) | 10.0 (2.8) | 10.0 (3.0) | 10.2 (2.9) | |
| CV DBP | 16.1 (3.7) | 15.6 (3.9) | 11.7 (3.6) | 11.7 (4.2) | 12.8 (3.8) | 13.3 (4.0) | |
| wSD SBP | 12.2 (2.8) | 13.3 (3.3) ** | - | - | - | - | |
| wSD DBP | 8.6 (2.0) | 9.0 (2.3) * | - | - | - | - | |
| Dip SBP | 14.6 (6.9) | 12.1 (7.3) ** | - | - | - | - | |
| Dip DBP | 18.8 (7.7) | 16.7 (7.4) ** | - | - | - | - | |

Table 4-3 Variability Indices and Nocturnal Dip by Microalbuminuria

Data are mean (SD). ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, SBP: Systolic Blood pressure, DBP: Diastolic blood pressure, *p<0.05; ** p<0.01 represents significance between those with and without microalbuminuria during each period of the day

| Variability (per SD | Unadjusted OR | Model 1 OR | Model 2 OR |
|---------------------|--------------------|-------------------|------------------|
| change in mm Hg) | (95% CI) | (95% CI) | (95% CI) |
| ARV: 24h ARV | 1.13 (0.90-1.41) | 1.01 (0.79-1.30) | 0.87 (0.66-1.14) |
| Day ARV | 1.04 (0.83-1.31) | 0.97 (0.76-1.25) | 0.87 (0.67-1.13) |
| Night ARV | 1.19 (0.96-1.49) | 1.08 (0.84-1.39) | 0.97 (0.74-1.26) |
| SD: 24h SD | 1.03 (0.82-1.30) | 1.01 (0.79-1.30) | 0.91 (0.70-1.18) |
| Day SD | 1.17 (0.94-1.46) | 1.09 (0.86-1.39) | 0.98 (0.76-1.28) |
| Night SD | 1.09 (0.87-1.36) | 1.02 (0.79-1.32) | 0.90 (0.69-1.19) |
| CV: 24h CV | 0.88 (0.68-1.12) | 0.89 (0.69-1.14) | 0.91 (0.71-1.17) |
| Day CV | 1.08 (0.86-1.36) | 1.01 (0.79-1.29) | 1.02 (0.80-1.30) |
| Night CV | 0.91 (0.71-1.16) | 0.87 (0.67-1.13) | 0.87 (0.67-1.14) |
| wSD | 1.18 (0.94-1.46) | 1.08 (0.85-1.40) | 0.94 (0.71-1.24) |
| SBP Nocturnal Dip | 0.71 (0.57-0.89)** | 0.78 (0.62-0.98)* | 0.81 (0.64-1.02) |
| 24h SBP | 1.42 (1.14-1.76)** | 1.37 (1.07-1.75)* | - |

Table 4-4 Association between SBP BPV and LVH

ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, LVH: Left ventricular hypertrophy, SBP: Systolic blood pressure, BP: Blood pressure, BPV: Blood pressure variability, OR: Odds ratio, * p<0.05, **p-value<0.01

Each variability measure entered separately and adjusted as follows: Model 1: Age, gender, smoking, BMI, diabetes, antihypertensive treatment; Model 2: Age, gender, smoking, BMI, diabetes, anti-hypertensive treatment, 24h SBP

| Variability (per SD | Unadjusted OR | Model 1 OR | Model 2 OR |
|---------------------|--------------------|--------------------|-------------------|
| change in mm Hg) | (95% CI) | (95% CI) | (95% CI) |
| ARV: 24h ARV | 1.51 (1.27-1.79)** | 1.44 (1.19-1.73)** | 1.20 (0.97-1.47) |
| Day ARV | 1.31 (1.11-1.56)** | 1.31 (1.10-1.57)** | 1.12 (0.93-1.37) |
| Night ARV | 1.46 (1.24-1.74)** | 1.35 (1.12-1.63)** | 1.16 (0.95-1.42) |
| SD: 24h SD | 1.10 (0.92-1.31) | 1.09 (0.90-1.31) | 0.89 (0.72-1.09) |
| Day SD | 1.29 (1.09-1.53)** | 1.29 (1.07-1.54)** | 1.09 (0.89-1.33) |
| Night SD | 1.35 (1.15-1.60)** | 1.27 (1.06-1.52)** | 1.06 (0.87-1.30) |
| CV: 24h CV | 0.84 (0.70-1.02) | 0.87 (0.72-1.06) | 0.89 (0.73-1.08) |
| Day CV | 1.09 (0.92-1.30) | 1.11 (0.92-1.34) | 1.12 (0.92-1.35) |
| Night CV | 1.07 (0.89-1.27) | 1.02 (0.84-1.23) | 1.01 (0.84-1.23) |
| wSD | 1.39 (1.18-1.64)** | 1.35 (1.12-1.62)** | 1.10 (0.89-1.35) |
| SBP Nocturnal Dip | 0.70 (0.59-0.84)** | 0.74 (0.62-0.90)** | 0.78 (0.65-0.94)* |
| 24h SBP | 1.78 (1.49-2.13)** | 1.67 (1.38-2.02)** | - |

Table 4-5 Association between SBP BPV and Microalbuminuria

ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, SBP: Systolic blood pressure, BP: Blood pressure, BPV: Blood pressure variability, Microalbuminuria: albumin:creatinine ratio \geq 1.1 mg/mmol, OR: Odds ratio *p<0.05; ** p<0.01

Each variability measure entered separately and adjusted as follows: Model 1: Age, gender, smoking, BMI, diabetes, antihypertensive treatment; Model 2: Age, gender, smoking, BMI, diabetes, anti-hypertensive treatment, 24h SBP

5. EXPLORING DIURNAL VARIATION USING PIECEWISE LINEAR SPLINES: AN EXAMPLE USING BLOOD PRESSURE (PAPER 3)

J. M. MADDEN^a

X. Li^b

P. M. KEARNEY^a

K. TILLING^c

A. P. FITZGERALD^{a,d}

^a Department of Epidemiology & Public Health, University College Cork, Ireland

^b Graduate Entry Medical School, University of Limerick, Ireland

^c School of Social and Community Medicine, University of Bristol, Bristol, UK

^d Department of Statistics, University College Cork, Ireland

This paper was published in Emerging Themes in Epidemiology Journal 2017 (see Appendix J)

5.1Abstract

There are many examples of physiological processes that follow a circadian cycle and researchers are interested in alternative methods to illustrate and quantify this diurnal variation. Circadian BP deserves additional attention given uncertainty relating to the prognostic significance of BPV in relation to CVD. However, the majority of studies exploring variability in ABPM collapse the data into single readings ignoring the temporal nature of the data. Advanced statistical techniques are required to explore complete variation over 24h.

We use piecewise linear splines in a mixed-effects model with a constraint to ensure periodicity as a novel application for modelling daily BP. Data from the Mitchelstown Study, a population based study of Irish adults aged 47-73 years (n=2,047) was utilized. A subsample (1,207) underwent 24h ABPM. We compared patterns between those with and without evidence of subclinical TOD (microalbuminuria).

We were able to quantify the steepest rise and fall in SBP, which occurred just after waking (2.23 mmHg/30min) and immediately after falling asleep (-1.93 mmHg/30min) respectively. The variation about an individual's trajectory over 24h was 12.3mmHg (SD). On average those with microalbuminuria were found to have significantly higher SBP (7.6 mm Hg, 95%CI: 5.0-10.1) after adjustment for age, sex and BMI. Including an interaction term between each linear spline and microalbuminuria did not improve model fit.

We have introduced a practical method for the analysis of ABPM where we can determine the rate of increase or decrease for different periods of the day. This may be particularly useful in examining chronotherapy effects of antihypertensive medication. It offers new measures of short-term BPV as we can quantify the variation about an individual's trajectory but also allows examination of the variation in slopes between individuals (random-effects).
5.2 Introduction

There are many examples of physiological processes that follow a circadian cycle such as cortisol, intraocular pressure and body temperature where abnormalities in these patterns have been shown to be related to depression [176], glaucoma [177] and delayed sleep-phase disorder [178]. The ability to analyse and capture features of these cycles remains a challenge but is necessary to get a deeper understanding of the mechanisms behind them. For example, the cardiovascular system shows clear circadian rhythmicity where researchers are interested in alternative methods to illustrate and quantify this diurnal variation [179]. Circadian BP represents a situation where diurnal variation deserves additional attention given the uncertainty relating to the prognostic significance of BPV [25, 32-34]. The benefits of using ABPM in addition to clinic measurements in the diagnosis and management of hypertension are well established [13, 18]. As well as mean day, night and dip values, ABPM provides measures of short-term BPV and individual profile patterns. The majority of studies examining short-term BPV have focused on summary measures such as the SD of ABPM readings over the day. These summary measures are easily obtained without the need for advanced statistical techniques [25, 32-34, 37, 39] but ignore the temporal nature of the data. To date relatively little work has modelled 24h ABPM profiles to exploit the full potential of ABPM data to capture short-term BPV [48]. Moreover, there are a lack of studies exploring circadian patterns and specifically, studies examining differences in patterns among different groups of individuals.

Cosinor analysis which incorporates a sinusoidal function has been the most common approach to modelling 24h BP [114-117], while a similar method, Fourier analysis [123, 124], has also been implemented. These approaches have focused on between-group effects (fixed-effects) where typically inferences are based on estimated differences in model parameters between particular groups of patients, such as comparing the estimated amplitude or MESOR between groups of individuals on different antihypertensive agents obtained in cosinor analysis [116]. The focus of fixed-effects is on population trajectories. However one of the main

advantages of ABPM is that we obtain individual BP profiles and modelling subjectspecific trajectories involves incorporating subject-specific effects (random-effects). To model mean profiles Selwyn *et al.* [106] used a hierarchical model incorporating a 4th degree polynomial. Lambert *et al.* extended on this by incorporating restricted cubic splines to model the mean BP profiles [48]. More recently, Edwards *et al.* [49] utilised orthonormal polynomials in a linear mixed model in a group of hypertensive subjects. Both polynomials and cubic splines, by their nature, have the ability to produce well-fitting curves to the data but have the disadvantage that the corresponding coefficients are challenging to interpret directly.

As an alternative we propose using piecewise linear splines in a mixed-effects model as a different approach for modelling ABPM data. Although linear splines have been used to model BP change over years and gestational age [109], using them to explore daily patterns of BP represents a novel method for analysing ABPM. This approach has the advantage that coefficients represent something meaningful, in this case the slope of BP at different periods of the day. To date it is unclear if different underlying circadian BP patterns exist across various groups of the population. This method allows slopes at a group level (and individual level) to be easily compared. Furthermore, using random-effects we want to predict and plot curves at an individual level and to explore BPV within each period of the day. Thus the aim of this study is twofold 1) to introduce and describe a mixed-effects piecewise linear model in relation to BP; 2) to apply our method to a middle-aged population sample and explore their circadian BP patterns. We also introduce and present a constraint for our model that ensures periodicity, so that on the average BP is the same 24h later. We are particularly interested in identifying distinct differences in the shape of mean curves at a group level. For purposes of illustration of the models at a group level, we will compare those with and without evidence of subclinical TOD, specifically microalbuminuria.

5.3 Methods

Study Population

The analysis utilises data from the Mitchelstown Study, a population based study of middle-aged men and women, recruited in Ireland 2010-2011. A description of the study design is available from previous publications [74, 180]. The study recruited patients attending a single large primary care centre, the LHC, in Mitchelstown. Participants completed a detailed health and lifestyle questionnaire, and attended for a physical examination including height, weight, BP, fasting blood samples and urine samples. ABPM was offered to all participants. All participants provided written informed consent and ethical approval was obtained from the Clinical Research Ethics Committee Cork.

BP Measurements

Study BP was measured three times after 5 minutes of rest in a seated position by experienced research nurses using an OMRON M7 BP monitor (OMRON Healthcare, The Netherlands). The average of the second and third measurements was used for analyses. ABPM was measured using dabl ABPM system (dabl ltd., Ireland) with the Meditech ABOM-05 Monitor (Meditech LTD., Hungary). The monitors were programmed to obtain readings every 30mins and remained in place for 24h. Participants kept diaries of wake and sleep periods, which were used to calculate sleep and waking times. Only participants with a minimum of 20 measurements during the day and a minimum of 7 measurements during the night period were included in the analysis (see Appendix E) [13, 181]. Additionally, any participants with data lacking for more than two consecutive hourly intervals were excluded [13, 73, 181].

Target Organ Damage

Each participant provided an early-morning spot urine sample on the day of their appointment. Laboratory analyses included analysis for ACR. Microalbuminuria is defined as $ACR \ge 1.1 \text{ mg mmol}^{-1}$ [78].

Statistical Analysis

Linear Mixed Model - Linear Splines

The linear mixed model [87, 88] is a well-recognised tool in the analysis of longitudinal data and its ability to obtain both population (fixed-effects) and subject-specific (random-effects) trajectories makes it particularly appealing for the analysis of ABPM data. However, its use to-date has focused on BP following a smooth curvature trajectory which results in spline and polynomial coefficients that are of no direct clinical relevance. Piecewise linear functions or linear splines offer an alternative. These involve segregating the data into different segments across time initially assuming the segments are the same for everyone. Within each partition, a linear spline is fitted and where these are connected are known as knot points. The corresponding coefficient of each spline represents the rate of increase or decrease of BP during each time period.

Knot selection

The position of the knot points were determined based on a number of factors. Firstly, to get a general sense of the shape of the data and determine regions of interest (how many knots were required), we plotted an average curve of BP including all participants to determine common knot points. In addition we incorporated prior known characteristics of BP. The period of awakening corresponds with an abrupt and steep acceleration of BP and for many the maximum value obtained during this morning period corresponds to their maximum BP reached throughout the day [182]. We also know that BP gradually falls throughout the day and usually dips to its lowest value during the sleeping period [18]. Since waking and sleeping times are clearly important in terms of changes in BP we decided that the use of these times as two additional subject-specific knot points was appropriate. We were able to create these subject-specific knots using the wake and sleep times reported by the participant. We had 49 readings for each individual, where the first reading was t_{1} (12pm) and the final reading was t_{49} (12pm the following day). Individual waking and sleeping times

were included within this range $(t_1 - t_{49})$. For each individual we created *m* linear splines, where the kth spline:

$$s_{k}(t) = 0 \text{ if } t \le t_{ki}$$

$$s_{k}(t) = t_{i} - t_{ki} \text{ if } t_{k} < t \le t_{ki+1} \text{ for } k=1,...,m \quad (22)$$

$$s_{k}(t) = t_{ki+1} - t_{ki} \text{ if } t > t_{ki+1}$$

Incorporating these linear splines into a linear mixed effects model for BP we get:

$$BP_{ij} = (\beta_0 + b_{0i}) + \sum_{k=1}^{m} (\beta_k + b_{ki}) s_k + \varepsilon_{ij}$$

$$b_i \sim MVN(0, \Sigma_b), \qquad \varepsilon_{ij} \sim N(0, \Sigma_{\varepsilon})$$
(23)

where BP_{ij} is the BP value for the j^{th} measurement on the i^{th} person, at time t_{ij} , the β 's are the fixed effects coefficients associated with the average intercept at β_0 (BP at 12pm) and the average slopes (β_k 's) between knot points, b's are the random-effects associated with the average intercept (b_{0i}) and average slopes between knot points, and ε_{ij} representing the individual-level residuals from the model. The model is extended by incorporating the subject-specific knots in the s_k term. It is assumed the random-effects (b_i) have zero mean and an unstructured variance-covariance matrix Σ_{b} . The individual level residuals have mean zero and variance-covariance matrix Σ_{ε} .

To expand equation (23) to include the restriction that on the average BP is the same 24h later we define an equation which states average, subject-specific change in BP over 24h is zero:

$$\sum_{k=1}^{m} w_{ki}\beta_{ki} = 0 \tag{24}$$

where w_{ki} is the width of the k^{th} interval (and where those involving wake and sleep times are subject-specific width intervals). Rewriting this in terms of β_1 gives:

$$\beta_{1i} = -\frac{\sum_{k=2}^{m} w_{ki} \beta_{ki}}{w_{1i}}$$
(25)

which implies:

$$\sum_{k=1}^{m} \beta_{ki} s_k = \beta_{1i} s_1 + \sum_{k=2}^{m} \beta_{ki} s_{ki} = \sum_{k=2}^{m} \beta_{ki} s_{ki}^*$$
(26)

where $s_{ki}^* = s_{ki} - \frac{w_{ki}}{w_{1i}} s_{1i}$, which allows us rewrite (23) as

$$BP_{ij} = (\beta_0 + b_{0i}) + \sum_{k=2}^{m} (\beta_k + b_{ki}) s_{ki}^* + \varepsilon_{ij}$$
(27)

To explore a group effect the model can easily incorporate a variable of interest, in this case TOD (microalbuminuria), as a dichotomous covariate. We further extended the model allowing the shape of the trajectory to depend on TOD by including interactions between TOD and each linear spline slope. Comparing this model with one without any interactions allowed us to test if the overall trajectory of BP was different between the two groups across the day. Additionally we were able to test if slopes between the groups differed at specific locations throughout the day. We adjusted for confounders by adding them into the model as fixed effects. In additional models we tested the effect of allowing the residual variance to differ between those with and without microalbuminuria. Similarly we tested the impact of allowing the interaction terms of microalbuminuria with each linear spline to be random to determine if there was heterogeneity of variance between the groups at any period of the day. Although we used an unstructured covariance structure for our models, we assumed these interaction terms to be independent of the other random-effects parameters. These interactions represented the difference in variation between the microalbuminuria groups within each segment. For all models explored we allowed all the linear spline terms to be random.

As individual ABPM readings taken close in time are likely to be correlated, a model with an independent residual correlation structure may not be appropriate. We compared this to a model with a first-order autoregressive AR(1) structure and examined a plot of the ACF to detect violations of the assumption of independence. Allowing for temporal correlation can potentially result in a large improvement in the precision of parameter estimates [93].

Models were compared formally by a LRT [88, 183]. The appropriate variance and residual function structures were also identified using a LRT in addition with an ACF plot. R-squared (R^2) statistic is often presented as a summary measure for linear models but due to theoretical or practical problems is rarely presented for mixed-models. Nakagawa and Schielzeth discuss these issues and present a general but simple method for calculating an appropriate R^2 for random intercept mixed-models [184]. Johnson extended this to include random slope models which we implement in our analysis [185].

The parameters for our final models were estimated using REML as this method produces unbiased estimates unlike ML estimation [93]. Subject-specific trajectories were estimated using Empirical Best Linear Unbiased Predictors (EBLUPs) of the random-effects [88]. Residual diagnostic plots were examined to verify model distribution assumptions. In addition a visual predictive check (VPC) was performed in which the estimated mean and the 90% prediction interval from our model were plotted together with the observed BP values and the 90% interquantile range of the observations. The purpose of the VPC is to assess graphically if predictions from the fitted model reproduce the central trend and variability of BP in the observed data, when plotted against time. It is an internal validation method that assesses the goodness-of-fit. [186] All analysis was completed for both SBP and DBP. All analysis were implemented in R [187] and parameter estimation for the mixed-effect model was carried out by means of the Ime command in nlme package [188].

Validation

Although difficult to interpret the coefficients, polynomial regression can still be a useful tool in the analysis of medical data to plot the trajectory of non-linear or curvilinear relationships [101]. As a method of validation for our approach we

additionally implemented a linear mixed model with orthogonal polynomials across time in both the fixed and random-effects, similar to that of Edwards *et al.* [49]. We wanted to determine if linear splines were capable of capturing the circadian rhythm of BP. This was investigated by comparing the trajectories obtained from both methods to determine if they followed similar patterns. A similar process to that of the piecewise model was followed when fitting the polynomial model. As we were only concerned to know if the general shape could be captured by our piecewise approach we were not worried about over-fitting the polynomial model. We implemented a model up to a 6th order polynomial allowing all the terms to be random.

5.4 Results

Of 3051 individuals invited to participate, 2047 (response rate: 67%) completed the questionnaire and physical examination component. ABPM was offered to all 2047 participants and it was completed by 1207 (response rate: 58%) people, of whom 1008 had a minimum of 20 day and 7 night measurements respectively. Of these 886 had no data missing for more than two consecutive hourly intervals, and the main clinical characteristics of these participants are presented in Table 5-1. Overall, participants had a mean age of 59.9 (5.5) and the majority were female (55%). Sixty percent were classified as hypertensive. Also presented in Table 5-1 are the characteristics of the full sample which shows the ABPM sub-sample follows a similar distribution in terms of age, sex, BMI and the presence of microalbuminuria. However, the proportion of those with hypertension was higher amongst those in ABPM sub group than in overall study population (60% vs 47%).

The plot of average SBP for the 886 subjects is presented in Figure 5-1. Based on this plot we identified two common knot points where the trajectory of SBP changed notably at 6pm and 4am. In addition to these two points we were able to include two subject-specific knot points for each participant based on the time an individual woke and went to sleep (Figure 5-1). This meant each participant was assigned 4 knot points which in turn resulted in their SBP pattern being broken into 5 linear segments.

Figure 5-2 represents subject-specific trajectories as a function of time only from a linear mixed-effects model using both orthogonal polynomials (6th order) (red line) and piecewise linear splines (blue lines). The plots suggest that the individual curves can be adequately captured by using piecewise linear splines.

We initially included the 5 linear splines as fixed-effects. A significant improvement in fit was observed when additionally including each term individually as a randomeffect, based on a LRT (all p<0.001). As a consequence we included all the linear spline terms as random-effects. To allow for temporal correlation we incorporated an AR1 structure which resulted in a significant improvement in fit (p<0.001) (rho=0.27). Examining the ACF plot indicated that the inclusion of an AR1 residual

structure adequately accounted for the auto-correlation in the data. This unadjusted model, which only incorporates linear splines as a function of time was our base model, Table 5-2 (Model 1). Presented are the parameter estimates (fixedeffects, random-effects correlation matrix, the autocorrelation decay ρ along with fit criteria values). With the exception of the slope for the period from 12.00 to 18.00 (0.02(0.04) mmHg/30min), all slopes differed significantly from zero (all p<0.001). This suggests that on average this is the period during the day where average SBP remains constant. The largest rise and fall in SBP occurred between wake and 12.00 (2.23 mmHg/30min) and, between sleep and 04.00 (-1.93 mmHg/30min) respectively. These segments correspond to the period when an individual wakes up and the period immediately after they fall asleep. The variation in slopes was lowest from 12.00 to 18.00 where the variance was 0.51. The largest variation in slopes was observed between waking and 12.00 where the variance was 2.05 which is substantially larger in comparison to the rest of the day. The model R^2 value which illustrates the proportion of variance explained by both the fixed and random factors was quite high (0.67).

In subsequent models we adjusted for age, sex and BMI. We also included our variable of interest, microalbuminuria, to determine if it could help explain the larger variation in the period, wake to 12.00 (Model 2, Table 5-2). The residual variance, which represents the variation about an individual's trajectory, was 12.3mmHg. We additionally allowed the residual variance to vary between microalbuminuria groups (ratio of SD of those with to without microalbuminuria was 1.09). On average, over the day, those with microalbuminuria were found to have significantly higher SBP (7.6 mm Hg, 95% CI: 5.0-10.1, p<0.001). However, adjusting for age, sex, BMI and microalbuminuria had almost no effect on the model parameter estimates (except the intercept). To determine if slopes were different between groups at different times of the day we included an interaction between each linear spline and microalbuminuria (Model 3, Table 5-2). Although two of the interaction terms were marginally significant, a LRT suggested that including interaction terms did not improve the overall fit to the data (p=0.12). Based on additional models (results not shown) we found no evidence that the

variance of the random-effects varied with microalbuminuria. With the inclusion of age, sex, BMI and microalbuminuria we concluded that Model 2 offered the best fit to the data (Model 2 vs Model 1, p<0.01). Residual diagnostic plots of the models showed no violation of assumptions (results not shown). The VPC plot showed the model was adequately predicting central trend and variability of SBP in the observed data, when plotted against time (see Appendix C).

Figure 5-3 represents the average piecewise linear curve along with a 95% confidence interval for those with and without the presence of microalbuminuria using Model 2. The numbers on the plot correspond to the time periods presented in Table 5-2. It is clear that those with microalbuminuria have a higher average SBP throughout the day. For the purposes of this plot we have set the sleep and wake time knots at 23.00 and 08.00 respectively. A similar plot using model 3 can be found in Appendix C. Similar findings were found for all analysis when repeated using DBP (results not shown).

5.5 Discussion

In this large population based study we present an alternative method of modelling 24h BP that can easily be applied to any physiological process that follows a circadian cycle. Our novel but simple approach utilising a piecewise linear random-effects model, with an adjustment to ensure that the average level is the same at the beginning and end of each 24h period, offers a practical alternative to other methodological modelling techniques for researchers exploring circadian patterns. The flexible model has the ability to capture overall average, group and individual trajectories (in addition to being capable of examining slopes at different periods of the day).

Despite the large amounts of literature relating to BP, those specifically modelling 24h ABPM remain sparse. Our method offers new measures of short-term BPV as we can quantify the variation about an individual's trajectory but it also allows examination of the variation in slopes between individuals (random-effects). Our results indicated that after adjustment for age, sex and BMI the sharpest fall in BP occurred just after an individual went to sleep and the steepest rise occurred just after waking. Although there was a significant difference on average between those with and without microalbuminuria we found there was no overall improvement in fit after including interaction effects with the spline terms. However interestingly we found that the variation after awaking, representing what is known as the morning surge was considerably larger than the other periods of the day.

It has been acknowledged there is not a generally accepted "standard" method of analysing 24h ABPM [49]. Cosinor analysis has been highlighted as the most common approach [114-117] while Fourier analysis [123], has also been implemented which are both based on the idea that any time series can be described by a series of cosine (and sine) waves of various frequencies [189]. It has been suggested that these methods impose too many restrictions on the shape of the profile and have been shown to fit real profiles poorly [119]. Wang *et al.* [120] suggest problems with fitting a sinusoidal function to a circadian pattern include (i) that the pattern over time may not be symmetric; that is, the peak and nadir may

not be separated by 12 hours and/or the amplitude and width of the peak may differ from those of the nadir, (ii) sometimes there are local minimum and maximum points. Additionally Wang *et al.* [121] suggests that the sinusoidal function is too restrictive and "rhythms with a shape closely approximating a cosine curve are uncommon" [122]. Alternative methods have examined restricted cubic splines and more recently orthonormal polynomials [48, 49]. As we highlighted previously these approaches may model the data quite well and their curvature nature may look graphically appealing but it is difficult to understand and compare their resulting coefficients.

Piecewise regression which allows separate slopes to be fitted to observations before and after a certain period or event (knot points) has been cited as a useful tool that should be implemented more often in the context of epidemiological studies [190] but has not, to the best of our knowledge been used with ABPM data or other physiological processes that are circadian. The benefit of this method as opposed to polynomials is that the regression coefficients represent something meaningful directly without the need for further manipulation of the results - in our context, the rate of increase or decrease of BP for a certain time of day. The position of the knot points can easily be altered depending on the requirements of a specific study. For example if we were examining the effect of dialysis on BP in haemodialysis patients we could fix knot points at the time their dialysis began and at period(s) a number of hours later.

The morning is recognised as the most important period in relation to CVDs [23] and cardiovascular events occur more frequently in this period [23, 45, 46]. In our study we found that the steepest rise (slope) occurred during the period just after waking which is in line with the literature, thus verifying that our method is capturing known features of the data. It is suggested that the abrupt steep rise in BP may explain the link between cardiovascular events and the morning period [182]. In a review of morning surge with cardiovascular risk, 3 different definitions of morning surge were identified, all of which simply use BP differences where they subtracted some average night value minus an average of morning BP readings [45]. We argue that our method offers a more accurate estimate, as by definition of

a slope we can specifically quantify the rate of "surge". In fact, Parati *et al.* argue that a method that would be capable of capturing a slope similar to one purposed by our method would provide an accurate method of estimating the morning surge [142]. Considering that morning surge has been cited as a predictor of stroke and advanced TOD independent of ambulatory BP and nocturnal BP [23, 46], accurately quantifying it remains an important issue, particularly when we are assessing the benefits of antihypertensive medication in their ability to reduce this steep rise. This may not only have health implications but also financial benefits. A similar argument could be put forward for the dipping effect at night which is usually quantified just as a ratio of the mean BP between night-day periods. The slope at night obtained by our approach may represent a more accurate measure but further work would be needed to explore this.

Kario argues that the perfect 24h BP control is not limited to reducing mean BP but includes restoring disrupted circadian BP rhythms and reducing exaggerated BPV [23]. As highlighted previously most studies examining BPV have concentrated on summary measures of variability such as SD over 24h or separated into day and night values [25, 32-34, 37, 39, 180]. With the use of our mixed-effects model we were are able to obtain superior measures of BPV that take into account the temporal nature of the data. We were able to quantify the variation about an individual's trajectory but also the variation in slopes between individuals. Our work highlighted that the largest variation between individuals occurred during the morning surge period. Adjusting for age, sex and BMI did not help explain this variation. Similarly the presence of microalbuminuria had little impact on the variation. Ideally we would have preferred to explore if the variation could in part predict cardiovascular events but as data is currently only available for wave one, we have been restricted to explore a surrogate marker in microalbuminuria and have acknowledged this as a limitation. Further work is warranted to include CVD endpoints but perhaps an underlining physiological phenomenon of BP is that it is most variable in the morning possibly because this period of the day has an abrupt rise. Although some of the knots are subject-specific, others are at common fixed locations which may not represent the best position for a specific individual and this

assumption is recognised as a limitation. In addition to the average plot, we have attempted to incorporate our knowledge of the underlying pattern of BP to help inform our knot positions as suggested by Howe *et al.* [91].

As debate remains in relation to how to correctly quantify short-term BPV [180, 191] our approach offers new alternatives that utilise the full power of ABPM that is often lost when using summary measures such as SD because it only reflects the dispersion of measurements around a single value (mean) not accounting for the order in which BP measurements were obtained [37, 40]. The ability to determine variation over specific periods of the day offers a novel measure of variability in the analysis of BP which may have benefits when attempting to determine the optimal timing of antihypertensive medication administration in future studies. Finally, the approach and discussion outlined is not restricted to the use of BP and can easily be implemented on any physiological process that demonstrates a circadian cycle. BP is not the only biological process where disruptions to circadian rhythms are clinical relevant. Wang et al. found that those with Cushing syndrome exhibited no circadian rhythm of cortisol, while those with depression showed a dampened rhythm compared to the normal group [121]. Liu *et al.* found that larger short-term fluctuations in intraocular pressure are more common in glaucoma [192]. Similar to the morning BP surge, it was found that intraocular pressure was higher in the morning and more prevalent in those with glaucoma. This suggests that our approach may be beneficial to the exploration of other biological rhythms that have similar features to that of BP.

5.6 Conclusion

This study has introduced a novel but practical method for the analysis of ABPM data. Based on our work circadian BP patterns can be modelled using a mixedeffects model with piecewise linear splines. The main advantage of our method compared to other approaches is that the resulting regression coefficients have direct interpretation. We can determine the rate of increase or decrease at different periods of the day. In addition we can determine alternative measures of

variability compared to classical BPV indices. Future research in this area should focus on the association between the measures obtained from this method to stronger clinical outcomes.



Figure 5-1 Plot of average SBP over 24h which helped identify 6pm and 4am as common knot points for all participants where there was a notable change in trajectory of BP. Also highlighted are the periods where individuals woke and went to sleep. In addition to the two common points, we were able to obtain additional (two) subject-specific knot points at wake and sleep times.

| Characteristic | Total (n=2047) | ABPM (sub-sample) |
|--|----------------|-------------------|
| Characteristic | | Total (n=886) |
| Age, y | 59.8 (5.5) | 59.9 (5.5) |
| Gender, Male n(%) | 1008 (49.2) | 401 (45.3) |
| BMI, n (%) | | |
| Underweight/normal (<25kg/m ²) | 447 (21.9) | 195 (22.0) |
| Overweight (25-30 kg/m ²) | 925 (45.3) | 380 (42.9) |
| Obese (≥30 kg/m²) | 668 (32.8) | 310 (35.0) |
| Office SBP, mm Hg | 129.6 (16.9) | 134.7 (17.7) |
| Office DBP, mm Hg | 80.1 (9.8) | 83.1 (10.2) |
| Hypertension, n (%) | 951 (46.5) | 528 (59.7) |
| Microalbuminuria | 215 (10.6) | 62 (7.0) |

Table 5-1 Baseline Characteristics

Data are mean (SD). BMI:Body mass index, ABPM: Ambulatory blood pressure monitor. Hypertension: ≥140/90

mmHg and/or on antihypertensive treatment.



Figure 5-2 Individual SBP readings along with predicted subject-specific trajectories from a linear mixed effects model as a function of time only using two different approaches; polynomials (red line) and piecewise linear splines (blue lines).

| Parameter | Model 1 | Model 2 | Model 3 |
|---------------------------------|---------------|---------------|----------------|
| Fixed effects (SBP mmHg/30mins) | Estimate (SE) | Estimate (SE) | Estimate (SE) |
| SBP at 12.00 | 134 (0.54) | 119.2 (4.6) | 119.3 (4.6) |
| Microalbuminuria | - | 7.57 (1.30)* | 5.79 (1.67)* |
| Slope for spline time period: | | | |
| 1. 12.00 – 18.00 | 0.02 (0.04) | 0.03 (0.04) | 0.03 (0.04) |
| 2. 18.00 - Sleep | -1.00 (0.04)* | -1.00 (0.04)* | -1.01 (0.04)* |
| 3. Sleep – 04.00 | -1.93 (0.05)* | -1.95 (0.06)* | -1.99 (0.06)* |
| 4. 04.00 - Wake | 1.69 (0.05)* | 1.70 (0.05)* | 1.71 (0.05)* |
| 5. Wake – 12.00 | 2.23 (0.07)* | 2.21 (0.07)* | 2.26 (0.07)* |
| Microalbuminuria x Spline | 2 | | |
| interaction | | | |
| 1. 12.00 – 18.00 | - | - | -0.06 (0.14) |
| 2. 18.00 - Sleep | - | - | 0.05 (0.13) |
| 3. Sleep – 04.00 | - | - | 0.37 (0.18)** |
| 4. 04.00 - Wake | - | - | -0.06 (0.16) |
| 5. Wake – 12.00 | - | - | -0.48 (0.22)** |
| Continued on following page | | | |

Table 5-2 Various models with parameter estimates for slopes at each segment along with corresponding correlations and variances

| Parameter | Model 1 | Model 2 | Model 3 |
|---------------------------------|----------------------------------|----------------------------------|----------------------------------|
| Fixed effects (SBP mmHg/30mins) | Estimate (SE) | Estimate (SE) | Estimate (SE) |
| | | | |
| | | | |
| Random effects | | | |
| | 223.6 | 199.5 | 200.5 |
| | -0.23 0.51 | -0.23 0.50 | -0.24 0.51 |
| Σ | -0.23 -0.10 0.55 | -0.25 -0.10 0.54 | -0.25 -0.11 0.55 |
| | -0.23 -0.45 0.03 1.39 | -0.28 -0.46 0.02 1.41 | -0.28 -0.44 0.02 1.40 |
| | 0.46 - 0.28 -0.74 -0.05 0.66 | 0.47 -0.31 -0.74 -0.05 0.65 | 0.49 -0.33 -0.73 -0.04 0.65 |
| | 0.34 -0.06 -0.21 -0.78 0.19 2.05 | 0.42 -0.03 -0.22 -0.80 0.23 2.00 | 0.42 -0.04 -0.20 -0.81 0.24 1.97 |
| σ | 12.3 | 12.3 | 12.2 |
| ρ | 0.27 | 0.27 | 0.27 |
| | | | |
| R^2 | 0.67 | 0.68 | 0.68 |
| Log-likelihood | -149608 | -149505 | -149502 |
| | | Model 2 vs Model 1 (p<0.001) | Model 3 vs Model 2 (p=0.12) |

Microalbuminuria: albumin:creatinine ratio ≥ 1.1 mg/mmol

*P<0.001,**P<0.05

Model 1: Fixed effects (5 linear splines), random effects (5 linear splines).

Model 2: Fixed effects (5 linear splines, microalbuminuria, age, sex, BMI), random effects (5 linear splines).

Model 3: Fixed effects (5 linear splines and interaction with microalbuminuria, age, sex, BMI), random effects (5 linear splines).

Random Effects matrix shown has variances on the diagonal and correlation coefficients on off-diagonals.



Figure 5-3 Predicted average (95% CI) piecewise linear trajectory of those with/without presence of microalbuminuria adjusted for age, sex and BMI using a linear mixed-effects model (Model 2). Each linear spline represents the rate of SBP increase or decrease (slope) for that segment and has been given a corresponding number which is referred to in Table 5-2. For the purposes of this plot we have set the sleep and wake time knots at 23.00 and 08.00 respectively.

6. MORNING SURGE IN BLOOD PRESSURE USING A RANDOM-EFFECTS MULTIPLE-COMPONENT COSINOR MODEL (PAPER 4)

J. M. MADDEN^a

L. D. BROWNE^a

P. M. KEARNEY^a

X. LI^b

K. TILLING^c

A. P. FITZGERALD^{a,d}

^a Department of Epidemiology & Public Health, University College Cork, Ireland

^b Graduate Entry Medical School, University of Limerick, Ireland

^c School of Social and Community Medicine, University of Bristol, Bristol, UK

^d Department of Statistics, University College Cork, Ireland

This paper is currently under review at Statistics in Medicine.

6.1Abstract

BP fluctuates throughout the day. The pattern it follows represents one of the most important circadian rhythms in the human body. For example it is well known that the absence of a dip in night-time BP is associated with poor cardiovascular outcomes. Similarly the morning BP surge has been suggested as a potential risk factor for cardiovascular events occurring in the morning but the accurate quantification of this phenomenon remains a challenge. Here, we outline a novel method to quantify morning surge that can also be used to obtain other measures of BP variability throughout the day.

We demonstrate how the most commonly used method to model 24h BP, the single-cosinor approach, can be extended to a multiple-component cosinor random-effects model. We outline how this model can be used to obtain a measure of morning BP surge by obtaining derivatives of the model fit. The model is compared to a FPCA which determines the main components of variability in the data. Data from the Mitchelstown Study, a population based study of Irish adults (n=2,047) was utilized where a subsample (1,207) underwent 24h ABPM.

Our findings demonstrate that the most common approach to model ABPM, the single cosinor, does not provide the best fit to the data. Our two-component model developed with random-effects analysis provided a significant improvement in fit. In addition, it provided a similar fit to the more complex three-component model and a model captured by b-splines using functional principle component analysis. The estimate of the average maximum slope obtained from the derivatives from the model was 2.857 mmHg/30min (bootstrap estimates; 95% CI: 2.855-2.858 mmHg/30min). Simulation results allowed us to quantify the between-individual SD in maximum slopes which was 1.02 mmHg/30min.

By obtaining model derivatives we have demonstrated a novel approach to quantify morning BP surge and its variation between individuals. This is the first demonstration of a cosinor approach to obtain a measure of morning surge.

6.2 Introduction

Elevated BP is the most prevalent treatable risk factor for cardiovascular disease affecting one billion people globally [2, 3]. It is well known that BP does not remain stationary but fluctuates throughout the day and follows a circadian rhythm. Morning surge refers to the phenomenon that occurs in individuals during the first few hours after waking up in the morning when there is an exaggerated spike or surge in BP [23, 45]. It has frequently been suggested that this surge may be a risk factor for cardiovascular events occurring in the morning [23, 45, 46]. However, the accurate quantification of this phenomenon remains a challenge. A recent meta-analysis examining the prognostic significance of morning surge in 17 studies, identified seven different calculations for the term which highlights the problem [193]. All but one of these seven estimates involved simple subtraction of BP values where they subtracted some average night value minus an average of morning BP readings. While these measures can easily be calculated without the need for advanced methodology, they may not accurately quantify a surge or rate of change.

Rather than focusing on one value (mean BP) as obtained by traditional measurement techniques, ABPM which obtains multiple readings over a 24h period offers a unique insight into an individual's underlying circadian rhythm [13]. Parati *et al.* argue that rather than simplifying ABPM data into mean summary measures, incorporating all the data in more advanced models can lead to more robust estimates of clinically relevant parameters such as dipping status and morning BP surge [142]. In order to get a more advanced measure of surge, it is paramount that we can first accurately model ABPM.

To date, the main approaches proposed to model 24h BP that incorporate all the data and fully utilise the benefits of the longitudinal nature of ABPM include the cosinor method [114-118], cubic splines [48], polynomials [49] and recently a double logistic model [194, 195]. Moreover, there is no accepted "standard" method for analysing ABPM [49], and research on the longitudinal analysis of 24h ABPM is lacking [48]. Cosinor analysis has traditionally been the most common approach to modelling 24h BP. This method oversimplifies the data as it attempts to describe a 24h circadian pattern with the use of a single sinusoidal function. The

assumption of a simple symmetrical pattern for diurnal BP is unable to account for large variation in BP over a 24h period [142]. Although not as common as the single-component cosinor model, attempts to extend the model by including multiple cos terms (Fourier analysis) allow more flexible curves to be obtained while remaining periodic [123-125]. The majority of the studies to date exploring the use of sinusoidal functions have focused on fixed-effects where the inference is on population effects [114-118]. A key feature of ABPM analysis however is the exploration of subject-specific effects (random-effects) where its use in cosinor models has been limited [126].

Thus the purpose of this study is twofold. To first demonstrate that extending the traditional single cosinor to a multiple-component cosinor in a random-effects model can be achieved while offering a substantial improvement in fit to ABPM data compared to the single-component model. Moreover the model is compared to a FPCA which determines the main components that account for the majority of the variation in the data. The parameters from the cosinor model are compared to the functional principle component scores.

Secondly, by calculating first-order derivatives of the model fit, we present a novel alternative method to locate and quantify the magnitude of slopes at critical points on the trajectory. This simple application of derivatives allows us to quantify a measure of morning BP that specifically represents a surge parameter. This may be beneficial in future studies exploring the prognostic significance of morning BP and chronotherapy effects of antihypertensive medication. For purposes of illustration of the models at a group level, we compare first derivative curves in those with and without evidence of subclinical TOD, specifically microalbuminuria.

6.3 Methods

Study Design and ABPM

The analysis utilises data from the Mitchelstown Study, a population based study of middle-aged men and women, recruited in Ireland 2010-2011. A description of the study design is available from previous publications [74, 180]. The study recruited patients attending a single large primary care centre, the LHC, in Mitchelstown. Participants completed a detailed health and lifestyle questionnaire including a question on use of anti-hypertensive medication, and attended for a physical examination including height, weight, blood pressure and fasting blood samples. Each participant provided an early-morning spot urine sample on the day of their appointment. Laboratory analyses included analysis for ACR. Microalbuminuria is defined as ACR \ge 1.1 mg mmol⁻¹ [78]. Study BP was measured three times after 5 minutes of rest in a seated position by experienced research nurses using an OMRON M7 BP monitor (OMRON Healthcare, The Netherlands). The average of the second and third measurements was used. The classification of hypertension was based on SBP≥140 mmHg and/or DBP≥90 mmHg and/or on anti-hypertensive treatment. ABPM was offered to all participants and was measured using dabl ABPM system (dabl ltd., Ireland) with the Meditech ABOM-05 Monitor (Meditech LTD., Hungary). The monitors were programmed to obtain readings every 30mins and remained in place for 24-h. Participants kept diaries of wake and sleep periods, which were used to calculate sleep and waking times. Only participants with a minimum of 20 measurements during the day and a minimum of 7 measurements during the night period were included in the analysis [13, 181]. Additionally, any participants with data lacking for more than two consecutive hourly intervals were excluded (see Appendix E) [13, 73, 181]. All participants provided written informed consent and ethical approval was obtained from the Clinical Research Ethics Committee Cork.

6.4 Statistical Analysis

Cosinor Analysis

The single-component cosinor model, which was first developed by Halberg [112, 113], uses a single cosine function as a model for physiological processes that have a circadian rhythm. This can be extended to a multiple-component model in the context of BP:

$$BP(t) = M + \sum_{i=1}^{n} A_i \cos\left(\frac{2\pi t}{\tau_i} + \phi_i\right) + e_t , i = 1, 2..., n$$
(28)

where BP(t) is BP as a function of time (t), M is the MESOR, the average value over the period, A is the amplitude for each cosine term (half the difference between the highest and lowest values, or the distance between the MESOR and the highest (lowest) value), τ is the period or duration of one cycle corresponding to each cosine term, ϕ is the acrophase (a measure of the time of the overall high values recurring in each cycle for each cosine term), e is the error term and n represents the number of cosine curves (n=1 represents the case of the single-component cosinor model).

The fixed-effects multiple-component model in equation (28) can be incorporated into a random-effects model. The mixed-effects model [87, 88] is a well-recognised tool in the analysis of longitudinal data that allows both population (fixed-effects) and subject-specific (random-effects) trajectories to be obtained which makes it useful for the analysis of ABPM data. It is assumed the random-effects have mean zero and an unstructured variance-covariance matrix. The individual level residuals (*e*) have mean zero and variance-covariance matrix Σ_e . As individual ABPM readings taken close in time are likely to be correlated, a model with an independent residual correlation structure may not be appropriate. We compared this to a model with a first-order autoregressive AR(1) structure and examined a plot of the ACF to detect violations of the assumption of independence. We determined the appropriate number of cosine terms by graphically comparing subject-specific predicted BP fits to the data (while increasing the number of terms) and formally by comparing models with different number of cosines using a LRT [88, 183]. Based on a LRT we univariately tested the inclusion of each term as a random-effect. If a

significant improvement was obtained, the term was included as a random-effect in the final model. The appropriate variance and residual function structures were also identified using a LRT in addition with the ACF plot.

The parameters for our final model were estimated using REML as this method produces unbiased estimates unlike ML estimation [93]. Subject-specific trajectories were based on empirical Bayes estimates of the random-effects [88].

Cosinor Derivatives and Morning Surge

To obtain an estimate of the maximum morning surge we first estimate BP(t) for each individual from our random-effects model and then obtain its first derivative, BP(t)'. This will give the rate of change or slope at each time of the day. Next, by limiting our analysis between 02:00 to 12.00 we can obtain the maximum slope and corresponding time during this period by numeric estimation for each individual which corresponds to their maximum morning slope or surge. As participants kept diary entries, we know their waking times. The earliest waking time is 05:00 but we broadened our period of interest further back to 02:00 in the unlikely event that an individual's maximum surge occurred just before waking. As the shape of the data is not overly complex over this period, assuming only one local maximum for this time is a reasonable assumption.

Cosinor Bootstrap and Simulations

Bootstrap estimation was performed in order to get an unbiased estimate of the standard error for the morning slope. A total of 1000 bootstrap datasets were created by randomly resampling from the original dataset with replacement. The bootstrap estimates were examined and determined to follow a normal distribution from which a standard error could be obtained. Results from the analysis were used to obtain bias corrected 95% CI for the slopes.

Additionally, to obtain an estimate of the between-individual variance in maximum slopes we ran 1000 simulations based on our final model. From this we obtained an estimate of the distribution of the slopes. This is similar to exploring the distribution

of the individual slopes from our final model but should result in a more precise estimate of between-person variation.

Functional Principle Component Analysis

The aim of FPCA is to find a combination of a few functions which capture the largest proportions of variation in the data. As a method of validation we compared model fits from the final multiple-component cosinor model to a model arising from FPCA. The purpose of this was to determine if the cosinor model was capable of capturing fluctuations in BP as well as the FPCA which is less restricted and can provide a flexible fit.

We implemented a FPCA which can be seen as an analogous to multivariate PCA where we can identify the main types of variation in patterns as a function of time as opposed to discrete measures [129, 130]. Instead of eigenvectors we obtain eigenfunctions which are associated with each eigenvalue and represent the FPCs which describe the different variations in the data.

To begin with, the mean curve was obtained through a method developed by Yao et al. which first ignores the hierarchical structure of the data and then fits a smooth curve to the pooled data [134]. This estimate of the mean curve is then used to obtain the covariance matrix of the deviations from the mean for each pair of timepoints. This covariance matrix is smoothed using a bivariate smoother and the main diagonal is removed to normalise data. This smoothed covariance matrix is then decomposed into a linear combination of orthogonal (uncorrelated) eigenfunctions and eigenvalues i.e. into its principal components and scores [133]. The first principal component captures the most variation, the 2nd captures the secondmost variation and so forth. Therefore, a linear combination of a few efficient functions can account for a high proportion of the variance. In our analysis, 10 Bspline basis functions were used to estimate the mean function, and for the bivariate smoothing of the covariance function. The number of FPCs to include was then determined by visual inspection of a scree plot. In the context of randomeffects models these FPCs can be seen as patterns of within-subject variation remaining after the mean fit.

The weights that define the optimal fit to each function are the principle component scores. These can be used to obtain individual curves by multiplying the weights by the functions. Zero scores for an individual would result in their trajectory following the mean pattern. The scores would usually be estimated through numerical integration but with sparse data, as in the case of ABPM (compared to high sampling frequency data e.g. every second, that is often associated with functional data [131]), the approximation is sometimes deemed inadequate and in this case the scores were estimated by the principal component analysis through conditional expectation (PACE) method [133, 134]. Using this method the scores are estimated for each individual using their repeated measures while borrowing strength from the cohort with sample estimates of the mean function, covariance, eigenvalues and eigenfunctions [133, 134].

As a final step the random-effects of the multiple-component cosinor model were correlated with the individual FPC scores from the FPCA. This allowed us to determine if the cosinor model and their parameters were capturing the main components of 24h BP obtained through FPCA which was a more elaborate, flexible and data driven approach. If they correlated well it would help advocate the use of our model.

In a separate analysis and final method of validation we compared model fits from the final multiple-component cosinor model to a spline model. The purpose of this was to determine if the cosinor model was capable of capturing fluctuations in BP as well as a cubic spline model which is less restricted and can provide a flexible fit. We implemented a random-effects model cubic spline model with four knots at 18:00, 24:00, 04:00 and 08:00 allowing all terms to be random similar to our cosinor model. It was agreed that allowing four knots gave sufficient flexibility in the curve to capture BP pattern. Model fits were visually compared to the cosinor model.

Target Organ Damage

Finally, after the model has been compared to FPCA and as a method to illustrate the approach outlined we run the final model separately on those with and without evidence of microalbuminuria. Subject-specific curves are obtained and plotted for each group. In addition, mean curves for both groups, over the 24h period are overlaid on the same plot giving a graphical comparison. We also obtain and compare first derivative curves in both groups.

Software

All analysis including bootstrapping and simulations were implemented in R [187]. Although the model can be rewritten in a linear form [196], when *n* (number of cosine terms) is greater than 1 it is easier to obtain estimates of the parameters directly from the non-linear model as opposed to calculating them post-hoc using trigonometry with the linear model. For this reason the nonlinear mixed-effects model was solved using the nlme command in the nlme R-package [188]. Initial starting values were obtained from a model incorporating fixed effects only. FPCA was utilised using the refund R-package [197]. A shiny app was also built in R using the shiny R-package [198] to illustrate different model fits that can be obtained from our final random-effects model using simulated data based on the model.

6.5 Results

The study questionnaire and physical examination was completed by 2047 participants (response rate: 67%). ABPM was completed by 1207 participants (response rate: 58%), of whom 886 had a minimum of 20 day and 7 night measurements and no data missing for more than two consecutive hourly intervals. Their main clinical characteristics are presented in Table 6-1.

Figure 6-1 is a graphical representation of ABPM for four individuals. Included are subject-specific fits from a random-effects cosinor model with varying number of cosine terms; single, two and three-component models (n=1,2,3). It can be seen that the single component model offers a very simplistic curve that struggles to capture the shape of the data. There is a large improvement observed however with both the two and three-component models where large fluctuations are accounted for more than in the restricted single model. When visually comparing all the two and three-component model fits, there was little difference between them.

For this reason and to obtain the most parsimonious model we identified the twocomponent as a satisfactory model to describe ABPM.

The final two-component SBP cosinor model parameters estimates are presented in Table 6-2. Initially all the parameters were included as fixed-effects. A significant improvement in fit was observed when additionally including each term individually as a random-effect, based on a LRT (all p<0.001). As a consequence we included all parameters as random-effects. The MESOR or average BP over 24h was 124 mmHg. The values of the parameters for the first and second cosine curves are presented separately. As expected, the first cosine with period 24h is the dominant curve with amplitude of 13.2 mmHg while the second cosine curve with period 12h has amplitude 5.6 mmHg. Similarly, the phase shift of the first cosine is larger than the second one which are both measured from 12:00 in units of 30mins, 5.3 (2.6h) compared to 1.0 (0.5h), which corresponds to a time of approximately 14:18 and 12:30 respectively. Exploring the random-effects covariance matrix suggests that there is a moderate positive correlation between the two amplitudes (r=0.51) and the two phase shifts (r=0.44). However, there is a weak correlation between the amplitudes and their corresponding phase shifts (-0.01 and -0.16). As some correlations were quite low we considered reducing the size of the covariance matrix by removing weak correlation terms but as Harrell et al. suggest removing separate terms in this way provides very little gains in terms of precision and power [199]. As a result the covariance structure was not altered. The variation betweenindividuals in the first cosine amplitude was greater compared to the second (SD, 6.1 mmHg vs 2.8 mmHg). However in contrast, there was less variation in the phase shift of the first cosine compared to the second (9mins vs 18mins). The withinsubject SD (σ) was 11.9 mmHg. Examining the ACF plot indicated that the inclusion of an AR1 residual structure adequately accounted for the auto-correlation in the data. Incorporating an AR1 structure resulted in a significant improvement in fit (p=0.22, p<0.001).

After applying the final two-component cosinor model to the data and obtaining the random-effects coefficients, derivatives of the function were calculated. Figure 6-2 presents ABPM readings for three individuals with their fitted subject-specific

trajectories and corresponding first derivative curves. This represents the rate of change or slope at each time point during the day. By focusing on the morning period we can obtain the magnitude and location (time) of the maximum surge. The estimate of the maximum slope obtained from the derivatives from our final model was 2.857 mmHg/30min, Table 6-3. Also presented in Table 6-3 is the bias corrected bootstrap distribution estimates from which a standard error for the estimate of average maximum slope could be obtained (SE=0.0012), resulting in 95% CI: 2.855-2.858 mmHg/30min. The simulation results allowed us to quantify the between-individual SD in maximum slopes which was 1.01. The distribution of the maximum slopes from the original model and the bootstrap estimates were also obtained which provided evidence of a normal distribution (see Appendix F).

As a separate analysis we compared a two-component model to a spline model as outlined previously. Model fits and derivative plots from a random-effects cubic spline model indicated a similar pattern giving further justification of the use of two-component cosinor model (see Appendix F).

B-splines were explored in FPCA. From a visual inspection of the scree plot, three principle components were retained (see Appendix F). Results indicated that the first three FPCs which accounted for 76.3% (FPC1), 9.2% (FPC2) and 6.8% (FPC3) accounted for 92.3% of total variation in the data. To help visualise and interpret the individual FPCs, Figure 6-4 illustrates the mean curve along with the effects to the pattern when a small amount of the component is added and subtracted from the mean. It is evident that the first component which accounts for the majority of the variance in the data represents a relatively constant shift in the mean. Individuals with positive scores on the second component have a slightly higher BP during the day and a lower BP at night indicating a large peak-to-tough value, those who score negatively have a slightly lower value during the day and higher value at night indicating a small peak-to-trough. It can be argued that this component is capturing dippers and non-dippers. Individuals with positive scores on the third component are associated with an earlier dip at night and a large morning rise, those with a low value seem to be shifted to the right and have a slightly less

pronounced morning rise. A larger version of Figure 6-4 can be found in the appendix with the first six FPCs included (see Appendix F). The correlation matrix presented in Figure 6-5 shows strong correlations between the FPC scores and the individual random-effects from the two-component cosinor model. It demonstrates, as expected that the first principle component score summarizes the MESOR (mean curve). The second principle component score has a strong correlation with both amplitudes especially the first one (r=0.9). Similarly the third principle component score has a strong correlation with both applitudes as the strong correlation with both phases especially the first one (r=0.9). A similar correlation matrix including all the FPC scores is included in Appendix F.

The model was applied separately to those with and without evidence of microalbuminuria. A graphical comparison of the mean curves and their associated first-derivatives are presented in Figure 6-6. It can be seen that, on the average over 24h, those with microalbuminuria had higher SBP but the patterns were similar. As a result, the overall pattern in the rate of change over time is broadly similar in both groups. Although significantly lower in those with microalbuminuria, the difference in the maximum surge reached in the morning period between both groups was small, (2.6 vs 2.3 mm Hg/30mins, p<0.01). The time of the maximum surge reached in the morning first derivative or surge, a difference of 9 mmHg in average SBP was observed between those with and without microalbuminuria (129 vs 120 mmHg, p<0.01).

In addition, we have created a simple shiny app to illustrate different fits our model can provide to the data. We have simulated data based on our model and fits and derivatives are shown to different simulated individuals. The link to the app is https://user632.shinyapps.io/App_Double_Cosinor/

6.6 Discussion

In this study we have demonstrated that extending the traditional single cosinor to a two-component cosinor in a random-effects model results in a substantial improvement in fit. From our findings, the evidence suggested that the twocomponent model offered similar fits to that of a three-component model and a spline model. In addition, using FPCA we have demonstrated that the main components of variation in the data correlate extremely well with the parameters from our model. By obtaining model derivatives we have demonstrated a novel approach to quantify rate of change of BP throughout the day. This is the first demonstration of the cosinor model to obtain a measure of morning surge. The use of FPCA on ABPM data also offers a novel method to quantify BPV.

Considering the traditional single cosinor model has been the most commonly used approach for the longitudinal analysis of ABPM, further research in an attempt to extend the model seemed warranted. There has been much criticism of the single cosinor method [121, 122, 142] but we have illustrated that the inclusion of just one additional cosine term gives the model substantial flexibility which helps alleviate many of the concerns raised. The main criticism has focused on the unrealistic assumption that diurnal BP follows a simple symmetrical pattern. Unlike the single model, multiple-component models, while retaining the periodic property, can capture local minimum and maximum points as evident in this study. It has been stated that shapes closely approximating a single cosine curve are uncommon which is not under question [122]. We realise that a two-component model does not offer a perfect fit but is much more flexible than a single model and in order to compromise for a parsimonious model and interpretable terms, it offers a good alternative to other methods with fewer parameters and a similar fit (e.g. three-component cosinor).

The use of FPCA enabled us to obtain the main patterns of variation in the data and determine how much each component contributed to the total variation in the data. This offers a novel method to describe variation within BP, that to our knowledge, has not been used before on ABPM data. The use of functional data analysis allows flexible fits to be obtained without the need to pre-specify a model.
Although similar, this is not the same as a traditional random-effects model. FPCA will be inherently more flexible. For example, if we consider a two-component cosinor random-effects model – it would assume a mean and two cosine terms with two different periods, FPCA however will try to estimate the "optimal" functions that explain the variability in the dataset. However, if the mean, a one-period cosine and two-period cosine are close to optimal then the rest of FPCA and the model fitting will be similar: random-effects will be used to estimate the subjectspecific coefficients for each of these functions. The fact that the random-effects from our model correlated well with the FPC scores help justify the use of our model. To elaborate, the two-component model correlates well with the optimal fit obtained through FPCA. Our finding that the first FPC correlates exactly with the mean value is not surprising where other studies have found in practise, the first FPC is essentially a mean shift [132, 200]. Not only do our values correlate, visually examining the effects of the scores on the mean pattern illustrate that the main components refer to a mean shift, peak-to-tough and a shift left to right which is being captured by the parameters of the cosinor model. In fact using FPCA and the cosinor model together complement each other well. Sometimes interpreting FPCs can be difficult and subjective but when it is used in parallel with a cosinor model and results correlate so well it is easier to explain findings. Although not as directly interpretable as a single-cosinor model, the two-component model is still more intuitive than for example, a spline model. An individual's amplitude being made up of the weighted sum of two cosine amplitudes throughout the day is more intuitive than an arbitrary spline coefficient. In addition, the MESOR which is the average over 24h represents the most important parameter of BP.

Of the definitions identified in the recent meta-analysis examining the prognostic significance of morning surge, only one used a more advanced technique than simply obtaining differences between arbitrary night and day averages [193]. Head *et al.* developed a six parameter double logistic model which is characterised by a day and night plateau of variable length, an independent slope for the fall and rise over the day and a midpoint for each transition [194, 195]. The double logistic model can be used to obtain the rate of rising during the morning period and the

power of the morning surge which is the derivative of the curve multiplied by the amplitude [201]. However, it has been suggested that the parametric structure of the model is very simple and because of the day and night plateaus important BP fluctuations may be averaged out [142]. The approach does however offer alternative morning BP measures that attempt to incorporate more data in a mathematical model than traditional methods and Head *et al.* have tried to refute some of the criticism. They argue that their model is quite flexible and can follow a single-component cosinor model, a saw tooth shape in either direction as well as a square wave-like shape [143]. There are, however, some limitations. Importantly, they stress that the model cannot capture complex fluctuations associated with multiple-component cosinor models and can miss short-term peaks. Research suggests that more complex patterns such as the ones obtained through our approach are not necessarily an advantage as it may be difficult for an investigator to obtain a coherent picture from wavy curves [143]. This may be true if the sole purpose is to decipher individual model fits. We argue however the main purpose of our method is to obtain an estimate of morning slope through derivative estimation and this requires capturing the most accurate curve possible while simultaneously obtaining interpretable parameters that describe it. Obtaining the balance between obtaining complex curves and interpretable parameters is difficult. Complex wavy curves will not alter our ability to describe in simple terms the rate in change or surge. In addition their analysis has focused on analysing each individual ABPM curve one by one which will result in inflated standard errors for their estimates unlike our model that incorporates random-effects.

The use of estimated derivatives in medical research can often offer new, intuitive clinical markers [202, 203]. In the context of modelling ABPM, considering the emphasis is on exploring curves, the use of derivatives is surprisingly rare. As highlighted we have demonstrated that obtaining derivatives from the cosinor model offers a novel method of determining morning slope that has not to the best of our knowledge been implemented elsewhere. An advantage of this approach is it is not restricted to the analysis of morning slope but can be used to obtain critical points throughout the day e.g. dip. We have focused our analysis in this study to

morning BP due to the substantial literature surrounding its potential prognostic significance and the debate surrounding its quantification [23, 45, 46, 193]. Morning parameters to date have focused on summary measures, usually the difference between a pre-awaking value and post-waking value [193]. The primary issue with this approach is that by definition, a morning surge represents a spike or rate of increase in BP during this period which is not accurately captured by differences between two time points. We propose that our method which specifically obtains a rate of change parameter is a better estimate. In fact Parati et al. argue that a tangent with the steepest slope to a curve from a model that accurately captures morning BP could be the most appropriate estimate of the morning rise in BP [142]. Provided the model fits the data well, estimates derived from it should be more robust which in turn will lead to more precise inferences being made to outcomes. Despite our comprehensive knowledge of BP, obtaining new measures and methods to model BP remains a crucial research priority which may help advance our understanding of different aspects of BPV including morning surge [52, 66]. This may be particularly useful in clinical trials where we may be able to provide evidence that a new antihypertensive medication outperforms another in a way we would not traditionally be able to detect e.g. their mean values may indicate no difference. In this study, we illustrated how the model can be compared in two groups using those with and without evidence of microalbuminuria as an example. Although the microalbuminuria group had higher SBP, there was little difference in the pattern of the curves. This was similar to findings observed in Chapter 5 when we examined the same data using a piecewise linear model. Using that approach, we found no evidence to suggest the overall shape differed, although the microalbuminuria group had significantly higher SBP. In addition, the slope parameter that was obtained for the morning period using the piecewise approach (2.21 mmHg/30mins) is similar to the cosinor approach (2.86 mmHg/30mins). It must be noted that the cosinor method obtained a maximum value while the piecewise approach obtained an average value. A comparison of both methods using hard endpoints (cardiovascular events) as opposed to a surrogate marker of CVD is recommended in future work.

There are a number of limitations to the study. Primarily, we note the model does not give the optimal fit to the data (although the parameters correlated well with the more flexible FPCA). In fact, it is acknowledged that it is extremely difficult for any model to capture all the features of a 24h BP profile simultaneously. Perhaps the chosen model should be dependent on the research question posed. For instance as highlighted complex models may not be the most intuitive to understand directly but if the purpose is to obtain a BP measure (e.g. morning surge, dip) from the model post-hoc through additional analyses (derivatives), an initial simple model may not necessarily be the best choice. Another limitation of the study was that we were not able to include the effect of antihypertensive medication in the analysis. Although we knew if a participant was on treatment, we had insufficient data on the specific class of antihypertensive medication the individual was prescribed which meant drug-class comparisons were not possible.

In conclusion, we have demonstrated a simple method to obtain a measure of morning BP surge using a random-effects multiple-component cosinor where our focus was not only at a group level but also the individual level. In addition to the ability of the model to obtain estimates for the morning BP, we derived derivatives of the circadian curves which allow us to locate and quantify the magnitude of other slopes at critical points on the trajectory. The approach offers novel alternative methods of quantifying new BP indices that may be useful in the exploration of BPV where there remains debate over its prognostic significance. The use of FPCA also offers a new alternative approach to quantify BPV. Considering the single-component cosinor has been the most common method of analysis for ABPM, a recommendation for future studies from the evidence presented in this study is to incorporate a second cosine in the context of a random-effects model. The method offers a substantial improvement in fit compared to the traditional cosinor that is capable of capturing short-term peaks and can be implemented in standard statistical software. Future studies should also investigate the clinical prognostic significance of the morning surge parameter obtained through the analysis outlined in this study.

| Characteristic | Total (n=2047) | ABPM (sub-sample) |
|--|-----------------|-------------------|
| | 10tal (11–2047) | Total (n=886) |
| Age, y | 59.8 (5.5) | 59.9 (5.5) |
| Gender, Male n(%) | 1008 (49.2) | 401 (45.3) |
| BMI, n (%) | | |
| Underweight/normal (<25kg/m ²) | 447 (21.9) | 195 (22.0) |
| Overweight (25-30 kg/m ²) | 925 (45.3) | 380 (42.9) |
| Obese (≥30 kg/m²) | 668 (32.8) | 310 (35.0) |
| Office SBP, mm Hg | 129.6 (16.9) | 134.7 (17.7) |
| Office DBP, mm Hg | 80.1 (9.8) | 83.1 (10.2) |
| Hypertension, n (%) | 951 (46.5) | 528 (59.7) |

Table 6-1 Baseline Characteristics

Data are mean (SD). BMI:Body mass index, ABPM: Ambulatory blood pressure monitor. Hypertension: ≥140/90 mmHg and/or on antihypertensive treatment.



Figure 6-1 ABPM readings (circles, thin black line) of four individuals along with predicted subject-specific trajectories from a random-effects model as a function of time using (i) single cosinor (thick black line) (ii) two-component cosinor (thick red line) and (iii) three-component cosinor (thick blue line) models.

| Parameter | Model |
|-----------------------------|---------------------------|
| Fixed effects | Estimate (SE) |
| 24h MESOR (mmHg) | 124 (0.44)* |
| First Cosine (24h period): | |
| Amplitude (mmHg) | 13.2 (0.23)* |
| Phase shift (30mins) | 5.3 (0.02)* |
| Time of phase shift | 14:18 |
| Second Cosine (12h period): | |
| Amplitude (mmHg) | 5.6 (0.14)* |
| Phase shift (30mins) | 1.0 (0.03)* |
| Time of phase shift | 12:30 |
| Random-effects | |
| | 172.3 |
| | 0.18 37.0 |
| Σ | -0.03 -0.01 0.1 |
| | 0.30 0.51 -0.14 7.8 |
| | -0.03 0.01 0.44 -0.16 0.4 |
| σ | 11.9 |
| ρ | 0.22 |

Table 6-2 Model parameter estimates (SBP) along with corresponding correlations and variances

*P<0.001, Random-effects matrix shown has variances on the diagonal and correlation coefficients on off-diagonals. Phase shift measured from 12:00 noon. Time presented in 24h clock.



Figure 6-2 ABPM readings of three individuals with fitted subject-specific trajectories from a two-component cosinor random-effects model (left panels). Their corresponding rate of change curves (first derivatives) are also plotted on right panels (red line indicating reference zero mark).

| Table 6-3 | Maximum | Morning Sur | ge (mmHg | /30 mins) |
|-----------|---------------|-------------|------------|-------------|
| Table 0-3 | IVIAAIIIIUIII | worning Sur | ge (inning | / 30111113/ |

| | Median | Mean | Variance/CI |
|--------------------------|--------|-------|--------------------|
| Original model | 2.779 | 2.857 | 0.994 |
| Simulations (1000) | 2.840 | 2.840 | 1.040 |
| Bias Corrected Bootstrap | 2.857 | 2.857 | CI (2.855 - 2.858) |



Figure 6-3 Histograms of maximum morning slope by simulations



Figure 6-4 FPCA: Each of the first three FPCs as variations about the mean along with the percentage of total variation explained by the component. The solid black line represents the mean SBP over the day and the functions obtained by adding and subtracting \pm SD of the eigenfunctions to the mean. Plus signs indicate addition and minus signs indicate subtraction.



Figure 6-5 Scatter plots and the corresponding correlations between the two-component random-effects cosinor model parameters and the first three FPC scores from FPCA.



Figure 6-6 A two-component cosinor random-effects model implemented separately on those with and without presence of microalbuminuria. Subject-specific curves for those with (black lines) and without (light grey lines) evidence of microalbuminuria are also displayed. Red (microalbuminuria) and blue (no microalbuminuria) lines represent average curves for both groups. The corresponding first derivative curves indicating the rate of change over the day for both groups are also presented.

7. SHORT REPORT: REPRODUCIBILITY

J. M. MADDEN^{a†}

A. M. O'FLYNN^{a†}

P. M. KEARNEY^a

E. DOLAN^b

A. P. FITZGERALD^{a,c}

^a Department of Epidemiology & Public Health, University College Cork, Ireland

^b Stroke and Hypertension unit, Connolly Hospital, Blanchardstown, Dublin, Ireland

^c Department of Statistics, University College Cork, Ireland

[†] Both authors contributed equally to this work

7.1 Introduction

The use of ABPM provides the ability to classify individuals into different hypertension categories that cannot be achieved with a clinic BP reading. The reproducibility of these hypertension categories has been examined with inconsistent results [204-207]. Of particular interest is nocturnal hypertension which is a predictor of CVD and mortality [15, 16, 208]. However, the limited reproducibility of night-time BP patterns is recognised [209] and using absolute BP categories rather than dipping status may be more reproducible [210]. Yet, of those with isolated nocturnal hypertension (defined as nocturnal hypertension without daytime hypertension) at baseline only one third retained this pattern after 2 to 4 years in one small study [211]. This was again examined more recently in a community-based sample of adults (n=282) with similar findings where isolated nocturnal hypertension had poor reproducibility four weeks apart [212].

The majority of studies examining changes in ABPM readings between two timepoints have focused on mean values and hypertension categories. Few longitudinal studies explore how variability changes between two ABPM readings. McDonald *et al.* found that while mean BP did not change, BPV, measured by day/night/24h SD and coefficient of variation, were significantly higher at 10 year follow-up compared to baseline in a community cohort of older people (n=83, median age 70 years) [213]. The percentage of patients taking antihypertensive medication increased from 46% at baseline to 69% at follow-up which may explain no change in mean BP. Conversely, in a community sample of 162 individuals aged 55 to 80 years, Goldstein *et al.* found that mean BP increased while day and night variability measured by SD decreased over two time-points 5 years apart [214]. Although there were no individuals diagnosed with hypertension at baseline, 14 subjects had BP with hypertensive levels. At baseline, no participant was on antihypertensive medication while there were 14 at follow-up.

In addition to exploring changes in hypertension categories and variability, we are interested in the reproducibility of the overall shape of the BP trajectories. Examining changes to variability measures only allows us to quantify changes in

fluctuation about the circadian rhythm but does not inform us if the pattern is changing. For example, an individual's BP trajectory may follow a similar pattern when measured at a second time-point but the variability about the pattern could increase or decrease. The purpose of this study is to outline an approach that can be used to comprehensively interrogate the reproducibility of ABPM readings over time. Changes in hypertension categories and variability measures are explored between two ABPM readings four years apart. In addition, parameters from a twocomponent cosinor model at each time-point are compared as a method of comparing changes in the shape of the circadian rhythm.

7.2 Methods

Baseline ABPM was obtained from the Mitchelstown Study, a population based study of middle-aged men and women, recruited in Ireland [74]. ABPM measurements were performed using the MEDITECH ABPM-05 in 2010 and data was stored using the dabl ABPM system (dabl ltd., Ireland). Based on the initial ABPM results the sample was divided into 4 groups: normotension, isolated nocturnal hypertension, isolated daytime hypertension and day-night hypertension [81]. Twenty participants were randomly selected from each group and invited to attend for follow-up ABPM measurements in 2014 using the Spacelabs 90217 monitor. Data was stored using the Spacelabs 92506 Ambulatory BP Report Management System software.

For both time-points, the monitors were programmed to record the BP every 30min throughout the 24h period. Participants kept a diary of the times they went to bed and got up. Diary times were used to calculate mean daytime and night-time BP. Mean 24h BP was calculated as the mean of all the readings throughout the 24h period. Use of antihypertensive medication was also recorded at both time-points.

Night-time BP was categorised by dipping status as follows:

- (1) Dipping pattern: 10 to 20% fall in night-time SBP
- (2) Non-dipping pattern: <10% fall in night-time SBP
- (3) Extreme dipping pattern: >20% fall in night-time SBP

(4) Reverse dipping pattern: Rise in night-time SBP

BP was also categorised based on the absolute BP levels into four groups:

(1) Normotension: Daytime BP <135/85mmHg and night-time BP <120/70mmHg

(2) Isolated daytime hypertension: Daytime BP ≥135/85mmHg and night-time BP <120/70mmHg</p>

(3) Isolated nocturnal hypertension: Daytime BP <135/85mmHg and night-time BP ≥120/70mmHg

(4) Sustained day-night hypertension: Daytime BP≥135/85mmHg and night-time BP≥120/70mmHg

As indices of short-term BPV, we estimated the SD over 24h, wSD, coefficient of variation ((24h BP SD / 24h BP mean)*100) and ARV. The wSD is the mean of day and night standard deviation values corrected for the number of hours included in each of these two sub-periods. The ARV averages the absolute differences between consecutive measurements. Additional indices of SD, coefficient of variation, and ARV were calculated for both the day and night periods.

7.3 Statistical Analysis

Data are presented as the arithmetic mean (SD). The difference between the two ABPM results was assessed using a paired t-test. The reproducibility of night-time dipping status and absolute BP patterns were assessed using Cohen's kappa statistic. Agreement between the measurements was plotted using the Bland-Altman method. The per cent change in variability measures between the two timepoints was calculated as the difference between the measures divided by the initial value. Changes of less than 5% were regarded as reproducible [205]. To quantify changes in the shape of BP as opposed to variability, a two-component randomeffects cosinor model (see Chapter 6) was utilised on both time-points separately. Comparing model parameters between time-points identified differences in the shape of the curve. Empirical Bayes estimates were obtained for both time-points for each individual and overlaid on the same plot giving a graphical comparison of the curves.

In addition, to determine how much of the variation in the difference between parameters over the two time-points can be attributed to measurement error, a simulation was performed:

- (i) A random-effects cosinor model based on the data from the first timepoint (2010) was used to estimate the mean parameter values, the variance covariance matrix of the random-effects, and the variance of the error term.
- Subject-specific average curves were determined based on a random sample (n=47, same as the original sample size) from the estimated parameter coefficient distribution.
- (iii) Observed values at 30min intervals were generated by adding random error to the values generated in (ii)
- (iv) Step (iii) was repeated to generate a second dataset

The two simulated datasets were then analysed separately representing two different time-points. As before, the difference between model parameters was obtained. However, using the simulated datasets, any differences observed was due to measurement error. Thus, by comparing this variation to the variation observed between the actual readings taken at the two time-points, we can determine what percentage of the total variation is due to measurement error. This was repeated for n=1000 participants to examine if sample size effected the differences observed. Using the simulated data, a similar approach was used to explore how much of the variation in the differences in BPV summary measures over time could be attributed to measurement error. All analysis was implemented in R.

7.4 Results

At baseline, 1207 participants (response rate: 58%) underwent ABPM. At four years follow-up 80 participants were invited to attend for repeat ABPM and fifty

(response rate: 63%) of these participants consented to participation in the follow up study. We excluded three participants from the current analysis due to incomplete follow-up ABPM data giving a sample of 47 for this study. The mean period of follow up was 3.9 years. Raw SBP data for four individuals over 24h for the two time-points is presented in Figure 7-1, plots for all individuals are included in the Appendix G. Overall mean BPs were similar in 2010 and 2014, Table 7-1. The correlation between age and mean SBP was the same in 2010 and 2014 (r=0.17). Agreement between the two time-points for BP is given by the Bland-Altman plots in Figure 7-2. The reproducibility of BP profiles categorised by dipping status was low at 24% with a kappa statistic of -0.11 (p = 0. 89) while reproducibility based on categorisation by absolute BP was fair at 40% with a kappa statistic of 0.21 (p < 0.005), Table 7-2 and Table 7-3.

Overall, compared to baseline, BPV at follow-up had reduced, Table 7-4. However, only SBP variability, measured by wSD, SD and ARV, over 24h and the night period had fallen significantly. The Bland-Altman plots for both 24h SD and ARV are given in Figure 7-3, where a large spread is observed. Reproducibility of 24h SD (17%), ARV (10%) and wSD (23%) was poor. Parameters from a two-component cosinor model were obtained for both time-points, Table 7-5. With the exception of the second phase, all cosinor parameters were similar four years later. The variation in the difference in MESOR between time-points that could be attributed to measurement error based on simulations was low (3%), Table 7-6. Measurement error explained 19% and 32% of the variation in the difference in amplitudes, between the two-points, for the first and second cosines respectively. However, measurement error explained a larger amount of the total variation in the differences in the first (62%) and second phase shifts (88%). Repeating the simulations based on 1000 participants made little difference to our findings, Table 7-6. Bland-Altman agreement plots for the various parameters are presented Figure 7-4 and Figure 7-5. Reproducibility of MESOR (36%), first amplitude (14%), first phase shift (51%), second amplitude (21%) and second phase (4%) was poor. A separate boxplot of the differences between the parameters is given in Figure 7-6 showing a relatively even spread in the data. Using the simulated data,

measurement error between time-points accounted for 79%, 22% and 41% of the total variation in the differences of 24h ARV, SD and wSD respectively. Cosinor model fits for six individuals is provided in Figure 7-7 highlighting how the overall shape of the trajectory for two ABPM readings can easily be compared by overlaying the curves.

The use of antihypertensive medications increased between 2010 and 2014 from 42% (n=21) to 61% (n=30). In analysis limited to those not on medication (n=17) the kappa statistics were -0.2 (p = 0.9) for dipping status and 0.43 (p < 0.001) for absolute BP categories. Mean BPs for those not on medication were similar in 2010 and 2014. Moreover, there was no change in variability measures for individuals not on medication. However, for those on medication, variability measured by wSD, SD and ARV, over 24h (all SBP) remained significantly reduced while the mean BP was similar.

7.5 Discussion

We have outlined methods to compare two ABPM readings taken at separate occasions. No one method can fully capture the reproducibility of a circadian BP trajectory but we have explored approaches that can account for the main features including the comparison of mean and variability values, hypertension categorization and a comparison of the shape of the curves. Our findings demonstrate the limited reproducibility of night-time BP profiles with poor reproducibility of dipping status and only fair reproducibility measured by wSD, SD and ARV (all SBP only) significantly reduced over 4 years but reproducibility within the measures was fair. Finally, parameters from a two-component cosinor model which can be used to describe the shape of BP trajectories demonstrated that on the average patterns were similar but reproducibility of the parameters was poor.

Long-term changes in ABPM variability have rarely been explored. However, our findings are similar to Goldstein *et al.* who reported a reduction in variability over

two time-points 5 years apart in a similarly aged sample [214]. This is in contrast to McDonald et al. who reported BPV obtained by ABPM increased over a 10 year follow-up in an older sample [213]. Changes in mean SBP variability, despite no significant change in mean SBP is an interesting finding. There has been significant work over the last decade to control mean BP levels, coupled with advancements in antihypertensive therapy and perhaps an absence of an increase reflects these efforts. Much of the debate regarding BPV focuses on antihypertensive medications. While different classes of antihypertensives have similar BP lowering effects; significant differences between classes in their effects of BPV have been observed. Some studies suggest that calcium channel blockers and diuretics are superior to other drugs in reducing BPV and preventing stroke and other vascular events compared to other classes [61, 160]. In our study, the use of antihypertensive medications increased over the four years between the BP measurements. While the effect over time on mean BP was similar irrespective of medication use, BPV differed by medication use. Although numbers were small and hence, power was limited, we were still able to detect a significant reduction in BPV within the group on medication. In contrast, there was no difference in BPV over time for those not on medications. An analysis stratified by type of antihypertensive class was not possible and is recognised as a study limitation.

Overall reproducibility was poor on all features of the circadian pattern. BP is a continuous risk factor [215]. Thresholds define the levels where investigation and treatment do more good than harm [216], and for this reason are important but clinicians and public health professionals need to recognise the limitations of thresholds and that significant numbers of events occur in those below definitions of normal [217].

Using simulations, we were able to obtain the fraction of the total variation in the changes between time-points that was attributable to measurement error alone. If there was no change in the true readings between the two time-points, the only difference observed would be due to random error. Assuming a two-component cosinor model is an adequate fit for ABPM, which has been illustrated in Chapter 6,

we can estimate this random error. Comparing changes in summary BPV measures and model parameters from the two simulated datasets to the changes using the actual measured readings allow us to quantify the influence of measurement error between time-points. This highlights a benefit of using a random-effects model when examining reproducibility where the variance of differences between timepoints can be separated into measurement error and true changes that have occurred during the intervening time-period.

The two-component cosinor model we have outlined provides a method that not only quantitatively assesses differences in the overall shape of two ABPM readings, but also offers a quick graphical comparison that may be useful in clinical practice when a clinician is trying to gauge changes in an individual's underlying circadian rhythm. The inclusion of the predicted curves in ABPM reports, in combination with the standard output may improve clinical decisions and ultimately, BP control.



Figure 7-1 Data from four individuals 2010 vs 2014

| N =47 | | 2010 | 2014 | p-value |
|--------------|----------|------|------|---------|
| Daytime sys | tolic | 133 | 129 | 0.05 |
| Daytime dia | stolic | 79 | 77 | 0.2 |
| Night-time s | systolic | 117 | 114 | 0.1 |
| Night-time | | 66 | 66 | 0.9 |
| Twenty | four | 127 | 123 | 0.05 |
| Twenty | four | 74 | 73 | 0.3 |

Table 7-1 BP levels 2010 v 2014



Figure 7-2 Bland-Altman scatter-plots of agreement of measurements of 24h SBP and 24h DBP four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines represent the 95%CI for the differences in mean values.

| | 2014 | | | | |
|-----------------|---------|---------|---------|---------|-------|
| 2010 | Dippers | Non- | Extreme | Reverse | Total |
| | | dippers | dippers | dippers | |
| Dippers | 5 | 11 | 5 | 1 | 22 |
| Non-dippers | 8 | 3 | 2 | 2 | 15 |
| Extreme | 2 | 3 | 1 | 0 | 6 |
| dippers | | | | | |
| Reverse dippers | 1 | 1 | 0 | 2 | 4 |
| Total | 15 | 18 | 8 | 5 | 47 |

Table 7-2 Dipping status 2010 v 2014

kappa statistic = -0.11 (p = 0. 89)

Table 7-3 BP categories 2010 v 2014

| | 2014 | | | | |
|--------------|-------------|-------------|-------------|-------------|-------|
| 2010 | Isolated | Day/night | Isolated | Normotensio | Total |
| | nocturnal | hypertensio | daytime | n | |
| | hypertensio | n | hypertensio | | |
| | n | | n | | |
| Isolated | 4 | 1 | 0 | 7 | 12 |
| nocturnal | | | | | |
| hypertension | | | | | |
| Day/night | 2 | 3 | 2 | 4 | 11 |
| hypertension | | | | | |
| Isolated | 1 | 6 | 3 | 4 | 14 |
| daytime | | | | | |
| hypertension | | | | | |
| Normotension | 1 | 0 | 0 | 9 | 10 |
| Total | 8 | 10 | 5 | 24 | 47 |

kappa statistic = 0.21 (p < 0.005)

| Variability | 24-hour (mm | Hg) | Day (mm Hg) | | Night (mm H | g) |
|-------------|-------------|--------------|-------------|------------|-------------|--------------|
| Measure | 2010 | 2014 | 2010 | 2014 | 2010 | 2014 |
| ARV SBP | 11.7 (2.5) | 10.4 (2.2)** | 12.2 (3.1) | 11.2 (2.9) | 11.1 (3.5) | 9.2 (2.9)** |
| ARV DBP | 7.8 (2.2) | 7.5 (1.6) | 8.2 (2.5) | 8.0 (2.1) | 7.1 (2.5) | 6.7 (2.2) |
| SD SBP | 15.9 (4.0) | 14.6 (4.1)* | 13.0 (3.0) | 12.5 (3.7) | 12.2 (4.1) | 10.1 (3.7)** |
| SD DBP | 11.3 (3.5) | 10.1 (2.7) | 8.9 (3.0) | 8.8 (2.1) | 8.4 (2.7) | 7.2 (2.6) |
| CV SBP | 12.5 (3.1) | 12.0 (3.2) | 9.8 (2.1) | 9.8 (2.8) | 10.4 (3.3) | 8.8 (2.8) |
| CV DBP | 15.3 (4.4) | 14.1 (4.1) | 11.3 (3.5) | 11.6 (3.2) | 12.8 (4.0) | 11.1 (4.0) |
| wSD SBP | 12.7 (2.7) | 11.7 (3.2)* | - | - | - | - |
| wSD DBP | 8.8 (2.4) | 8.3 (1.8) | - | - | - | - |

Table 7-4 Variability Indices 2010 v 2014

Abbreviations: ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, SBP: SBP, DBP: Diastolic BP. Data presented as mean(SD). *p<0.05; **p<0.01 represents significance between 2010 and 2014 for each period of the day



Figure 7-3 Bland-Altman scatter-plots of agreement of measurements of 24h SBP SD and 24h SBP ARV four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines represent the 95%CI for the differences in mean values.

| SBP Parameter | Model Estimate (SE) | Model Estimate (SE) | | | |
|--|---|---------------------|--|--|--|
| | Year = 2010 | Year = 2014 | | | |
| 24h MESOR (mmHg) | 127.1 (1.6) | 123.2 (2.0)* | | | |
| First Cosine (24h period): | | | | | |
| Amplitude (mmHg) | 11.4 (1.3) | 9.7 (1.1) | | | |
| Phase shift (30mins) | 5.0 (0.8) | 5.1 (0.1) | | | |
| Time of phase shift | 14:30 | 14.33 | | | |
| Second Cosine (12h period): | | | | | |
| Amplitude (mmHg) | 3.6 (0.6) | 3.5 (0.7) | | | |
| Phase shift (30mins) | 0.6 (0.1) | 1.6 (0.1)* | | | |
| Time of phase shift | 12:18 | 12.48 | | | |
| σ | 12.7 | 11.5 | | | |
| *p<0.05 represents significance between 2010 and 202 | *p<0.05 represents significance between 2010 and 2014, Phase shift measured from 12:00 noon. Time presented in 24h clock. | | | | |

 Table 7-5 Comparison of SBP parameters from two separate two-component random-effects cosinor models (2010 vs 2014)

| SBP Parameter | Total Variation in Differences | Variance due to Measurement Error (% of Total Vari | |
|-----------------------------|--------------------------------|--|---------------------------|
| | | Based on 47 Simulations | Based on 1000 Simulations |
| 24h MESOR (mmHg) | 168.1 | 5.6 (3.3%) | 6.1 (3.6%) |
| First Cosine (24h period): | | | |
| Amplitude (mmHg) | 42.2 | 8.1 (19.2%) | 8.8 (20.9%) |
| Phase shift (30mins) | 2.1 | 1.3 (61.9%) | 1.1 (52.4%) |
| Second Cosine (12h period): | | | |
| Amplitude (mmHg) | 6.6 | 2.1 (31.8%) | 1.9 (28.8%) |
| Phase shift (30mins) | 1.7 | 1.5 (88%) | 0.9 (52.9%) |

Table 7-6 Comparison of SBP parameters from two separate two-component random-effects cosinor models (2010 vs 2014)



Figure 7-4 Bland-Altman scatter-plots of agreement of measurements of the first and second amplitudes four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines represent the 95%CI for the differences in mean values.



Figure 7-5 Bland-Altman scatter-plots of agreement of measurements of the first and second phase shifts four years apart. The x-axis represents the mean of the two measurements and the y-axis represents the difference between them. The black dashed lines represent the mean and limits of agreement. The dashed red lines represent the 95%CI for the differences in mean values.



Figure 7-6 Boxplots of differences between each of the cosinor parameters between the two time-points (2010 minus 2014).



Figure 7-7 Comparison of fits, 2010 vs 2014, for six individuals using two separate two-component random-effects cosinor models.

8. DISCUSSION

This thesis aimed to explore circadian BP patterns by reviewing current and previous approaches for analysing ABPM data and describes novel methodologies that offer new measures of BP and may help us to obtain new insights into BP. The recent debate towards the prognostic significance of BPV and how it is quantified was the original motivating reason to undertake the analysis. The thesis first explores summary measures used to describe fluctuations in circadian patterns and then goes beyond this to obtain other indices through more advanced methods. This chapter outlines the main findings, and strengths and limitations of the thesis. Recommendations for areas of future research and a brief conclusion are also outlined.

8.1 Summary of Main Findings

In Chapter 3, the systematic review collated data to identify current summary measures that are used to quantify BPV and explored their association with the presence of TOD, specifically LVH. Four measures were identified; SD, ARV, wSD and CV. The meta-analysis suggested there is a weak positive correlation, between all measures of BPV and LVMI. Despite recent interest in the subject of BPV the systematic review highlighted the lack of good epidemiological studies exploring the relationship between BPV and LVH. The fact that there were so few studies (n=12) and only data available to perform a meta-analysis of such a weak marker (correlation coefficient) is evidence of this. In addition, a third of studies in the systematic review did not make adjustment for mean BP in any of their analysis which is a major methodological limitation. This is an important issue in the context of BPV: the adjustment for mean BP. The significance of mean BP has been well documented and is not under question but in order to determine if BPV is an additional clinic target it is paramount that any analysis adjusts for average BP.

In Chapter 4, following on from the systematic review, the measures identified were calculated for the Mitchelstown ABPM dataset and their association with subclinical TOD, documented by microalbuminuria and ECG LVH, were examined. The study found no association between any measure of BPV and LVH. Conversely, we found all but one measure (CV) associated with microalbuminuria and remained associated after adjustment for age, gender, smoking, BMI, diabetes and anti-

hypertensive treatment. However, when the models were further adjusted for mean BP the association did not persist for all indices.

Piecewise regression which is a simple but underutilised approach in epidemiological studies [190], allows separate slopes to be fitted to observations representing different periods throughout the day or more specifically periods before or after a "critical point". In Chapter 5, we implemented this technique to our data in the context of a mixed-effects model. This allowed us to obtain new measures of BP. In particular, the method allowed quantification of the steepest rise and fall in BP (along with measures of variation between-individuals for these parameters), which occurred just after waking (2.23 mmHg/30min) and immediately after falling asleep (-1.93 mmHg/30min) respectively. By definition of the model and in the context of the morning period this represents a novel measure of the rate of change of BP or morning surge. The within-subject variation about an individual's curve was also estimated (12.3 mmHg) and represents a measure similar to that of a SD summary measure. Despite a significant difference in mean BP values, we found no evidence that slopes were different within different periods of the day in those with and without TOD, suggesting a similar circadian pattern in both groups.

Cosinor analysis which uses a single cosine to model a circadian BP cycle has been the most common longitudinal method to analyse ABPM data. The method has rarely been extended to include more than one cosine term in the context of a random-effects model. In Chapter 6, we fitted a two-component cosinor randomeffects model to our data and found that it provided a better fit than a single model and a similar fit to a more complex three-component model. Additionally, FPCA was performed on the data and it was determined that the first three principle components accounted for over 90% of the variation in the data. Not only did this offer an alternative method to quantify BPV, it also allowed us to determine if the two-component model was comparable to a more complex data driven approach

that obtained the optimal functions from the data. Our findings indicated that the parameters from the two-component cosinor model were highly correlated with the first three components from the FPCA. By obtaining first-order derivatives of the two-component cosinor model, we outlined a novel approach in the context of BP which can be used to visualise changes over time and, locate and quantify the magnitude of slopes at critical points along the circadian rhythm.

In Chapter 7, approaches to compare two ABPM readings obtained at different occasions were outlined. The purpose was to explore changes in variability over time and examine the reproducibility of patterns of circadian BP curves that go beyond looking at mean values and hypertension categories. To this end, we implemented the two-component model from Chapter 6 and compared parameters from the two time-points. Results indicated that mean levels of BP were similar four years later but SBP variability, measured by wSD, SD and ARV, over 24h and the night period had significantly fallen. When examining the variability measures for those on medication, wSD, SD and ARV, over 24h (all SBP) remained significantly reduced. Interestingly, there was no change in variability measures for individuals not on medication. Overall, the reproducibility of patterns was poor based on mean levels, variability measures, hypertension categories and cosinor parameters. Based on simulations we were also able to determine how much of the total variation in the difference between parameters and summary measures was attributable to measurement error alone.

8.2 Implications of Findings

What summary measure of BPV to use?

We have demonstrated that quantifying variability is difficult and attempting to do this with one summary measure makes the task significantly more challenging. The main disadvantage of using summary measures to quantify BPV is the loss of information as the statistical power associated with longitudinal data is lost by collapsing the data into one value. However, as previously stated, many ABPM software programs routinely include estimates of variability measures such as SD (dabl ABPM system, dabl LTD, Ireland) in their reports and are important reference points considering the ease with which they can be obtained.

We demonstrated that most but not all summary indices of BPV were associated (before adjustment for mean BP) with microalbuminuria (Chapter 4). This reiterates the problem of inconsistences among indices which makes understanding and collating of findings from studies particularly problematic. It is these inconsistences that illustrate how sensitive BPV measurement is and accentuates that each measure is in fact capturing different characteristics of variability. However, compared to other summary measures, we believe ARV is the most suitable index to quantify BPV. To reiterate, this measure is the mean of the absolute differences between successive readings which is independent of the mean value [36]. It has been argued that it is a more reliable representation of time series variability than 24h SD which only captures variability about a mean value [33, 36]. wSD has also been suggested as a better measure to use than 24h SD as it attempts to remove the influence of the day-night difference [33]. To elaborate, the primary reason that a number of studies have suggested SD is a poor measure is because it only reflects the dispersion of measurements around a single value (mean) and does not account for the order in which BP measurements were obtained and the longitudinal variation in the circadian data [36]. However, if we are only considering variability during a specific period i.e. awake or night period separately as opposed to the
whole 24h, the use of SD may not be as problematic as the day-night fall is not present.

What longitudinal model to use?

Parati et al. argue that rather than simplifying ABPM data into mean summary measures, incorporating all the data in more advanced models can lead to more robust estimates of clinically relevant parameters such as dipping status and morning BP surge [142]. In Chapter 2, a number of approaches to model ABPM data were outlined where we focused on the use of a piecewise linear and a cosinor model in subsequent chapters. Both a piecewise linear model and the use of derivative estimation is not new in medical research but in the context of BP they offer a simple but novel approach to analyse its circadian rhythm that help extract features from the pattern. As outlined in the thesis, there have been a number of modelling approaches applied to 24h BP but the ultimate goal remains the development of a physiological model that captures known features of the pattern while not over-fitting the data so that we can obtain clinically relevant parameters [142]. It is widely acknowledged that achieving this is a difficult task and there is no perfect model that captures all the features of the circadian rhythm simultaneously [49, 142-144]. Using a mathematical model over simple summary measures allows us to obtain a smooth predicted curve over the whole 24h period which removes some of the "noise" in the data. Removing noise in data is usually perceived as an advantage where we remove measurement error and the impact of any outliers and are left with the underlying pattern. In the case of BPV however where we are specifically trying to understand and capture fluctuations in BP, perhaps smoothing the data in a model that captures the whole 24h period may not be the optimal approach. The sudden jumps or spikes that are seen in ABPM data may be a response to emotional or psychological stress which can sporadically increase BP values and are therefore important to measure [50]. Although these spikes may be perceived as noise they may also represent an underlining feature of an individual's natural circadian rhythm. Some individuals may be more prone to abrupt changes due to stress or anxiety. Obtaining lower sampling frequency measurements such as beat-to-beat measurements may help in this regard but are not feasible in large

population based studies. We do however have to reach some compromise and perhaps modelling the data in separate segments similar to the piecewise model represents a suitable approach. However, graphically comparing circadian curves obtained at different time-points may be better achieved using the cosinor method which gives a smoother fit where differences in the pattern of the shape can easily be seen. We reiterate the point made in Chapter 2 that no model is optimal in terms of capturing all features of the curve and choosing one involves making a compromise based on what the researcher deems most important for their particular research question. Using mathematical models does allow us to obtain an almost infinite number of BP indices. Comparing these measures between new antihypertensive medications may allow us to determine if a drug outperforms another when there is no significant difference in means values.

Policy and Practice

With the HSE recently acknowledging the importance of ABPM by approving the reimbursement of ABPM in primary care for those with a medical card, obtaining values of BPV is now more accessible [22]. In practice, summary measures will continue to be the most common method of quantifying BPV due to the ease with which they can be calculated coupled with their automatic inclusion in some ABPM output reports. A recommendation to improve monitors and their output would be the inclusion of the ARV index which could be achieved with minimal effort. There have been calls in the most recent ABPM position paper to include all measures of BPV in ABPM research reports but many just include SD [13]. However, there are no current threshold levels for variability in BP guidelines which makes interpretation of BPV measures difficult for clinicians. A long-term goal would be to eventually have standardised cut-points for which levels of BPV would be controlled to, similar to that of mean BP. For this to happen however we will need stronger evidence of the importance of BPV. As highlighted in the position paper, the current threshold values for mean BP levels in the NICE guidelines [17], the JNC 7 guideline [218], and the ESH/ESC guidelines for 2003, 2007 and 2013 [13, 18, 219], and the results of

outcome studies, such as IDACO [220] and Ohasama [221] have contributed to the definition of ABPM consensus values currently being used in practice [13]. A similar level of research output is required to obtain conclusive evidence on the prognostic significance of BPV before we can begin to arrive at suitable variability thresholds.

More advanced methods to model BP and measure BPV have been suggested as particularly useful in research studies rather than directly used in practice [13, 142]. This is where we currently suggest the piecewise and cosinor approaches outlined in this thesis should be primarily implemented where they may be particularly beneficial in clinical trials. However, we suggest that the inclusion of the cosinor approach in an ABPM report could be quite useful when comparing two ABPM patterns from the same individual. As outlined in Chapter 7, it offers a quick graphical comparison that maybe helpful for a clinician when they are trying to gauge changes in an individual's underlying circadian rhythm. We argue the inclusion of the predicted curves in ABPM reports, in combination with the standard output may improve clinical decisions and ultimately, BP control.

8.3 Strengths and Limitations

The strengths and limitations of each paper have been discussed in previous chapters. In this section a summary is provided with a focus on the thesis as a whole.

The thesis inevitably has a number of limitations. Although the use of ABPM is considered the gold standard of BP measurement and is recommended by guidelines, the monitors will not always provide accurate readings. Undoubtedly, there will be some measurement error from the true BP value especially for example, during the night when an individual may be lying on the cuff. In the Mitchelstown cohort every effort was made by study nurses to emphasise the importance of wearing the monitor correctly. In addition, all monitors were recalibrated before the study began as recommended [17]. The physical inconvenience of wearing the monitor may introduce selection bias where certain groups of individuals do not feel comfortable wearing the device. For example, perhaps frailer individuals would be more concerned wearing a monitor than stronger individuals. Wood et al. explored reasons for ABPM uptake among 770 participants of different ethnicities in the United Kingdom [222]. In their qualitative analysis, reasons suggested for poor uptake included discomfort, sleep disruption, stress and embarrassment, and interruption of activities of daily living. In the Mitchelstown study, ABPM was offered to all participants but those with high BP were encouraged by the study nurses to wear a monitor. No qualitative data was recorded for reasons of not availing of ABPM. Despite a response rate of 58%, the ABPM subsample was found to be broadly representative of the full sample based on age, gender, BMI and diabetes prevalence (see Appendix E for details). However, the prevalence of hypertension (≥140/90 mmHg and/or on antihypertensive treatment) was higher in the baseline ABPM subsample (46% vs 59%). This introduces a selection bias based on hypertension only where there is a systematic difference between the subsample and the main sample. This may affect interpretation of the results where caution must be exercised when discussing the generalizability of the results to the population. It must be stressed that the only difference was that based on hypertension and this does not necessarily affect the

validity of the comparisons and inferences made within the study or the study's internal validity [223, 224].

Although a lot of the work in this thesis involved exploring the shape of circadian BP curves, Chapter 3 and 4 directly examined the association between BPV and TOD. A statistically significant association between an exposure (BPV measure) and an outcome (TOD) is not sufficient to imply causality. Hill's casual considerations offer guidance [225]. He listed nine considerations, none of which are essential, that are used in epidemiology to build up evidence for a causal relationship: strength, consistency, temporality, dose-response, plausibility, reversibility, coherence, analogy and specificity [223]. The strength of the association in this study between BPV and TOD was weak and did not persist after adjustment for mean BP. The systematic review examining the relationship between BPV and LVMI suggested a weak association and also provides evidence of poor consistency among studies. Temporality in this study is difficult to ascertain. BPV leading to the development of TOD and in turn CVD is the hypothesis made in most of the literature but perhaps as argued by Hansen et al., as TOD is a forerunner of cardiovascular complications, BPV will inevitably increase [168]. This raises the question of reverse causality with increased BPV being a marker of underlying disease rather than being an independent predictor.

Although the precise mechanisms responsible for BPV are not entirely understood [50], there are a number of acknowledged factors that influence variations in BP which have been postulated as to why BPV may cause the development of TOD, and ultimately CVD. Parati *et al.* [51, 52] argue BPV is primarily modulated by neural (increased central sympathetic drive and reduced arterial and cardiopulmonary reflexes), humoral (angiotensin II, insulin, bradykinin, nitric oxide) and vascular effects (elastic properties of arteries). Variability has been shown to be correlated with arterial stiffness and peripheral vascular disease which is a blood circulation disorder that causes blood vessels outside of your heart and brain to narrow, block or spasm which can lead to CVD [25]. Rothwell *et al.* suggest that BPV could lead to both altered central autonomic control of BP and an increased risk of stroke [25].

Reversibility refers to whether an intervention to remove or reduce the exposure results in the elimination or reduction of the outcome [223]. We cannot assess this from our results. A clinical trial comparing the impact of different antihypertensive classes on BPV and ultimately TOD would help to address this. Coherence is also difficult to assess based on the evidence in this study. However, experimental data from animal models lend support to a causal link. It was previously shown that experimental sinoaortic denervation in rats, which increases variability in BP without changing mean BP, caused LVH and aortic vasoconstriction [24, 226]. Analogy refers to the similarity of the effect of interest to other established cause-effect relationships that would help support the argument for causality [223]. In our case, elevated mean BP has long been established as a causal factor for the development of TOD and CVD. As a result, it could be argued that small or large fluctuations in BP may also contribute to CVD but this is speculative. In conclusion, a causal relationship between BPV and TOD is difficult to prove based on the findings in this study and the literature available at present.

Another limitation of the thesis is the use of TOD as surrogate markers of CVD rather than the use of hard endpoints. The reason for this was simply the feasibility and time constraints of the thesis where only data from wave one is currently collected. Although hard endpoints were recorded in the study (MI, stroke), the number of events were small which precluded using them. Data collection for wave two is currently on-going but we may need to wait for subsequent waves to have adequate number of events to detect an association, if one exists.

Due to the nature of short-term BPV, we are examining a small time frame, only 24h. Ideally, we would like participants to continue to wear the monitors for a number of days or perhaps obtain separate ABPM recordings taken a few days apart to be sure of reliable observations. This would also allow the comprehensive analysis of the reproducibility of day-to-day circadian patterns. However, the practicalities and feasibility of wearing a monitor for a prolonged period of time in this cohort study meant we were unable to obtain more readings.

As highlighted in the thesis one of the key aspects of BPV has been the suggestion that different classes of antihypertensive medications have different effects in terms of increasing or reducing variability. Emerging evidence suggests that those receiving either CCBs or diuretics, alone or in addition to other drugs have significantly lower variability compared with ACEIs, ARBs or β -blockers alone or in combination [61-63, 65, 227]. A limitation of the study was that we were not able to include the effect of antihypertensive medication in the analysis. Although we knew if a participant was on treatment, we had insufficient data on the specific class of antihypertensive medication the individual was prescribed which meant drug-class comparisons were not possible. As one of the questions in the study questionnaire asked if the participant was taking antihypertensive drugs, it allowed analysis to be stratified by those taking and not taking medications. Findings were broadly similar across both groups (Chapter 4, 5). It could be the case that there are differences between classes but by analysing them all together the effect has been diluted. When examining changes to short-term variability over time (Chapter 7), evidence suggested that BPV in those on medication significantly reduced while there was no change in variability measures for individuals not on medication.

Despite the limitations outlined, the thesis has several strengths. Specifically, it addresses a relevant and highly topical area in BP research. To date, two of the chapters have been published in peer reviewed scientific journals while a third is under review. The work is both warranted and appropriate to increase awareness and to help develop a deeper understanding of the mechanisms responsible for BPV which are not entirely known. Methods to analyse circadian BP patterns that account for the longitudinal nature of the data are sparse and this thesis outlines new approaches. The thesis includes the first review to our knowledge that quantifies the correlation between BPV and LVMI which identifies a research gap where stronger epidemiological studies are needed in the area. To the best of our knowledge, the thesis is the first to apply a piecewise linear model and FPCA to

diurnal BP over 24h. We have demonstrated that incorporating more cosine terms into the cosinor model offers a substantial improvement in fit compared to the traditional method of ABPM analysis, the single-component model. One of the main strengths of the study lie in the robust community based design and large sample of ABPM data recorded. Although the inferences that are made may not be applicable across all age groups and populations there are no reasons why the methods outlined cannot be used on them to obtain valid estimates of BP. This argument can be extended where many of the chapters and methods are not limited to the exploration of BP but are flexible enough to be applied to other medical data that follows a circadian rhythm.

8.4 Future Recommendations for Research

As we have alluded to, it is evident that more evidence from large longitudinal studies examining the prognostic significance of short-term BPV is needed before suitable variability thresholds can be introduced. There is a need for standardisation of indices that we use to define BPV. This has been highlighted by a recent systematic review of BPV examining its prognostic value for all-cause mortality, cardiovascular mortality, all cardiovascular events, stroke and coronary heart disease [191]. Taylor et al. argue that the interpretation and use of 24h BPV in clinical practice, as a prognostic indicator of cardiovascular events is hampered by insufficient evidence and divergent methodologies [191]. Agreement and standardisation of indices may not be fully possible for a number of years until there is a plateauing effect in the number of methods used to quantify variability. We have however, made the recommendation to use the ARV index as the primary summary measure of BPV. Rather than using different indices, if all future studies included this measure it would make collating results easier. Indeed this thesis has developed yet further methods to explore circadian patters but as Dolan et al. argue each new measure helps to bring about new insights and is required to advance our understanding of BP [66]. We are still however, searching for the optimal measure to quantify variability over a short-period; this is as much a mathematical problem as it is clinical. This as the thesis has shown, is not easily achieved.

In the context of the single-component cosinor model which has been the most common method of analysis for ABPM, a recommendation for future studies from the evidence presented in this thesis is to incorporate a second cosine in the context of a random-effects model. The method offers a substantial improvement in fit compared to the traditional cosinor that is capable of capturing short-term peaks and can be implemented in standard statistical software.

In Chapter 7, we found that short-term BPV measured at two occasions, four years apart, decreased in a small sample. Few longitudinal studies explore how variability

changes between two ABPM readings. In those that have, inconsistent findings were reported [213, 214]. More research into how short-term variability changes over time is warranted and further studies exploring the reproducibility of ABPM patterns is needed. It must be remembered that the emphasis in this thesis has solely been on short-term patterns and not long-term variability which represents how much BP fluctuates between visits (months or years).

The prognostic significance of visit-to-visit BPV seems to be more transparent than in the case of short-term BPV. The most recent meta-analysis of 77,299 patients suggests that visit-to-visit SD was significantly predictive of all-cause mortality, cardiovascular mortality and stroke, after adjustment for mean BP [26]. As there is usually less data-points when analysing long-term compared to short-term data there is less opportunity to apply more advanced techniques. Studies have primarily used SD as a measure of variability. This has helped in collating data from studies, although it may not be the best measure of BPV. However, the issues associated with measuring short-term BPV may not be as evident for long-term BPV. The primary reason is that long-term variability does not have a cyclic component to it and there is no dipping effect like ABPM. Long-term variability usually refers to fluctuations in clinic BP values which have no night component. In this case, the use of SD as a measure of variation may be adequate. Another recommendation would be to investigate the association between short-term BPV and the development of long-term BPV which there is no data to the best of our knowledge. Perhaps longterm BPV is the cumulative aggregate of short-term BPV but we will not know the answer to this until large longitudinal studies are conducted. Perhaps obtaining multiple ABPM readings over the course of a short period e.g. one month, may yield the same information and benefits of running a long-term study over a couple of years which would save on time and reduce financial costs.

In context of the work presented in this thesis and following on from the limitation of examining surrogate markers, the most obvious recommendation is to explore the methods presented in the thesis with long-term follow up data that have hard endpoints e.g. cardiovascular events. The second wave of data collection is currently on-going for the Mitchelstown cohort. However, in order to have

adequate number of events to detect an association, if one exists; data from subsequent waves will probably have to be obtained.

The benefit of our findings may lie in the analysis of clinical trial data, particularly in the analysis of chronotherapy effects. The methods offer estimates of BP at important times of the day that define a rate of change. This could be applied after the ingestion of different antihypertensive drugs to identify which medication produced the quickest fall in BP levels. Debate is on-going into the benefits of nighttime administration of antihypertensive medication compared to daytime administration but there has been substantial evidence in its favour [67-70]. The rationale behind the argument for this seems both plausible and reasonable. That is, administering medication at night will help pre-empt the magnitude of the morning surge by reducing BP during the night so that on waking, the surge is beginning at a lower value and thus not reaching as high a peak. When medication is ingested at night there should also be a larger reduction in night-time BP which has been highlighted as a stronger predictor of outcome than 24h mean. The latest on-going Treatment in Morning versus Evening (TIME) trial with 10,200 patients followed for 5 years is anticipated to provide definitive evidence of whether or not there is a benefit to administering antihypertensive medications in the evening to provide better protection against major adverse cardiovascular events [228, 229].

The methods outlined in this thesis are not restricted to the analysis of BP but can be applied to the many other physiological processes that follow a circadian rhythm such as cortisol, heart rate etc. In this regard it would be interesting to know if the methods, especially the more advanced approaches could be of use in analysing and answering questions in other medical fields. Indeed, it could be the case that the methods outlined could be more suited to important aspects of a particular circadian process other than BP. For purposes of dissemination of methods and particularly to act as a source for other researchers who are interested in extending on the work in the thesis, sample R code has been included in Appendix H. This may also be useful for researchers in other fields' aside from BP. Relevant R packages utilised are also included along with the version of R used. The run time for the nonlinear model is also presented

8.5 Conclusion

This thesis has demonstrated different approaches to explore circadian BP patterns over a 24h period. Results indicated that most summary measures of BPV were associated with microalbuminuria but the association did not persist after adjustment for mean BP. There was no association between BPV and LVH. Where more advanced models were used incorporating all the data, the rate of increase or fall throughout different periods over 24h remained the same between those with and without TOD. A novel method to quantify morning surge was also presented which may be useful in future studies. FPCA may also be a novel method to determine major components of variations in 24h BP. BPV quantification, its predictive value and potential as a therapeutic target will remain controversial until more studies are conducted and an agreement on how best to accurately quantify BPV has been reached. The thesis, which had a large methods and statistical component, illustrated novel techniques that can be implemented with standard software and whose application may also be useful for analysing other medical datasets that follows a circadian process.

9. REFERENCES:

1. Mozaffarian D, Benjamin EJ, Go AS, Arnett DK, Blaha MJ, Cushman M, et al. Heart disease and stroke statistics--2015 update: a report from the American Heart Association. Circulation. 2015;131(4):e29-322.

2. Danaei G, Finucane MM, Lin JK, Singh GM, Paciorek CJ, Cowan MJ, et al. National, regional, and global trends in systolic blood pressure since 1980: systematic analysis of health examination surveys and epidemiological studies with 786 country-years and 5.4 million participants. Lancet. 2011;377(9765):568-77.

3. Lim SS, Vos T, Flaxman AD, Danaei G, Shibuya K, Adair-Rohani H, et al. A comparative risk assessment of burden of disease and injury attributable to 67 risk factors and risk factor clusters in 21 regions, 1990-2010: a systematic analysis for the Global Burden of Disease Study 2010. Lancet. 2012;380(9859):2224-60.

4. Mills KT, Bundy JD, Kelly TN, Reed JE, Kearney PM, Reynolds K, et al. Global Disparities of Hypertension Prevalence and Control: A Systematic Analysis of Population-Based Studies From 90 Countries. Circulation. 2016;134(6):441-50.

5. Balanda KP, Barron, S., Fahy, L., McLaughlin, A. Making Chronic Conditions Count: Hypertension, Coronary Heart Disease, Stroke, Diabetes. A systematic approach to estimating and forecasting population prevalence on the island of Ireland. Dublin: Institute of Public Health in Ireland. 2010.

6. O'Brien E. First Thomas Pickering memorial lecture*: ambulatory blood pressure measurement is essential for the management of hypertension. J Clin Hypertens (Greenwich) 2012 Dec;14(12):836-47 doi: 101111.

7. O'Brien E, Sheridan J, O'Malley K. Dippers and non-dippers. Lancet. 1988;2(8607):397.

8. Murphy C, Kearney, PM, Shelley, EB, Fahey, T,Dooley, C,Kenny, R. A. Hypertension prevalence, awareness, treatment and control in the over 50s in Ireland: evidence from The Irish Longitudinal Study on Ageing. J Public Health (Oxf) 2015 Apr 28 pii: fdv057.

9. Wright JT, Jr., Williamson JD, Whelton PK, Snyder JK, Sink KM, Rocco MV, et al. A Randomized Trial of Intensive versus Standard Blood-Pressure Control. N Engl J Med 2015 Nov 26;373(22):2103-16 doi: 101056.

10. Cushman WC, Whelton PK, Fine LJ, Wright JT, Jr., Reboussin DM, Johnson KC, et al. SPRINT Trial Results: Latest News in Hypertension Management. Hypertension. 2016;67(2):263-5.

11. Kjeldsen SE, Narkiewicz K, Hedner T, Mancia G. The SPRINT study: Outcome may be driven by difference in diuretic treatment demasking heart failure and study

design may support systolic blood pressure target below 140 mmHg rather than below 120 mmHg. Blood Press. 2016;25(2):63-6.

12. Xie X, Atkins E, Lv J, Bennett A, Neal B, Ninomiya T, et al. Effects of intensive blood pressure lowering on cardiovascular and renal outcomes: updated systematic review and meta-analysis. Lancet. 2016;387(10017):435-43.

13. O'Brien E, Parati G, Stergiou G, Asmar R, Beilin L, Bilo G, et al. European Society of Hypertension position paper on ambulatory blood pressure monitoring. J Hypertens. 2013;31(9):1731-68.

14. Dolan E, Stanton A, Thijs L, Hinedi K, Atkins N, McClory S, et al. Superiority of ambulatory over clinic blood pressure measurement in predicting mortality: the Dublin outcome study. Hypertension. 2005;46(1):156-61.

15. Fagard RH, Celis H, Thijs L, Staessen JA, Clement DL, De Buyzere ML, et al. Daytime and nighttime blood pressure as predictors of death and cause-specific cardiovascular events in hypertension. Hypertension. 2008;51(1):55-61.

16. O'Brien E. Dipping comes of age: the importance of nocturnal blood pressure. Hypertension. 2009;53(3):446-7.

17. National Institute for Health and Clinical Excellence: Guidance. Hypertension: The Clinical Management of Primary Hypertension in Adults: Update of Clinical Guidelines 18 and 34. London: Royal College of Physicians (UK) National Clinical Guideline Centre.; 2011.

18. Mancia G, Fagard R, Narkiewicz K, Redon J, Zanchetti A, Bohm M, et al. 2013 ESH/ESC Guidelines for the management of arterial hypertension: the Task Force for the management of arterial hypertension of the European Society of Hypertension (ESH) and of the European Society of Cardiology (ESC). J Hypertens. 2013;31(7):1281-357.

19. O'Brien E. Why Is It That We Continue to Deny Patients Ambulatory Blood Pressure Monitoring? Hypertension. 2016;67(3):484-7.

20. O'Brien E, Dolan E, Atkins N. Failure to Provide ABPM to All Hypertensive Patients Amounts to Medical Ineptitude. J Clin Hypertens (Greenwich). 2015;17(6):462-5.

21. Changing Cardiovascular Health National Cardiovascular Strategy 2010–2019. Dublin: Department of Health and Children, 2010.

22. Gantly D, O'Brien, E. New STC payment will help with BP control. Irish Medical Times. March 14th, 2016.

23. Kario K. Essential Manual of 24 Hour Blood Pressure Management: From morning to nocturnal hypertension: Wiley-Blackwell; 2015.

24. Rothwell PM. Limitations of the usual blood-pressure hypothesis and importance of variability, instability, and episodic hypertension. Lancet. 2010;375(9718):938-48.

25. Rothwell PM, Howard SC, Dolan E, O'Brien E, Dobson JE, Dahlof B, et al. Prognostic significance of visit-to-visit variability, maximum systolic blood pressure, and episodic hypertension. Lancet. 2010;375(9718):895-905.

26. Tai C, Sun Y, Dai N, Xu D, Chen W, Wang J, et al. Prognostic significance of visit-to-visit systolic blood pressure variability: a meta-analysis of 77,299 patients. J Clin Hypertens (Greenwich). 2015;17(2):107-15.

27. Parati G, Pomidossi G, Albini F, Malaspina D, Mancia G. Relationship of 24hour blood pressure mean and variability to severity of target-organ damage in hypertension. J Hypertens. 1987;5(1):93-8.

28. Leoncini G, Viazzi F, Storace G, Deferrari G, Pontremoli R. Blood pressure variability and multiple organ damage in primary hypertension. J Hum Hypertens. 2013;27(11):663-70.

29. Sega R, Corrao G, Bombelli M, Beltrame L, Facchetti R, Grassi G, et al. Blood pressure variability and organ damage in a general population: results from the PAMELA study (Pressioni Arteriose Monitorate E Loro Associazioni). Hypertension. 2002;39(2 Pt 2):710-4.

30. Tatasciore A, Renda G, Zimarino M, Soccio M, Bilo G, Parati G, et al. Awake systolic blood pressure variability correlates with target-organ damage in hypertensive subjects. Hypertension. 2007;50(2):325-32.

31. Zakopoulos NA, Tsivgoulis G, Barlas G, Spengos K, Manios E, Ikonomidis I, et al. Impact of the time rate of blood pressure variation on left ventricular mass. J Hypertens. 2006;24(10):2071-7.

32. Kikuya M, Hozawa A, Ohokubo T, Tsuji I, Michimata M, Matsubara M, et al. Prognostic significance of blood pressure and heart rate variabilities: the Ohasama study. Hypertension. 2000;36(5):901-6.

33. Hansen TW, Thijs L, Li Y, Boggia J, Kikuya M, Bjorklund-Bodegard K, et al. Prognostic value of reading-to-reading blood pressure variability over 24 hours in 8938 subjects from 11 populations. Hypertension. 2010;55(4):1049-57.

34. Schillaci G, Verdecchia P, Borgioni C, Ciucci A, Porcellati C. Lack of association between blood pressure variability and left ventricular mass in essential hypertension. Am J Hypertens. 1998;11(5):515-22.

35. Wei FF, Li Y, Zhang L, Xu TY, Ding FH, Wang JG, et al. Beat-to-beat, readingto-reading, and day-to-day blood pressure variability in relation to organ damage in untreated chinese. Hypertension. 2014;63(4):790-6. 36. Mena L, Pintos S, Queipo NV, Aizpurua JA, Maestre G, Sulbaran T. A reliable index for the prognostic significance of blood pressure variability. J Hypertens. 2005;23(3):505-11.

37. Pierdomenico SD, Di Nicola M, Esposito AL, Di Mascio R, Ballone E, Lapenna D, et al. Prognostic value of different indices of blood pressure variability in hypertensive patients. Am J Hypertens. 2009;22(8):842-7.

38. Muntner P, Shimbo D, Tonelli M, Reynolds K, Arnett DK, Oparil S. The relationship between visit-to-visit variability in systolic blood pressure and all-cause mortality in the general population: findings from NHANES III, 1988 to 1994. Hypertension. 2011;57(2):160-6.

39. Roman MJ, Pickering TG, Schwartz JE, Pini R, Devereux RB. Relation of blood pressure variability to carotid atherosclerosis and carotid artery and left ventricular hypertrophy. Arterioscler Thromb Vasc Biol. 2001;21(9):1507-11.

40. Bilo G, Giglio A, Styczkiewicz K, Caldara G, Maronati A, Kawecka-Jaszcz K, et al. A new method for assessing 24-h blood pressure variability after excluding the contribution of nocturnal blood pressure fall. J Hypertens. 2007;25(10):2058-66.

41. Mule G, Calcaterra I, Costanzo M, Geraci G, Guarino L, Foraci AC, et al. Relationship Between Short-Term Blood Pressure Variability and Subclinical Renal Damage in Essential Hypertensive Patients. J Clin Hypertens (Greenwich). 2015;17(6):473-80.

42. Mule G, Calcaterra I, Costanzo M, Morreale M, D'Ignoto F, Castiglia A, et al. Average real variability of 24-h systolic blood pressure is associated with microalbuminuria in patients with primary hypertension. J Hum Hypertens. 2016;30(3):164-70.

43. Muller JE, Tofler GH, Stone PH. Circadian variation and triggers of onset of acute cardiovascular disease. Circulation. 1989;79(4):733-43.

44. Scheer FA, Shea SA. Human circadian system causes a morning peak in prothrombotic plasminogen activator inhibitor-1 (PAI-1) independent of the sleep/wake cycle. Blood. 2014;123(4):590-3.

45. Kario K. Morning surge in blood pressure and cardiovascular risk: evidence and perspectives. Hypertension. 2010;56(5):765-73.

46. Kario K, Pickering TG, Umeda Y, Hoshide S, Hoshide Y, Morinari M, et al. Morning surge in blood pressure as a predictor of silent and clinical cerebrovascular disease in elderly hypertensives: a prospective study. Circulation. 2003;107(10):1401-6.

47. Polonia J, Amado P, Barbosa L, Nazare J, Silva JA, Bertoquini S, et al. Morning rise, morning surge and daytime variability of blood pressure and cardiovascular

target organ damage. A cross-sectional study in 743 subjects. Revista Portuguesa de Cardiologia. 2005;24(1):65-78.

48. Lambert PC, Abrams KR, Jones DR, Halligan AW, Shennan A. Analysis of ambulatory blood pressure monitor data using a hierarchical model incorporating restricted cubic splines and heterogeneous within-subject variances. Stat Med. 2001;20(24):3789-805.

49. Edwards LJ, Simpson SL. An analysis of 24-h ambulatory blood pressure monitoring data using orthonormal polynomials in the linear mixed model. Blood Press Monit. 2014;19(3):153-63.

50. Mancia G, Grassi G. Mechanisms and clinical implications of blood pressure variability. J Cardiovasc Pharmacol. 2000;35(7 Suppl 4):S15-9.

51. Parati G, Ochoa JE, Lombardi C, Bilo G. Assessment and management of blood-pressure variability. Nat Rev Cardiol. 2013;10(3):143-55.

52. Parati G, Ochoa JE, Lombardi C, Bilo G. Blood pressure variability: assessment, predictive value, and potential as a therapeutic target. Curr Hypertens Rep. 2015;17(4):537.

53. Kario K, Pickering TG, Hoshide S, Eguchi K, Ishikawa J, Morinari M, et al. Morning blood pressure surge and hypertensive cerebrovascular disease: role of the alpha adrenergic sympathetic nervous system. Am J Hypertens. 2004;17(8):668-75.

54. Somers VK, Dyken ME, Mark AL, Abboud FM. Sympathetic-nerve activity during sleep in normal subjects. N Engl J Med. 1993;328(5):303-7.

55. Mancia G. Short- and long-term blood pressure variability: present and future. Hypertension. 2012;60(2):512-7.

56. Bilo G, Villafuerte FC, Faini A, Anza-Ramirez C, Revera M, Giuliano A, et al. Ambulatory blood pressure in untreated and treated hypertensive patients at high altitude: the High Altitude Cardiovascular Research-Andes study. Hypertension. 2015;65(6):1266-72.

57. Aubiniere-Robb L, Jeemon P, Hastie CE, Patel RK, McCallum L, Morrison D, et al. Blood pressure response to patterns of weather fluctuations and effect on mortality. Hypertension. 2013;62(1):190-6.

58. Dahlof B, Sever PS, Poulter NR, Wedel H, Beevers DG, Caulfield M, et al. Prevention of cardiovascular events with an antihypertensive regimen of amlodipine adding perindopril as required versus atenolol adding bendroflumethiazide as required, in the Anglo-Scandinavian Cardiac Outcomes Trial-Blood Pressure Lowering Arm (ASCOT-BPLA): a multicentre randomised controlled trial. Lancet. 2005;366(9489):895-906. 59. Dolan E, Stanton AV, Thom S, Caulfield M, Atkins N, McInnes G, et al. Ambulatory blood pressure monitoring predicts cardiovascular events in treated hypertensive patients--an Anglo-Scandinavian cardiac outcomes trial substudy. J Hypertens. 2009;27(4):876-85.

60. Williams B, Lacy PS, Thom SM, Cruickshank K, Stanton A, Collier D, et al. Differential impact of blood pressure-lowering drugs on central aortic pressure and clinical outcomes: principal results of the Conduit Artery Function Evaluation (CAFE) study. Circulation. 2006;113(9):1213-25.

61. Webb AJ, Fischer U, Mehta Z, Rothwell PM. Effects of antihypertensive-drug class on interindividual variation in blood pressure and risk of stroke: a systematic review and meta-analysis. Lancet. 2010;375(9718):906-15.

62. Webb AJ, Fischer U, Rothwell PM. Effects of beta-blocker selectivity on blood pressure variability and stroke: a systematic review. Neurology. 2011;77(8):731-7.

63. Zhang Y, Agnoletti D, Safar ME, Blacher J. Effect of antihypertensive agents on blood pressure variability: the Natrilix SR versus candesartan and amlodipine in the reduction of systolic blood pressure in hypertensive patients (X-CELLENT) study. Hypertension. 2011;58(2):155-60.

64. Levi-Marpillat N, MacQuin-Mavier I, Tropeano AI, Parati G, Maison P. Antihypertensive drug classes have different effects on short-term blood pressure variability in essential hypertension. Hypertension Research. 2014;37(6):585-90.

65. Umemoto S, Ogihara T, Matsuzaki M, Rakugi H, Ohashi Y, Saruta T. Effects of calcium channel blocker-based combinations on intra-individual blood pressure variability: post hoc analysis of the COPE trial. Hypertens Res. 2016;39(1):46-53.

66. Dolan E, O'Brien E. Is It Daily, Monthly, or Yearly Blood Pressure Variability that Enhances Cardiovascular Risk? Curr Cardiol Rep. 2015;17(11):93.

67. Hermida RC, Ayala DE, Mojon A, Fernandez JR. Influence of circadian time of hypertension treatment on cardiovascular risk: results of the MAPEC study. Chronobiol Int. 2010;27(8):1629-51.

68. Hermida RC, Ayala DE, Smolensky MH, Fernandez JR, Mojon A, Portaluppi F. Chronotherapy with conventional blood pressure medications improves management of hypertension and reduces cardiovascular and stroke risks. Hypertens Res. 2016;39(5):277-92.

69. Hermida RC, Ayala DE, Mojon A, Fernandez JR. Bedtime ingestion of hypertension medications reduces the risk of new-onset type 2 diabetes: a randomised controlled trial. Diabetologia. 2016;59(2):255-65.

70. Hermida RC, Ayala DE, Fernandez JR, Calvo C. Chronotherapy improves blood pressure control and reverts the nondipper pattern in patients with resistant hypertension. Hypertension. 2008;51(1):69-76.

71. Stranges PM, Drew AM, Rafferty P, Shuster JE, Brooks AD. Treatment of hypertension with chronotherapy: is it time of drug administration? Ann Pharmacother. 2015;49(3):323-34.

72. Zhao P, Xu P, Wan C, Wang Z. Evening versus morning dosing regimen drug therapy for hypertension. Cochrane Database Syst Rev. 2011 (10):Cd004184.

73. Hermida RC, Smolensky MH, Ayala DE, Portaluppi F. 2013 ambulatory blood pressure monitoring recommendations for the diagnosis of adult hypertension, assessment of cardiovascular and other hypertension-associated risk, and attainment of therapeutic goals. Chronobiol Int. 2013;30(3):355-410.

74. Kearney PM, Harrington JM, Mc Carthy VJ, Fitzgerald AP, Perry IJ. Cohort profile: The Cork and Kerry Diabetes and Heart Disease Study. Int J Epidemiol. 2013;42(5):1253-62.

75. Devereux RB, Alderman MH. Role of preclinical cardiovascular disease in the evolution from risk factor exposure to development of morbid events. Circulation. 1993;88(4 Pt 1):1444-55.

76. Bakris GL. Microalbuminuria: what is it? Why is it important? What should be done about it? J Clin Hypertens (Greenwich). 2001;3(2):99-102.

77. Toto RD. Microalbuminuria: definition, detection, and clinical significance. J Clin Hypertens (Greenwich). 2004;6(11 Suppl 3):2-7.

78. Matsushita K, van der Velde M, Astor BC, Woodward M, Levey AS, de Jong PE, et al. Association of estimated glomerular filtration rate and albuminuria with all-cause and cardiovascular mortality in general population cohorts: a collaborative meta-analysis. Lancet. 2010;375(9731):2073-81.

79. Katholi RE, Couri DM. Left ventricular hypertrophy: major risk factor in patients with hypertension: update and practical clinical applications. Int J Hypertens. 2011;2011:495349.

80. Dahlof B, Devereux RB, Julius S, Kjeldsen SE, Beevers G, de Faire U, et al. Characteristics of 9194 patients with left ventricular hypertrophy: the LIFE study. Losartan Intervention For Endpoint Reduction in Hypertension. Hypertension. 1998;32(6):989-97.

81. O'Flynn A, Ho, E., Dolan, E., Curtin, RJ., Kearney, PM The association of night-time systolic blood pressure with ultrasound markers of subclinical cardiac and vascular disease. Blood Press Monit (2016 In Press).

82. Gibbons RD, Hedeker D, DuToit S. Advances in analysis of longitudinal data. Annu Rev Clin Psychol. 2010;6:79-107.

83. Zeger SL, Liang KY. Longitudinal data analysis for discrete and continuous outcomes. Biometrics. 1986;42(1):121-30.

84. Zeger SL, Liang KY, Albert PS. Models for longitudinal data: a generalized estimating equation approach. Biometrics. 1988;44(4):1049-60.

85. Liang KY, Zeger, S. L. Longitudinal data analysis using generalized linear models. Biometrika. 1986;73:13-22.

86. Goldstein H. Multilevel mixed linear model analysis using iterative generalized least squares. Biometrika. 1986:43–56.

87. Laird NM, Ware JH. Random-effects models for longitudinal data. Biometrics. 1982;38(4):963-74.

88. Fitzmaurice G, Laird, NM., Ware, JH. Applied Longitudinal Analysis: John Wiley & Sons: New Jersey; 2011.

89. Twisk JWR. Applied Longitudinal Data Analysis for Epidemiology. A Practical Guide. Cambridge; 2013.

90. Hedeker D, Gibbons, R. D. Longitudinal Data Analysis: Wiley; 2006.

91. Howe LD, Tilling K, Matijasevich A, Petherick ES, Santos AC, Fairley L, et al. Linear spline multilevel models for summarising childhood growth trajectories: A guide to their application using examples from five birth cohorts. Stat Methods Med Res. 2013

92. Liu X. Methods and Applications of Longitudinal Data Analysis: Elsevier; 2015.

93. Zuur A, Ieno, EN., Walker, NJ., Saveliev, AA., Smith, GM. Mixed Effects Models and Extensions in Ecology with R: Springer; 2009.

94. Galecki A, Burzykowski, T. Linear Mixed-Effects Models Using R: A Step-by-Step Approach: Springer; 2011.

95. West B, Welch KB., Galecki, AG. Linear Mixed Models: A Practical Guide Using Statistical Software: Chapman & Hall/CRC; 2007.

96. Ghisletta P, Spini, D. . An introduction to generalized estimating equations and an application to assess selectivity effects in a longitudinal study on very old individuals. Journal of Educational and Behavioral Statistics. 2004;29(4):421-37.

97. Huber PJ. The behavior of maximum likelihood estimates under nonstandard conditions. . Proceedings of the Fifth Berkeley Symposium on

Mathematical Statistics and Probability. 1967;Vol 1, pp 221-233, University of California Press, Berkeley, Calif.

98. White H. Maximum likelihood estimation of misspecified models. Econometrica. 1982;50:1-26.

99. Twisk J, de Boer M, de Vente W, Heymans M. Multiple imputation of missing values was not necessary before performing a longitudinal mixed-model analysis. J Clin Epidemiol. 2013;66(9):1022-8.

100. Jones RH, Boadi-Boateng F. Unequally spaced longitudinal data with AR(1) serial correlation. Biometrics. 1991;47(1):161-75.

101. Armitage P, Berry, G., Matthews, JNS. Statistical Methods in Medical Research: Blackwell Publishing; 2009.

102. Healy M. Fitting a quadratic. Biometrics. 1963;19:362-3.

103. Corrao S. SR, Arnone S, et al. Analysis of 24-h noninvasive ambulatory blood pressure profiles by a third-degree polynomial approach. J Cardiovasc Diagn Proc. 1996;13(4)::237-42.

104. Zwinderman AH, Cleophas TA, Cleophas TJ, van der Wall EE. Polynomial analysis of ambulatory blood pressure measurements. Neth Heart J. 2001;9(2):68-74.

105. Wang Y. Smoothing Splines Methods and Applications: CRC Press; 2011.

106. Selwyn MR, Difranco DM. The application of large Gaussian mixed models to the analysis of 24 hour ambulatory blood pressure monitoring data in clinical trials. Stat Med. 1993;12(18):1665-82.

107. Bornhorst C, Tilling K, Russo P, Kourides Y, Michels N, Molnar D, et al. Associations between early body mass index trajectories and later metabolic risk factors in European children: the IDEFICS study. Eur J Epidemiol. 2015

108. Patel R, Tilling K, Lawlor DA, Howe LD, Bogdanovich N, Matush L, et al. Socioeconomic differences in childhood length/height trajectories in a middle-income country: a cohort study. BMC Public Health. 2014;14:932.

109. Macdonald-Wallis C, Tilling K, Fraser A, Nelson SM, Lawlor DA. Associations of blood pressure change in pregnancy with fetal growth and gestational age at delivery: findings from a prospective cohort. Hypertension. 2014;64(1):36-44.

110. Marsh L, Cormier, DR. Spline Regression Models: Sage Publications; Thousand Oaks, CA; 2001.

111. Ben-Shlomo Y, McCarthy A, Hughes R, Tilling K, Davies D, Smith GD. Immediate postnatal growth is associated with blood pressure in young adulthood: the Barry Caerphilly Growth Study. Hypertension. 2008;52(4):638-44.

112. Halberg F TY, Johnson EA. Circadian System Phase – An Aspect of Temporal Morphology; Procedures and Illustrative Examples. In Proc International Congress of Anatomists The Cellular Aspects of Biorhythms, Symposium on Biorhythms Edited by Mayersbach HV :20–48.: New York: Springer-Verlag;; 1967.

113. Halberg F. Chronobiology. Annu Rev Physiol. 1969;31:675-725.

114. Portaluppi F, Montanari L. Consistency of circadian blood pressure pattern assessed by non-invasive monitoring and cosinor analysis in hospitalized hypertensive patients. Acta Cardiol. 1988;43(5):605-13.

115. Portaluppi F, Bagni B, degli Uberti E, Montanari L, Cavallini R, Trasforini G, et al. Circadian rhythms of atrial natriuretic peptide, renin, aldosterone, cortisol, blood pressure and heart rate in normal and hypertensive subjects. J Hypertens. 1990;8(1):85-95.

116. Munakata M, Imai Y, Hashimoto J, Sakuma H, Sekino H, Abe K, et al. The influence of antihypertensive agents on circadian rhythms of blood pressure and heart rate in patients with essential hypertension. Tohoku J Exp Med. 1992;166(2):217-27.

117. Ayala DE, Hermida RC, Mojon A, Fernandez JR, Iglesias M. Circadian blood pressure variability in healthy and complicated pregnancies. Hypertension. 1997;30(3 Pt 2):603-10.

118. Kelley K, Light RP, Agarwal R. Trended cosinor change model for analyzing hemodynamic rhythm patterns in hemodialysis patients. Hypertension. 2007;50(1):143-50.

119. Streitberg B, Meyer-Sabellek W, Baumgart P. Statistical analysis of circadian blood pressure recordings in controlled clinical trials. J Hypertens Suppl. 1989;7(3):S11-7.

120. Wang Y, Brown MB. A flexible model for human circadian rhythms. Biometrics. 1996;52(2):588-96.

121. Wang Y, Ke C, Brown MB. Shape-invariant modeling of circadian rhythms with random effects and smoothing spline ANOVA decompositions. Biometrics. 2003;59(4):804-12.

122. Reilly T, Atkinson, G., and Waterhouse, J. Biological Rhythms and Exercise: New York: Oxford; 1997. 151 p.

123. Somes GW, Harshfield GA, Arheart KL, Miller ST. A Fourier series approach for comparing groups of subjects on ambulatory blood pressure patterns. Stat Med. 1994;13(12):1201-10.

124. Staessen JA, Fagard R, Thijs L, Amery A. Fourier analysis of blood pressure profiles. Am J Hypertens. 1993;6(6 Pt 2):184s-7s.

125. Hermida RC, Ayala DE, Mojon A, Fernandez JR, Alonso I, Silva I, et al. Blood pressure patterns in normal pregnancy, gestational hypertension, and preeclampsia. Hypertension. 2000;36(2):149-58.

126. van Rijn-Bikker PC, Snelder N, Ackaert O, van Hest RM, Ploeger BA, van Montfrans GA, et al. Nonlinear mixed effects modeling of the diurnal blood pressure profile in a multiracial population. Am J Hypertens. 2013;26(9):1103-13.

127. Fourier Series ndi, viewed 9 June 2016, <<u>http://tex.stackexchange.com/questions/127375/replicate-the-fourier-transform-time-frequency-domains-correspondence-illustrati></u>.

128. Ramsey J, and Silverman, BW. Functional Data Analysis. Springer. 2005.

129. Ramsey J, Hooker, G, Graves, S. Functional Data Analysis with R and MATLAB. Springer. 2009.

130. Mairesse O, De Valck E, Quanten S, Neu D, Cortoos A, Pattyn N, et al. Sleepiness phenomics: modeling individual differences in subjective sleepiness profiles. Int J Psychophysiol. 2014;93(1):150-61.

131. Goldsmith J, Liu X, Jacobson JS, Rundle A. New Insights into Activity Patterns in Children, Found Using Functional Data Analyses. Med Sci Sports Exerc. 2016;48(9):1723-9.

132. Di CZ, Crainiceanu CM, Caffo BS, Punjabi NM. MULTILEVEL FUNCTIONAL PRINCIPAL COMPONENT ANALYSIS. Ann Appl Stat. 2009;3(1):458-88.

133. Simpkin AJ, Metcalfe C, Martin RM, Lane JA, Donovan JL, Hamdy FC, et al. Longitudinal prostate-specific antigen reference ranges: Choosing the underlying model of age-related changes. Stat Methods Med Res 2013, doi: 101177/0962280213503928.

134. Yao F MHaWJ. Functional data analysis for sparse longitudinal data. J Am Stat Assoc. 2005;100:577–90.

135. Degaute JP, van de Borne P, Linkowski P, Van Cauter E. Quantitative analysis of the 24-hour blood pressure and heart rate patterns in young men. Hypertension. 1991;18(2):199-210.

136. Van Cauter E. Method for characterization of 24-hour temporal variation of blood components. Am J Physiol. 1979;237:E255–264.

137. Stanton A, Cox J, Atkins N, O'Malley K, O'Brien E. Cumulative sums in quantifying circadian blood pressure patterns. Hypertension. 1992;19(1):93-101.

138. Beath KJ. Infant growth modelling using a shape invariant model with random effects. Stat Med. 2007;26(12):2547-64.

139. Cole TJ, Cortina-Borja M, Sandhu J, Kelly FP, Pan H. Nonlinear growth generates age changes in the moments of the frequency distribution: the example of height in puberty. Biostatistics. 2008;9(1):159-71.

140. Cole TJ, Donaldson MD, Ben-Shlomo Y. SITAR--a useful instrument for growth curve analysis. Int J Epidemiol. 2010;39(6):1558-66.

141. Cole TJ, Ahmed ML, Preece MA, Hindmarsh P, Dunger DB. The relationship between Insulin-like Growth Factor 1, sex steroids and timing of the pubertal growth spurt. Clin Endocrinol (Oxf). 2015;82(6):862-9.

142. Parati G, Vrijens B, Vincze G. Analysis and interpretation of 24-h blood pressure profiles: appropriate mathematical models may yield deeper understanding. Am J Hypertens. 2008;21(2):123-5; discussion 7-9.

143. Head GA, Lukoshkova EV, Reid CM. Response to "Analysis and Interpretation of 24-h Blood Pressure Profiles". Am J Hypertens. 2008;21(2):127-9.

144. Parati G. Assessing circadian blood pressure and heart rate changes: advantages and limitations of different methods of mathematical modelling. J Hypertens. 2004;22(11):2061-4.

145. Warlow C, Sudlow C, Dennis M, Wardlaw J, Sandercock P. Stroke. Lancet. 2003;362(9391):1211-24.

146. Parati G, Schumacher H. Blood pressure variability over 24 h: prognostic implications and treatment perspectives. An assessment using the smoothness index with telmisartan-amlodipine monotherapy and combination. Hypertens Res. 2014;37(3):187-93.

147. Sibiya MJ, Norton GR, Hodson B, Redelinghuys M, Maseko MJ, Majane OH, et al. Gender-specific contribution of aortic augmentation index to variations in left ventricular mass index in a community sample of African ancestry. Hypertens Res. 2014;37(11):1021-7.

148. Hayden JA, Cote P, Bombardier C. Evaluation of the quality of prognosis studies in systematic reviews. Ann Intern Med. 2006;144(6):427-37.

149. StataCorp. Stata Statistical Software: Release 12. College Station, TX: StataCorp LP. 2011

150. Ajayi OE, Ajayi EA, Akintomide OA, Adebayo RA, Ogunyemi SA, Oyedeji AT, et al. Ambulatory blood pressure profile and left ventricular geometry in Nigerian hypertensives. Journal of Cardiovascular Disease Research. 2011;2(3):164-71.

151. Zhang QQ, Zhang XJ, Chang BB, Qiu BY, Zhang Y, Li J, et al. [Blood pressure variability correlates with target-organ damage in elderly patients with hypertension]. Sichuan Da Xue Xue Bao Yi Xue Ban. 2011;42(2):252-5.

152. Kristensen KS, Hoegholm A, Bang LE, Gustavsen PH, Poulsen CB. No impact of blood pressure variability on microalbuminuria and left ventricular geometry: Analysis of daytime variation, diurnal variation and 'white coat' effect. Blood Press Monit. 2001;6(3):125-31.

153. Veerman DP, de Blok K, van Montfrans A. Relationship of steady state and ambulatory blood pressure variability to left ventricular mass and urinary albumin excretion in essential hypertension. Am J Hypertens. 1996;9(5):455-60.

154. Pascual JM, Baldo E, Bertolin V, Rovira E, Gonzalvo F, Gonzalez C, et al. [Ambulatory arterial pressure and left ventricular hypertrophy in untreated hypertensive patients]. Med Clin (Barc). 1999;112(5):166-70.

155. Frattola A, Parati G, Cuspidi C, Albini F, Mancia G. Prognostic value of 24hour blood pressure variability. J Hypertens. 1993;11(10):1133-7.

156. Palatini P, Penzo M, Racioppa A, Zugno E, Guzzardi G, Anaclerio M, et al. Clinical relevance of nighttime blood pressure and of daytime blood pressure variability. Arch Intern Med. 1992;152(9):1855-60.

157. Sterne JA, Davey Smith G. Sifting the evidence-what's wrong with significance tests? BMJ. 2001;322(7280):226-31.

158. London GM. Role of arterial wall properties in the pathogenesis of systolic hypertension. Am J Hypertens. 2005;18(1 Pt 2):19s-22s.

159. Hocht C, Bertera FM, Taira CA. Importance of blood pressure variability in the assessment of cardiovascular risk and benefits of antihypertensive therapy. Expert Rev Clin Pharmacol. 2010;3(5):617-21.

160. Dolan E, O'Brien E. Blood pressure variability: clarity for clinical practice. Hypertension. 2010;56(2):179-81.

161. Carlberg B, Lindholm LH. Stroke and blood-pressure variation: new permutations on an old theme. Lancet. 2010;375(9718):867-9.

162. Zanchetti A. Wars, war games, and dead bodies on the battlefield: variations on the theme of blood pressure variability. Stroke. 2011;42(10):2722-4.

163. Okada H, Fukui M, Tanaka M, Matsumoto S, Mineoka Y, Nakanishi N, et al. Visit-to-visit blood pressure variability is a novel risk factor for the development and progression of diabetic nephropathy in patients with type 2 diabetes. Diabetes Care. 2013;36(7):1908-12.

164. Yokota K, Fukuda M, Matsui Y, Hoshide S, Shimada K, Kario K. Impact of visitto-visit variability of blood pressure on deterioration of renal function in patients with non-diabetic chronic kidney disease. Hypertens Res. 2013;36(2):151-7. 165. Fagard RH, Staessen JA, Thijs L. Optimal definition of daytime and night-time blood pressure. Blood Press Monit. 1997;2(6):315-21.

166. Gillett MJ. International Expert Committee report on the role of the A1c assay in the diagnosis of diabetes: Diabetes Care 2009; 32(7): 1327-1334. Clin Biochem Rev. 2009;30(4):197-200.

167. Bland JM, Altman DG. Multiple significance tests: the Bonferroni method. BMJ. 1995;310(6973):170.

168. Hansen TW, Li Y, Staessen JA. Blood pressure variability remains an elusive predictor of cardiovascular outcome. Am J Hypertens. 2009;22(1):3-4.

169. de la Sierra A, Redon J, Banegas JR, Segura J, Parati G, Gorostidi M, et al. Prevalence and factors associated with circadian blood pressure patterns in hypertensive patients. Hypertension. 2009;53(3):466-72.

170. Kario K, Pickering TG, Matsuo T, Hoshide S, Schwartz JE, Shimada K. Stroke prognosis and abnormal nocturnal blood pressure falls in older hypertensives. Hypertension. 2001;38(4):852-7.

171. Boggia J, Li Y, Thijs L, Hansen TW, Kikuya M, Bjorklund-Bodegard K, et al. Prognostic accuracy of day versus night ambulatory blood pressure: a cohort study. Lancet. 2007;370(9594):1219-29.

172. Ohkubo T, Hozawa A, Yamaguchi J, Kikuya M, Ohmori K, Michimata M, et al. Prognostic significance of the nocturnal decline in blood pressure in individuals with and without high 24-h blood pressure: the Ohasama study. J Hypertens. 2002;20(11):2183-9.

173. Hata Y, Kimura Y, Muratani H, Fukiyama K, Kawano Y, Ashida T, et al. Office blood pressure variability as a predictor of brain infarction in elderly hypertensive patients. Hypertens Res. 2000;23(6):553-60.

174. Okada H, Fukui M, Tanaka M, Inada S, Mineoka Y, Nakanishi N, et al. Visit-tovisit variability in systolic blood pressure is correlated with diabetic nephropathy and atherosclerosis in patients with type 2 diabetes. Atherosclerosis. 2012;220(1):155-9.

175. Rossignol P, Kessler M, Zannad F. Visit-to-visit blood pressure variability and risk for progression of cardiovascular and renal diseases. Curr Opin Nephrol Hypertens. 2013;22(1):59-64.

176. Salgado-Delgado R, Tapia Osorio A, Saderi N, Escobar C. Disruption of circadian rhythms: a crucial factor in the etiology of depression. Depress Res Treat. 2011;2011:839743.

177. Sihota R, Saxena R, Gogoi M, Sood A, Gulati V, Pandey RM. A comparison of the circadian rhythm of intraocular pressure in primary phronic angle closure

glaucoma, primary open angle glaucoma and normal eyes. Indian J Ophthalmol. 2005;53(4):243-7.

178. Zee PC, Attarian H, Videnovic A. Circadian rhythm abnormalities. Continuum (Minneap Minn). 2013;19(1 Sleep Disorders):132-47.

179. Chen L, Yang G. Recent advances in circadian rhythms in cardiovascular system. Front Pharmacol. 2015;6:71.

180. Madden JM, O'Flynn AM, Dolan E, Fitzgerald AP, Kearney PM. Short-term blood pressure variability over 24 h and target organ damage in middle-aged men and women. J Hum Hypertens. 2015

181. O'Brien E, Parati G, Stergiou G. Ambulatory blood pressure measurement: what is the international consensus? Hypertension. 2013;62(6):988-94.

182. Parati G, Bilo G, Redon J. Morning and smooth 24-h ambulatory blood pressure control is not achieved in general practice: results from the SURGE observational study. J Hypertens. 2013;31(3):616-23; discussion 23.

183. Pinheiro JCD, D.M. Mixed-effects models in S and S-PLUS. New York, NY: Springer. 2000

184. Nakagawa S, Schielzeth H. A general and simple method for obtaining R2 from generalized linear mixed-effects models.

185. Johnson PC. Extension of Nakagawa & Schielzeth's to random slopes models. Methods Ecol Evol. 2014;5(9):944-6.

186. Bergstrand M, Hooker AC, Wallin JE, Karlsson MO. Prediction-Corrected Visual Predictive Checks for Diagnosing Nonlinear Mixed-Effects Models. Aaps j. 2011;13(2):143-51.

187. Team; RC. R: A language and environment for statistical computing. . R Foundation for Statistical Computing, Vienna, Austria. 2014;URL <u>http://www.R-project.org/</u>.

188. Pinheiro J BD, DebRoy S, Sarkar D and R Core Team. nlme: Linear and Nonlinear Mixed Effects Models. R package version 31-119. 2015

189. Refinetti R, Lissen GC, Halberg F. Procedures for numerical analysis of circadian rhythms. Biol Rhythm Res. 2007;38(4):275-325.

190. Hardy R. Commentary: Are piecewise mixed effects models useful in epidemiology? Int J Epidemiol. 2001;30(6):1341-2.

191. Taylor KS, Heneghan CJ, Stevens RJ, Adams EC, Nunan D, Ward A. Heterogeneity of prognostic studies of 24-hour blood pressure variability: systematic review and meta-analysis. PLoS One. 2015;10(5):e0126375.

192. Liu JH, Zhang X, Kripke DF, Weinreb RN. Twenty-four-hour intraocular pressure pattern associated with early glaucomatous changes. Invest Ophthalmol Vis Sci. 2003;44(4):1586-90.

193. Sheppard JP, Hodgkinson J, Riley R, Martin U, Bayliss S, McManus RJ. Prognostic significance of the morning blood pressure surge in clinical practice: a systematic review. Am J Hypertens. 2015;28(1):30-41.

194. Head GA, Lukoshkova EV, Mayorov DN, van den Buuse M. Non-symmetrical double-logistic analysis of 24-h blood pressure recordings in normotensive and hypertensive rats. J Hypertens. 2004;22(11):2075-85.

195. Head GA, Reid CM, Shiel LM, Jennings GL, Lukoshkova EV. Rate of morning increase in blood pressure is elevated in hypertensives. Am J Hypertens. 2006;19(10):1010-7.

196. Cornelissen G. Cosinor-based rhythmometry. Theor Biol Med Model. 2014;11:16.

197. L H, F, Scheipl, J, Goldsmith, J, Gellar, J, Harezlak, MW, McLean, B Swihart, L Xiao, C, Crainiceanu and P, Reiss. refund: Regression with Functional Data. R package version 0.1-14. <u>http://CRAN.R-project.org/package=refund</u>. 2016

198. W C, J, Cheng, JJ Allaire, Y, Xie and J, McPherson. shiny: Web Application Framework for R. R package version 0.13.2. <u>http://CRAN.R-project.org/package=shiny</u>. 2016

199. Harrell F. Regression Modeling Strategies With Applications to Linear Models, Logistic and Ordinal Regression, and Survival Analysis: Springer; 2015.

200. Greven S, Crainiceanu C, Caffo B, Reich D. Longitudinal functional principal component analysis. Electron J Stat. 2010;4:1022-54.

201. Head GA, Chatzivlastou K, Lukoshkova EV, Jennings GL, Reid CM. A novel measure of the power of the morning blood pressure surge from ambulatory blood pressure recordings. Am J Hypertens. 2010;23(10):1074-81.

202. Gamst A, Wolfson T, Parry B. Local polynomial regression modeling of human plasma melatonin levels. J Biol Rhythms. 2004;19(2):164-74.

203. Newell J, McMillan K, Grant S, McCabe G. Using functional data analysis to summarise and interpret lactate curves. Comput Biol Med. 2006;36(3):262-75.

204. de la Sierra A, Vinyoles E, Banegas JR, Parati G, de la Cruz JJ, Gorostidi M, et al. Short-Term and Long-Term Reproducibility of Hypertension Phenotypes Obtained by Office and Ambulatory Blood Pressure Measurements. J Clin Hypertens (Greenwich). 2016

205. Keren S, Leibowitz A, Grossman E, Sharabi Y. Limited reproducibility of 24-h ambulatory blood pressure monitoring. Clin Exp Hypertens. 2015;37(7):599-603.

206. Monte M, Cambao M, Mesquita Bastos J, Polonia J. Reproducibility of ambulatory blood pressure values and circadian blood pressure patterns in untreated subjects in a 1-11 month interval. Rev Port Cardiol. 2015;34(11):643-50.

207. Parati G, Staessen JA. Day-night blood pressure variations: mechanisms, reproducibility and clinical relevance. J Hypertens. 2007;25(12):2377-80.

208. Fan HQ, Li Y, Thijs L, Hansen TW, Boggia J, Kikuya M, et al. Prognostic value of isolated nocturnal hypertension on ambulatory measurement in 8711 individuals from 10 populations. J Hypertens. 2010;28(10):2036-45.

209. Manning G, Rushton L, Donnelly R, Millar-Craig MW. Variability of diurnal changes in ambulatory blood pressure and nocturnal dipping status in untreated hypertensive and normotensive subjects. Am J Hypertens. 2000;13(9):1035-8.

210. White WB, Larocca GM. Improving the utility of the nocturnal hypertension definition by using absolute sleep blood pressure rather than the "dipping" proportion. Am J Cardiol. 2003;92(12):1439-41.

211. Li Y, Staessen JA, Lu L, Li LH, Wang GL, Wang JG. Is isolated nocturnal hypertension a novel clinical entity? Findings from a Chinese population study. Hypertension. 2007;50(2):333-9.

212. Abdalla M, Goldsmith J, Muntner P, Diaz KM, Reynolds K, Schwartz JE, et al. Is Isolated Nocturnal Hypertension A Reproducible Phenotype? Am J Hypertens. 2016;29(1):33-8.

213. McDonald C, Pearce MS, Wincenciak J, Kerr SR, Newton JL. Ambulatory Blood Pressure Variability Increases Over a 10-Year Follow-Up in Community-Dwelling Older People. Am J Hypertens. 2016;29(5):560-7.

214. Goldstein IB, Shapiro D, Guthrie D. A 5-year follow-up of ambulatory blood pressure in healthy older adults. Am J Hypertens. 2003;16(8):640-5.

215. MacMahon S, Peto R, Cutler J, Collins R, Sorlie P, Neaton J, et al. Blood pressure, stroke, and coronary heart disease. Part 1, Prolonged differences in blood pressure: prospective observational studies corrected for the regression dilution bias. Lancet. 1990;335(8692):765-74.

216. Evans JG, Rose G. Hypertension. Br Med Bull. 1971;27(1):37-42.

217. Lawes CM, Vander Hoorn S, Law MR, Elliott P, MacMahon S, Rodgers A. Blood pressure and the global burden of disease 2000. Part II: estimates of attributable burden. J Hypertens. 2006;24(3):423-30.

218. Chobanian AV, Bakris GL, Black HR, Cushman WC, Green LA, Izzo JL, Jr., et al. Seventh report of the Joint National Committee on Prevention, Detection, Evaluation, and Treatment of High Blood Pressure. Hypertension. 2003;42(6):1206-52.

219. Erdine S, Ari O, Zanchetti A, Cifkova R, Fagard R, Kjeldsen S, et al. ESH-ESC guidelines for the management of hypertension. Herz. 2006;31(4):331-8.

220. Kikuya M, Hansen TW, Thijs L, Bjorklund-Bodegard K, Kuznetsova T, Ohkubo T, et al. Diagnostic thresholds for ambulatory blood pressure monitoring based on 10-year cardiovascular risk. Circulation. 2007;115(16):2145-52.

221. Ohkubo T, Imai Y, Tsuji I, Nagai K, Ito S, Satoh H, et al. Reference values for 24-hour ambulatory blood pressure monitoring based on a prognostic criterion: the Ohasama Study. Hypertension. 1998;32(2):255-9.

222. Wood S, Greenfield SM, Sayeed Haque M, Martin U, Gill PS, Mant J, et al. Influence of ethnicity on acceptability of method of blood pressure monitoring: a cross-sectional study in primary care. Br J Gen Pract. 2016;66(649):e577-86.

223. Carneiro I HN, Bailey L, Vardulaki K, Langham J, Chandramohan D. Introduction to Epidemiology, Second Edition. UK: McGraw-Hill Education, 2011.

224. Gordis L. Epidemiology. Saunders, Elsevier 2009.

225. Hill AB. THE ENVIRONMENT AND DISEASE: ASSOCIATION OR CAUSATION? Proc R Soc Med. 1965;58:295-300.

226. Shan ZZ, Dai SM, Su DF. Arterial baroreflex deficit induced organ damage in sinoaortic denervated rats. J Cardiovasc Pharmacol. 2001;38(3):427-37.

227. Levi-Marpillat N, Macquin-Mavier I, Tropeano AI, Parati G, Maison P. Antihypertensive drug classes have different effects on short-term blood pressure variability in essential hypertension. Hypertens Res. 2014;37(6):585-90.

228. European-Society-of-Hypertension-and-Cardiovascular-Protection-Meeting. <u>http://www.eshonline.org/annual-meeting-posts/harmony-similar-results-with-morning-and-evening-dosing-of-antihypertensive-therapy/</u> 2016 [29/07/2016].

229. Rorie DA, Rogers A, Mackenzie IS, Ford I, Webb DJ, Willams B, et al. Methods of a large prospective, randomised, open-label, blinded end-point study comparing morning versus evening dosing in hypertensive patients: the Treatment In Morning versus Evening (TIME) study. BMJ Open. 2016;6(2):e010313.

Appendix A Supplementary material (Chapter 2)

Polynomial Regression

Briefly, a 6th order polynomial was applied to the dataset and measures of variability were obtained from subject-specific predictions. As one measure of variability the sum squared of the differences between the observed and subject-specific profiles was calculated which reflected a participant's individual BP variation across the 24h period. This is a similar measure to that calculated by Sega *et al.* except we used the variation about the individual curve rather than the mean [29]. Maximum, minimum and number of minutes spent above certain hypertensive guidelines were also calculated as measures of variability. The association between the extracted BPV measures and the presence of TOD was assessed using logistic regression with adjustment for age, sex, smoking status, BMI, diabetes and antihypertensive treatment. Additional models adjusted for mean clinic BP.



Figure 9-1 Population Level Effects for polynomial regression (Chapter 2)



Figure 9-2 Population Level Group Average (95 % CI) Linear Mixed Effects Model (Age, Sex, BMI Adjusted) (Chapter 2)



Figure 9-3 Subject-Specific Variation (function of Time only)

| | | Microalbuminuria (10.6%) | |
|-------------------------|--------------|--------------------------|--------------------------|
| Characteristic (SBP) | Value (SD) | Model 1 | Model 2 |
| | | OR (CI%) per hour or per | OR (CI%) per hour or per |
| | | 10mmHg | 10mmHg |
| Max Morning, mm Hg | 133 (13.0) | 1.53 (1.30-1.81)* | 0.24 (0.03-1.83) |
| Time | 11.54pm | 0.85 (0.59-1.24) | 0.80 (0.55-1.15) |
| | (18min) | | |
| Minimum, mm Hg | 108 (12.7) | 1.60 (1.35-1.90)* | 2.74 (0.90-8.33) |
| Time | 4.51am | 1.08 (0.59-0.97) | 0.82 (0.45-1.49) |
| | (11min) | | |
| Hours ≥ 130 mm Hg (24 | 8.4 h (7.9h) | 1.07 (1.04-1.10)* | 0.98 (0.91-1.04) |
| h)† | | | |
| Hours ≥ 135 mm Hg | 5.2 h (6.4h) | 1.06 (1.02-1.10)* | 0.95 (0.89-1.01) |
| (Day)† | | | |
| Hours \geq 120 mm Hg | 2.5 h (3.1h) | 1.14 (1.07-1.21)* | 0.96 (0.85-1.09) |
| (NIght)T | | | |
| Variability About Curve | 123 (58.1) | 1.05 (1.02-1.08)* | 1.01 (0.97-1.05) |

Table 9-1 Association between parameters extracted from polynomial regression model and microalbuminuria (logistic regression).

 \pm European Society of Hypertension ABPM Guidelines: 24 h ≥130/80 mmHg; Day (awake) ≥135/85 mmHg; Night (sleep) ≥120/70 mmHg. *p-value <0.01. Model 1 adjusted for age, sex, smoking, BMI, diabetes, anti-hypertensive medication. Model 2 adjusted for age, sex, smoking, BMI, diabetes, anti-hypertensive medication, 24h SBP

Appendix B Supplementary material (Chapter 3) - search terms and search strategy for systematic review

EMBASE search strategy

- #1. exp abpm variability/
- #2. exp abp variability/
- #3. exp ambulatory bp variability/
- #4. exp ambulatory blood pressure variability/
- #5. exp ambulatory blood pressure monitor variability/
- #6. exp ambulatory blood pressure monitoring variability/
- #7. exp 24hour BP variability/
- #8. exp 24-hour blood pressure variability/
- #9. exp 24h blood pressure variability/
- #10. exp 24h bp variability/
- #11. exp 24 hour ambulatory blood pressure variability/
- #12. exp 24 hour ambulatory bp variability/
- #13. exp 24 hour ambulatory blood pressure monitor variability/
- #14. exp 24 hour ambulatory blood pressure monitoring variability/
- #15. exp short term bp variability/
- #16. exp short term abpm variability/
- #17. exp short term blood pressure variability/
- #18. exp short term abp variability/
- #19. exp 24 hour abpm variability/
- #20. exp 24 hour abp variability/
- #21. exp 24h abp variability/
- #22. exp 24-hour abp variability/
- #23. exp 24-hour abpm variability/
- #24. exp 24-hour ambulatory blood pressure monitoring variability/
- #25. exp 24-hour ambulatory blood monitoring variability/
#26. exp bp variability/

#27. exp blood pressure variability/

#28. (#1 or #2 or #3 or #4 or #5 or #6 or #7 or #8 or #9 or #10 or #11 or #12 or #13 or #14 or #15 or #16 or #17 or #18 or #19 or #20 or #21 or #22 or #23

or #24 or #25 or #26 or #27)

#29. left ventricular hypertrophy: ab,ti

- #30. left ventricular hyperthrophy: ab,ti
- #31. left ventricular hypertrophy: ab,ti
- #32. left ventricular hypertrophic: ab,ti
- #33. left ventricular mass: ab,ti
- #34. left ventricular mass index: ab,ti
- #35. target organ damage: ab,ti
- #36. organ damage: ab,ti
- #37. target-organ damage: ab,ti
- #38. (#29 or #30 or #31 or #32 or #33 or #34 or #35 or #36 or #37)
- #39. (#28 and #38)
- #40.#39[Limit: humans & article]

PubMed search strategy

- #1. abpm variability [all]
- #2. abp variability [all]
- #3. ambulatory bp variability [all]
- #4. ambulatory blood pressure variability [all]
- #5. ambulatory blood pressure monitor variability [all]
- #6. ambulatory blood pressure monitoring variability [all]
- #7. 24hour BP variability [all]
- #8. 24-hour blood pressure variability [all]
- #9. 24h blood pressure variability [all]
- #10. 24h bp variability [all]
- #11. 24 hour ambulatory blood pressure variability [all]
- #12. 24 hour ambulatory bp variability [all]
- #13. 24 hour ambulatory blood pressure monitor variability [all]

- #14. 24 hour ambulatory blood pressure monitoring variability [all]
- #15. short term bp variability [all]
- #16. short term abpm variability [all]
- #17. short term blood pressure variability [all]
- #18. short term abp variability [all]
- #19. 24 hour abpm variability [all]
- #20. 24 hour abp variability [all]
- #21. 24h abp variability [all]
- #22. 24-hour abp variability [all]
- #23. 24-hour abpm variability [all]
- #24. 24-hour ambulatory blood pressure monitoring variability [all]
- #25. 24-hour ambulatory blood monitoring variability [all]
- #26. bp variability [all]
- #27. blood pressure variability [all]
- #28. (#1 or #2 or #3 or #4 or #5 or #6 or #7 or #8 or #9 or #10 or #11 or #12 or #13 or #14 or #15 or #16 or #17 or #18 or #19 or #20 or #21 or #22 or #23
- or #24 or #25 or #26 or #27)
- #29. left ventricular hypertrophy [all]
- #30. left ventricular hyperthrophy [all]
- #31. left ventricular hypertrophy [all]
- #32. left ventricular hypertrophic [all]
- #33. left ventricular mass [all]
- #34. left ventricular mass index [all]
- #35. target organ damage [all]
- #36. organ damage [all]
- #37. target-organ damage [all]
- #38. (#29 or #30 or #31 or #32 or #33 or #34 or #35 or #36 or #37)
- #39. (#28 and #38)
- #40. #39[Limit: humans & adults]

PRISMA checklist (Systematic review checklist for Chapter 3)

| Section/topic | # | Checklist item | Reported on page # (see pdf version appendix) | | | |
|---------------------------|---|---|---|--|--|--|
| TITLE | | | | | | |
| Title | 1 | Identify the report as a systematic review, meta-analysis, or both. | Title page (p1) | | | |
| ABSTRACT | | | | | | |
| Structured summary | 2 | Provide a structured summary including, as applicable: background; objectives; data sources; study eligibility criteria, participants, and interventions; study appraisal and synthesis methods; results; limitations; conclusions and implications of key findings; systematic review registration number. | Abstract (p2) | | | |
| INTRODUCTION | | | | | | |
| Rationale | 3 | Describe the rationale for the review in the context of what is already known. | Introduction (p3,4) | | | |
| Objectives | 4 | Provide an explicit statement of questions being addressed with reference to participants, interventions, comparisons, outcomes, and study design (PICOS). | Introduction (p4) | | | |
| METHODS | | | | | | |
| Protocol and registration | 5 | Indicate if a review protocol exists, if and where it can be accessed (e.g., Web address), and, if available, provide registration information including registration number. | NA | | | |
| Eligibility criteria | 6 | Specify study characteristics (e.g., PICOS, length of follow-up) and report characteristics (e.g., years considered, language, publication status) used as criteria for eligibility, giving rationale. | Methods – Types of studies, Study populations, predictor variables, outcomes (p5) | | | |
| Information sources | 7 | Describe all information sources (e.g., databases with dates of coverage, contact with study authors to identify additional studies) in the search and date last searched. | Methods – search methods for identification of studies (p5,6) | | | |
| Search | 8 | Present full electronic search strategy for at least one database, including any limits used, such that it could be repeated. | Methods – search methods for identification of studies (p5,6) | | | |

| Study selection | 9 | State the process for selecting studies (i.e., screening, eligibility, included in systematic review, and, if applicable, included in the meta-analysis). | Methods – search methods for identification of studies, data extraction (p5,6) |
|------------------------------------|----|--|--|
| Data collection process | 10 | Describe method of data extraction from reports (e.g., piloted forms, independently, in duplicate) and any processes for obtaining and confirming data from investigators. | Methods –data extraction (p6) |
| Data items | 11 | List and define all variables for which data were sought (e.g., PICOS, funding sources) and any assumptions and simplifications made. | Methods –data extraction (p6) |
| Risk of bias in individual studies | 12 | Describe methods used for assessing risk of bias of individual studies (including specification of whether this was done at the study or outcome level), and how this information is to be used in any data synthesis. | Methods— search methods for identification of studies (p6), Quality assessment (Table 4) |
| Summary measures | 13 | State the principal summary measures (e.g., risk ratio, difference in means). | Methods—Statistical analysis (p6) Correlation Coefficient |
| Synthesis of results | 14 | Describe the methods of handling data and combining results of studies, if done, including measures of consistency (e.g., I^2) for each meta-analysis. | Methods—Statistical analysis (p6) |

Page 1 of 2

| Section/topic | # | Checklist item | Reported on page # |
|-----------------------------|----|---|---|
| Risk of bias across studies | 15 | Specify any assessment of risk of bias that may affect the cumulative evidence (e.g., publication bias, selective reporting within studies). | Methods—Statistical analysis (p6) |
| Additional analyses | 16 | Describe methods of additional analyses (e.g., sensitivity or subgroup analyses, meta- regression), if done, indicating which were pre-specified. | Methods—Statistical analysis (p6) |
| RESULTS | | | |
| Study selection | 17 | Give numbers of studies screened, assessed for eligibility, and included in the review, with reasons for exclusions at each stage, ideally with a flow diagram. | Results (p7) & Flowchart (Figure 1) |
| Study characteristics | 18 | For each study, present characteristics for which data were extracted (e.g., study size, PICOS, follow-up period) and provide the citations. | Results (p7) & Table 1,3 |
| Risk of bias within studies | 19 | Present data on risk of bias of each study and, if available, any outcome level assessment (see item 12). | Results (p8) & Quality assessment (Table 4) |

| Results of individual studies | 20 | For all outcomes considered (benefits or harms), present, for each study: (a) simple summary data for each intervention group (b) effect estimates and confidence intervals, ideally with a forest plot. | Results (p7), Table 1,3 & Forest Plot (Figure 2) |
|-------------------------------|----|--|--|
| Synthesis of results | 21 | Present results of each meta-analysis done, including confidence intervals and measures of consistency. | Results (p7) & Forest Plot (Figure 2) |
| Risk of bias across studies | 22 | Present results of any assessment of risk of bias across studies (see Item 15). | Results- Meta-analysis (p8,9) |
| Additional analysis | 23 | Give results of additional analyses, if done (e.g., sensitivity or subgroup analyses, meta- regression [see Item 16]). | Results- Meta-analysis (p8,9) & Forest Plot (Figure 2) |
| DISCUSSION | | | |
| Summary of evidence | 24 | Summarize the main findings including the strength of evidence for each main outcome; consider their relevance to key groups (e.g., healthcare providers, users, and policy makers). | Discussion (p9) |
| Limitations | 25 | Discuss limitations at study and outcome level (e.g., risk of bias), and at review-level (e.g., incomplete retrieval of identified research, reporting bias). | Discussion (p9,10, 11, 12) |
| Conclusions | 26 | Provide a general interpretation of the results in the context of other evidence, and implications for future research. | Discussion (p9,10, 11, 12) |
| FUNDING | | | |
| Funding | 27 | Describe sources of funding for the systematic review and other support (e.g., supply of data); role of funders for the systematic review. | Title page (p1) & Conflict of Interest (p12) |

From: Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009). Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6(6): e1000097. doi:10.1371/journal.pmed1000097

For more information, visit: www.prisma-statement.org.

Page 2 of 2

Appendix C Supplementary material (Chapter 4)

| | | Untreate | d (n=757) | Treated | (n=377) |
|----------------------------|--------------|--------------|--------------|--------------|--------------|
| Characteristic | Total | Men | Female | Men | Female |
| | (n=1134) | (n=362) | (n=395) | (n=165) | (n=212) |
| Age, y | 60.2 (5.5) | 59.5 (5.6) | 59.5 (5.5) | 61.6 (5.1) | 61.9 (5.3) |
| BMI, n (%) | | | | | |
| Underweight/normal | 230 (20.3) | 56 (15.5) | 130 (32.9) | 8 (4.9) | 36 (17.0) |
| Overweight | 498 (44.0) | 187 (51.8) | 171 (43.3) | 65 (39.4) | 75 (35.4) |
| Obese | 405 (35.7) | 118 (32.7) | 94 (23.8) | 92 (55.7) | 101 (47.6) |
| Smoking, n (%) | | | | | |
| Never | 564 (51.9) | 133 (38.8) | 241 (63.3) | 70 (43.2) | 120 (59.7) |
| Former | 355 (32.7) | 142 (41.4) | 86 (22.6) | 71 (43.8) | 56 (27.9) |
| Current | 168 (15.4) | 68 (19.8) | 54 (14.2) | 21 (13.0) | 25 (12.4) |
| Diabetes, n (%) | 103 (9.3) | 34 (9.6) | 14 (3.6) | 31 (19.1) | 24 (11.4) |
| Hypertension, n (%) | 487 (43.0) | 140 (38.7) | 148 (37.6) | 92 (55.8) | 107 (50.5) |
| LVH, n (%) | 75 (6.6) | 34 (4.5) | 15 (3.8) | 21 (12.7) | 20 (9.4) |
| Microalbuminuria, n (%) | 128 (11.4) | 73 (9.7) | 32 (8.2) | 24 (14.6) | 31 (14.8) |
| ABPM measurements | | | | | |
| 24-h SBP, mm Hg | 124.1 (13.3) | 126.0 (12.6) | 119.9 (13.7) | 128.0 (12.4) | 125.8 (12.7) |
| 24-h DBP, mm Hg | 71.8 (8.3) | 74.4 (8.4) | 69.6 (7.6) | 74.3 (8.6) | 69.5 (7.4) |
| Awake SBP, mm Hg | 131.4 (14.1) | 133.5 (13.4) | 127.2 (14.8) | 134.8 (12.6) | 132.9 (13.3) |
| Awake DBP, mm Hg | 77.4 (9.0) | 80.1 (9.0) | 75.3 (8.5) | 79.3 (9.3) | 74.9 (8.3) |
| Sleep SBP, mm Hg | 112.3 (14.0) | 112.9 (13.4) | 108.5 (13.5) | 117.1 (14.3) | 115.0 (13.9) |
| Sleep DBP, mm Hg | 62.8 (8.3) | 64.6 (8.7) | 60.6 (13.5) | 66.0 (9.2) | 61.3 (7.2) |

Table 9-2 Baseline Characteristics by anti-hypertensive treatment

Data are mean (SD). ARV: Average real variability, SBP: Systolic Blood pressure, DBP: Diastolic blood pressure, BMI:Body mass index, LVH: Left ventricular hypertrophy, ABPM: Ambulatory blood pressure monitor. Microalbuminuria: albumin:creatinine ratio \geq 1.1 mg/mmol

| Variability | 24-hour (mm Hg) | | Day (r | nm Hg) | Night (mm Hg) | | |
|-------------|-----------------|--------------|------------|--------------|---------------|-------------|--|
| Measure | No (n=986) | Yes (n=147) | No | Yes | No | Yes | |
| ARV SBP | 11.4 (2.6) | 11.1 (2.0) | 11.9 (3.2) | 11.5 (2.7) | 10.6 (3.3) | 10.6 (2.6) | |
| ARV DBP | 7.8 (1.8) | 7.3 (1.4) ** | 8.2 (2.4) | 7.6 (2.2) ** | 7.3 (2.2) | 6.9 (1.8) * | |
| SD SBP | 15.9 (4.0) | 15.4 (3.0) | 12.9 (3.6) | 13.0 (3.0) | 11.2 (3.6) | 10.7 (2.7) | |
| SD DBP | 11.5 (2.8) | 10.9 (2.4) * | 9.0 (2.8) | 8.6 (2.5) | 8.1 (2.4) | 7.6 (2.0) * | |
| CV SBP | 12.8 (2.9) | 12.8 (2.8) | 9.7 (2.4) | 10.1 (2.4) | 10.0 (3.0) | 9.8 (2.5) | |
| CV DBP | 16.1 (3.8) | 15.6 (3.4) | 11.6 (3.7) | 11.5 (3.5) | 12.9 (3.8) | 12.5 (3.5) | |
| wSD SBP | 12.3 (3.0) | 12.2 (2.2) | - | - | - | - | |
| wSD DBP | 8.7 (2.1) | 8.2 (1.7) | - | - | - | - | |
| Dip SBP | 14.4 (7.0) | 14.2 (6.8) * | - | - | - | - | |
| Dip DBP | 18.7 (7.7) | 18.2 (7.2) | - | - | - | - | |

Table 9-3 Variability Indices and Dip by White-Coat Hypertension

Data are mean (SD). ARV: Average real variability, SD: Standard deviation, wSD: Weighted standard deviation, CV: coefficient of variation, SBP: Systolic Blood pressure, DBP: Diastolic blood pressure, *p<0.05; represents significance between those with and without white-coat hypertension during each period of the day

ROC curves for microalbuminuria (Chapter 4 supplementary material)



ARV awake ROC curve. M1: Age, Sex, 24-h mean BP, ARV awake, Smoking, Diabetic, bmi, anti-hypertensive; M2: Age, Sex, 24-h mean BP, Smoking, Diabetic, bmi, anti-hypertensive (p-value=0.56)



wSD ROC curve. M1: Age, Sex, 24-h mean BP, wSD, Smoking, Diabetic, bmi, antihypertensive; M2: Age, Sex, 24-h mean BP, Smoking, Diabetic, bmi, antihypertensive (p-value=0.73)



SD awake ROC curve. M1: Age, Sex, 24-h mean BP, SD awake, Smoking, Diabetic, bmi, anti-hypertensive; M2: Age, Sex, 24-h mean BP, Smoking, Diabetic, bmi, anti-hypertensive (p-value=0.25)



24-h mean BP ROC curve. M1: 24-h mean BP, clinic BP; M2: clinic BP (p-value=0.01)



24h mean BP ROC curve. M1: Age, Sex, 24-h mean BP, clinic BP, Smoking, Diabetic, bmi, anti-hypertensive; M2: Age, Sex, clinic BP, Smoking, Diabetic, bmi, anti-hypertensive (p-value=0.13)



Appendix D Supplementary material (Chapter 5)

Figure 9-4 VPC (visual predictive check) plot. Black lines represent median of observed data with 90% interquantile range of observations. Red line is predicted mean along with 90% prediction interval.



Figure 9-5 Predicted average (95% CI) piecewise linear trajectory of those with/without presence of microalbuminuria adjusted for age, sex, BMI and including an interaction of the spline terms with microalbuminuria using a linear mixed-effects model (Model 3). Each linear spline represents the rate of BP increase or decrease (slope) for that time period which is referred to in Table 5-2.

Appendix E Supplementary material (Missing Data)

(References for this appendix are at the end of the section.)

Although some studies have monitors set to collect data less frequently at night compared to daytime, sometimes every 30mins during the day and every hour at night, our study had the advantage that data was collected at the same sampling frequency, 30mins throughout 24h. Earlier ESH guidelines recommended a minimum of 14 measurements during the day and a minimum of 7 measurements during the night as the satisfactory number of data points for analysis [1-3]. However the more recent guidelines from the current position paper on ABPM say that stricter criteria should be used especially for research purposes [3, 4]. Specifically they recommend only participants with a minimum of 20 measurements during the day and a minimum of 7 measurements during the night period should be retained. Additionally, any participants with data lacking for more than two consecutive hourly intervals should be excluded [3, 4]. Additionally, when examining day and night values they recommended fixed time periods to identify these periods but we had the advantage of having diary entries of actual bed and rising times.

Unfortunately we had no available data that was collected on the reasons why there may be missing data in our study. However, common reasons for missing ABPM values include the patient disconnecting the device, suspension of a reading by use of the cancelation button, turning the monitor off, dead batteries, movement artefact, or kinks in the tubing [5]. Many of these reasons may lead to an occasional or sporadic missed reading but multiple missing data may suggest there is something else at fault such as a monitor malfunctioning and ignoring data from such a monitor seems appropriate. Another issue perhaps is that a certain group of individuals may have increased level of missing data. One study examining the influence of patient characteristics on the success of ABPM recording found that those with diabetes and elevated BMI were associated with less complete ABPM session results [6]. In relation to our study we explored differences between

229

those with and without the minimum measurements criteria, see Table 9-4. Similar to the Fravel et al. study we found that those with higher BMI were more likely to have missing data. However there was no significant difference in the prevalence of diabetes. BP was slightly lower in those with missing data but importantly there was no significant difference between groups based on hypertension classification (≥140/90 mmHg and/or on antihypertensive treatment). There was no difference in gender but the missing group were marginally older (59.9 vs 60.7 years, p=0.01). The difference in BMI may be explained by the practicality of wearing the monitor. Perhaps those with a higher BMI were given a standard size cuff resulting in the device either being too restricted and returning an error or it was too painful to wear and the participant had to remove it. Without any evidence however we are only speculating and the difference between groups was not that large (1.3 kg/m^2) . Importantly from Table 9-4, the ABPM subsample was broadly representative of the full sample as prevalence rates were similar. However, those on ABPM had a higher prevalence of hypertension. ABPM was offered to all patients but those with high BP were encouraged to wear a monitor. This may affect the generalizability of the results but will not affect the estimates from our models for this study sample. It must be noted that even in the ABPM sample that met the criteria, there was still some missing data points. In terms of mixed-effects models, the approach is only valid when the incomplete dataset is MAR or MCAR [7-8]. In any real-life dataset missing data will never be completely MCAR, MAR or missing not at random (MNAR) but rather a combination of each. An assumption has to be made and in the case of our work we believe MAR is a reasonable assumption. As we have adhered to the most recent ABPM guidelines and feel our remaining ABPM sample is representative of the full sample it is reasonable to analysis those with the appropriate number of readings as set out by the guidelines. In addition there have been studies illustrating that methods that deal with missing data such as multiple imputation in mixed-effects models are not necessary and can result in unstable estimates and misleading inferences [7,9-10].

| | ABPM Subsample (n=1207 | | | | | |
|--------------------------|------------------------|--------------|---------------|--|--|--|
| Characteristic | Full Sample | ABPM ‡ | ABPM Missing | | | |
| | (n=2047) | (n=886) | (n=321) | | | |
| Age, y | 59.8 (5.5) | 59.9 (5.5) | 60.7 (5.6)* | | | |
| Gender, Male n(%) | 1008 (49.2) | 401 (45.3) | 163 (50.8) | | | |
| BMI (kg/m ²) | 28.6 (4.7) | 28.6 (4.6) | 29.9 (5.3)* | | | |
| Office SBP, mm Hg | 129.6 (16.9) | 134.7 (17.7) | 131.7 (17.1)* | | | |
| Office DBP, mm Hg | 80.1 (9.8) | 83.1 (10.2) | 80.7 (10.2)* | | | |
| Hypertension, n (%) | 951 (46.5) | 528 (59.7) | 182 (56.7) | | | |
| Diabetes n (%) | 174 (8.7) | 81 (9.3) | 34 (10.9) | | | |

Table 9-4 Table of demographics comparing full sample to ABPM sample

Data are mean (SD) unless otherwise stated. BMI: Body mass index, ABPM: Ambulatory blood pressure monitor. Hypertension: ≥140/90 mmHg and/or on antihypertensive treatment. Diabetes: based on hbA1c greter than 6.5% and doctor diagnosed. ‡ABPM group based on minimum 20 valid awake and minimum 7 valid asleep and ≥2 valid daytime and 1 valid night-time measurement per h. *p-value<0.05, where p-value is comparing two ABPM groups.

References for Appendix E:

[1] O'Brien E, Coats A, Owens P, *et al.* Use and interpretation of ambulatory blood pressure monitoring: recommendations of the British Hypertension Society. BMJ 2000; 320: 1128.

[2] Fagard RH, Staessen JA, Thijs L. Optimal definition of daytime and night-time blood pressure. Blood Pressure Monitoring. 1997;2(6):315-21.

[3] O'Brien E, Parati G, Stergiou G. Ambulatory blood pressure measurements: What is the International Consensus? Hypertension. 2013;62:988–94.

[4] O'Brien E, Parati G, Stergiou G, Asmar R, Beilin L, Bilo G, *et al.* European Society of Hypertension position paper on ambulatory blood pressure monitoring. Journal of Hypertension. 2013;31(9):1731-68.

[5] Flynn JT, Daniels SR, Hayman LL, Maahs DM, McCrindle BW, Mitsnefes M, Zachariah JP, Urbina EM; American Heart Association Atherosclerosis, Hypertension and Obesity in Youth Committee of the Council on Cardiovascular Disease in the Young. Update: ambulatoryblood pressure monitoring in children and adolescents: a scientific statement from the American Heart Association. Hypertension. 2014; 63:1116–1135.

[6] Fravel MA, Ernst ME, Weber CA, Dawson JD, Carter BL, Bergus GR. Influence of patient characteristics on success of ambulatory blood pressure monitoring. Pharmacotherapy 2008; 28: 1341–47.

[7] Twisk J, de Boer M, de Vente W, Heymans M. Multiple imputation of missing values was not necessary before performing a longitudinal mixed-model analysis. J Clin Epidemiol 2013;66:1022

[8] Gibbons RD, Hedeker D, DuToit S. Advances in analysis of longitudinal data. Annual Review of Clinical Psychology. 2010;6:79-107.

[9] Little, RJA and Rubin, DB. Statistical analysis with missing data 2nd ed. Wiley, New York, NY, 2002

[10] Peters, SAE, Bots, ML, den Ruijter, HM, Palmer, MK, Grobbee, DE, Crouse, JR III, *et al.* Multiple imputation of missing repeated outcome measurements did not add to linear mixed-effects models. J Clin Epidemiol, 65 (2012), pp. 686–695



Appendix F Supplementary material (Chapter 6)

Figure 9-6 ABPM readings of three individuals with fitted subject-specific trajectories from a two-component cosinor mixed-effects model (five parameters) and spline model (six parameters - cubic spline with four knots at 18:00, 24:00, 04:00 and 08:00) (left panels). Their corresponding rate of change curves (first derivatives) are also plotted (right panels). The plots indicated an extremely similar pattern thus giving further justification of the two-component cosinor model.

Max Slope One by One



Figure 9-7 Histograms of maximum morning slope by different methods.



Figure 9-8 Scree plot with eigenvalues and the eight principle components which make up 99% of the variation.



Figure 9-9 FPCA: Each of the six three FPC as variations about the mean along with the percentage of total variation explained by the component. The solid black line represents the mean SBP over the day. The "+" curves illustrate what happens when a small amount of the component is added to the mean and the "-" curves illustrate the effect of subtracting the component.

| | | -20 0 10 | | -5 0 5 | | -60 -20 20 | | -10 0 5 15 | | -6 -2 246 | | -4 -2 0 2 | |
|-------------------------|-----------|-------------|--------------|-------------|-------------------|------------|-------------|------------|------------|------------|-------------|------------|------------|
| | MESOR | 0.2 | 0.04 | 0.4 | 0.04 | 1.0 | 0.005 | 0.002 | 0.001 | 0.02 | 0.001 | 0.006 | 0.02 |
| 20 10 -10 -20 | | Amplitude 1 | 0.02 | 0.7 | 0.003 | 0.2 | 0.9 | 0.1 | 0.07 | 0.03 | 0.03 | 0.08 | 0.07 |
| | | | Phase 1 | 0.2 | 0.6 | 0.05 | 0.2 | 0.9 | 0.1 | 0.002 | 0.07 | 0.08 | 0.04 |
| 5 0 -5 | | | | Amplitude 2 | 0.2 | 0.4 | 0.6 | 0.1 | 0.2 | 0.5 | 0.02 | 0.002 | 0.06 |
| | | | | | Phase 2 | 0.03 | 0.05 | 0.6 | 0.7 | 0.05 | 0.02 | 0.1 | 0.02 |
| 20 -20 -40 -60 | | | | | | PC Score 1 | 0.004 | 0.001 | 0.01 | 0.01 | 0.01 | 0.003 | 0.03 |
| | | | | | | | PC Score 2 | 0.01 | 0.002 | 0.009 | 0.03 | 0.03 | 0.01 |
| 10.5000 | | | · **** | | | | | PC Score 3 | 0.007 | 0.002 | 0.003 | 0.009 | 0.01 |
| | | | | | | | | | PC Score 4 | 0.02 | 0.04 | 0.04 | 0.02 |
| a hiconae | | | | | | | | | | PC Score 5 | 0.02 | 0.01 | 0.006 |
| | | | | | | | | | | | PC Score 6 | 0.006 | 0.03 |
| www.orwy | | | | | | | | | | | | PC Score 7 | 0.01 |
| | | - | | | | | | | | | | | PC Score 8 |
| | -20 20 60 | | -1.0 0.0 1.0 | | -1.5 -0.5 0.5 1.5 | | -20 0 10 20 | | -5 0 5 | | -4 -2 0 2 4 | | -2024 |

Figure 9-10 Scatter plots and the corresponding correlations between all the two-component cosinor random-effects model parameters and the principle component scores from FPCA. Eight FPCs contribute to 99% of the total variance in the data.

ARV Sampling Frequency

(References at the end of the section)

Using the simulated data based on the two-component randon-effects cosinor model in Chapter 6, ARV values were explored with respect to their sampling frequency. ARV was calculated from the simulated data (1) every 10mins and (2) every 30mins. Results indicated a substantial difference between the values taken at 10min and 30min intervals (p=0.02), see Figure 9-11 and Figure 9-12. The results are not surprising but the sampling frequency is not always referred to when ARV is mentioned in the literature, even those who developed ARV [1] failed to highlight it with only some attempting to adjust for the time interval [2,3]. We believe highlighting the findings is worthwhile. Zakopoulos *et al.* have however, suggested and advocated the use of the time rate of BP varation measure which is similar to ARV but is independent of the time intervals between measurements [4]. The measure was additionally obtained and included in Chapter 3 when summary measures were examined. We found the results were similar to the ARV value. Future studies using ARV should consider accounting for time between readings and determine the time rate of variation index. It must also be noted that although possible, in practise however, measurements taken every 10mins are unrealistic. The inconvience casued to the participant may result in them becoming iritated with the device and in turn lead to misleading values.





Figure 9-11 Simulated ARV values taken every 10mins and every 30mins.



Figure 9-12 Simulated data for one individual with readings taken 1) every 30min and 2) every 10mins.

References

[1] Mena L, Pintos S, Queipo NV, Aizpurua JA, Maestre G, Sulbaran T. A reliable index for the prognostic significance of blood pressure variability. Journal of Hypertension. 2005;23(3):505-11.

[2] Leoncini G, Viazzi F, Storace G, Deferrari G, Pontremoli R. Blood pressure variability and multiple organ damage in primary hypertension. Journal of Human Hypertension. 2013;27(11):663-70.

[3] Hansen TW, Thijs L, Li Y, Boggia J, Kikuya M, Bjorklund-Bodegard K, *et al.* Prognostic value of reading-to-reading blood pressure variability over 24 hours in 8938 subjects from 11 populations. Hypertension. 2010;55(4):1049-57.

[4] Zakopoulos, NA, Tsivgoulis, G, Barlas, G, *et al.* (2005) Time Rate of Blood Pressure Variation Is Associated With Increased Common Carotid Artery Intima-Media Thickness. Hypertension. (4):505-512

















Appendix H Sample R Code

| | — |
|-----------------------|---|
| platform | x86_64-w64-mingw32 |
| arch | x86_64 |
| 05 | mingw32 |
| system | x86_64, mingw32 |
| status | |
| major | 3 |
| minor | 3.1 |
| year | 2016 |
| month | 06 |
| day | 21 |
| svn rev | 70800 |
| language | R |
| version.string | R version 3.3.1 (2016-06-21) |
| nickname | Bug in Your Hair |
| ##################### | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, |

rm(list = ls()) #clear

library(foreign) #imports stata file library(nlme) library(multcomp)

getwd() setwd("C:/Users/JM") w1 <- read.dta("w1.dta") w1<-w1[order(w1\$id, w1\$newtime),]

```
####newtime 0-47 represents 24h clock
####newsleep represents subject-specific sleep time
####newwawke represents subject-specific wake time
```

####create individual splines with restriction BP on the average is cyclical ####
####create 5 splines - t12pm,t6pm,tsleep,t4am,twake
w1\$t12pm<-ifelse(w1\$newtime<=12, w1\$newtime, 12)
w1\$t6pm <- ifelse(w1\$newtime<=12, 0, ifelse((w1\$newtime>12) & (w1\$newtime<=w1\$newsleep),
(w1\$newtime-12), (w1\$newsleep-12)))
w1\$tsleep<-ifelse(w1\$newtime<=w1\$newsleep, 0, ifelse((w1\$newtime>w1\$newsleep) & (w1\$newtime<=32),
(w1\$newtime-w1\$newsleep), (32-(w1\$newsleep))))
w1\$t4am<- ifelse(w1\$newtime<=32 ,0, ifelse((w1\$newtime>32) & (w1\$newtime<=w1\$newwake),
(w1\$newtime-32), (w1\$newtime<=w1\$newwake, 0, ifelse((w1\$newtime>w1\$newwake) & (w1\$newtime<=47),
(w1\$newtime-w1\$newwake), (47-(w1\$newwake))))
####code introducing restriction ensuring pattern is periodic</pre>

####code introducing restriction ensuring pattern is periodic w1\$s2time<-w1\$newsleep-12 w1\$s3time<-32-w1\$newsleep</pre>
summary(glht(m1r, linfct = c("(s2*10+s3*10+s4*8+s5*7)/-12=0"))) #0.0158 se=0.037

 library(foreign) #creates stata file library(Ime4) library(nlme) library(survey) library(gmodels) library(dplyr) library(ggplot2) library(multcomp) library(mgcv) library(ImerTest) library(splines) library(ares) #only works with R2.13.2, Ispline i.e. mkspline library(car) library(Ime4) library(RLRsim) #library(MASS) #Idahist, multiple histograms library(Hmisc) library(corrplot) library(reshape) library(reshape2) library(zoo) library(broom) library(Imtest) library(doBy) #tabstat stata

getwd()

a1 <- read.dta("dataset.dta")

```
####2 component cosinor####
#cos cos2 sin sin2
fm2 <- Ime(sbp ~ cos(2*pi*newtime/48) + sin(2*pi*newtime/48) + cos(4*pi*newtime/48)
                                                                                                 +
sin(4*pi*newtime/48),
     random =
                   ~
                        cos(2*pi*newtime/48)
                                                +
                                                    sin(2*pi*newtime/48)+cos(4*pi*newtime/48)
                                                                                                 +
sin(4*pi*newtime/48) | Study_ID,method="ML",
     data=a1,control = ImeControl(msMaxIter=1000,opt = "optim",msVerbose=T),
     na.action="na.omit",correlation = corAR1(form=~1|Study_ID))
#saveRDS(fm2, "fm2.rds")
fm2 <- readRDS("fm2.rds")
summary(fm2)
anova(fm1,fm2) #same models
```

```
####plot different fits- single,2 & 3 component - comparison####
somePDFPath = "some.pdf"
pdf(file=somePDFPath)
par(mar=c(5,4,1.5,1.5)+0.1) #remove title space
par(las=1) #y axis labels horizontal
par(mfrow = c(2,2)) #change depending how many per page
\# par(mar = c(4, 4.5, 1.5, 1.5) + 0.1)
for (i in unique(a1$Study ID)) {
plot(a1[a1$Study_ID==i, "newtime"], a1[a1$Study_ID==i, "sbp"], bty="I",
   #col=a1[a1$Study_ID==i, "visitno"],
   xaxt = "n",
   #xlim = range(a$bphour2), ylim = range(a$sbp), # base the axes on full data range
   ylim=c(80,180),
   # main = paste("Plot of", i),
   xlab="Time (24-h clock)", ylab="SBP (mmHg)",font.main = 1,cex.lab=1.5,cex.axis = 1.5)
lines(a1[a1$Study_ID==i, "sbp"]~a1[a1$Study_ID==i, "newtime"], lwd=1)
lines(a1[a1$Study ID==i, "fitted sub1"]~a1[a1$Study ID==i, "newtime"], lwd=3)
lines(a1[a1$Study_ID==i, "fitted_sub2"]~a1[a1$Study_ID==i, "newtime"], lwd=3,col="red")
lines(a1[a1$Study ID==i, "fitted sub3"]~a1[a1$Study ID==i, "newtime"], lwd=3,col="blue")
axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"), cex.axis = 1.5)
}
dev.off()
####derivatives, simulation & obtain max slope####
a1<-a1 %>% group_by(Study_ID) %>% mutate(derivative2 = c(NA,diff(fitted_sub2) / diff(newtime))) %>%
ungroup() %>%
as.data.frame()
####prediction every minute####
z<-expand.grid(newtime=seq(0,47,by=1),Study ID=unique(a1$Study ID)) #prediction of lme for every minute
of dav
z<-expand.grid(newtime=seq(0,47,by=1/30),Study_ID=unique(a1$Study_ID))</pre>
                                                                            #prediction of Ime for every
minute of day
z$pred <- predict(fm2, newdata=z,level=1)
z$pred <- predict(cos1qq, newdata=z,level=1)
z<-z %>% group_by(Study_ID) %>% mutate(derivative = c(NA,diff(pred) / diff(newtime))) %>%
ungroup() %>%
as.data.frame()
detach(package:plyr)
z<-z %>% group by(Study ID) %>% mutate(derivative = c(NA,diff(pred))) %>%
ungroup() %>%
as.data.frame()
#manually getting derivative
library(mosaic)
D(124.32941 + 6.58314*cos(2*pi*t/P) + 10.35877*sin(2*pi*t/P) ~ t, P=48) # default values for parameters.
####correct identification of max slope derivative ####
#coefficients
bc<-coef(fm2)
bc<-cbind(Study ID = rownames(bc), bc)</pre>
colnames(bc)[2] <- "b0"
colnames(bc)[3] <- "b1"
colnames(bc)[4] <- "b2"
colnames(bc)[5] <- "b3"
```

```
###one person
```

colnames(bc)[6] <- "b4"

```
t<-rep(28:47)
n<-function(t,p=48){
bc$b0[bc$Study_ID=="LHC0005"]
                                          bc$b1[bc$Study_ID=="LHC0005"]*cos(2*pi*t/p)
                                                                                    +
                                    +
bc$b2[bc$Study_ID=="LHC0005"]*sin(2*pi*t/p) +
 + bc$b3[bc$Study_ID=="LHC0005"]*cos(4*pi*t/p) + bc$b4[bc$Study_ID=="LHC0005"]*sin(4*pi*t/p)
}
max(grad(n, t))
##loop works
t<-seq(28,47,by=1/30)
for(row in 1:nrow(bc)){
n<-function(t,p=48){</pre>
 bc$b0[row]
            + bc$b1[row]*cos(2*pi*t/p) + bc$b2[row]*sin(2*pi*t/p) + bc$b3[row]*cos(4*pi*t/p) +
bc$b4[row]*sin(4*pi*t/p)
}
bc$newvar[row] <-max(grad(n, t))</pre>
}
summary(bc$newvar)
library(refund)
library(refund.shiny)
load("BPM.Rdata")
## Expand grid for all observations from 0 to 47 time points
res <- merge(
expand.grid(newtime=unique(BPM$newtime), Study_ID=unique(BPM$Study_ID)),
BPM, all=TRUE)
BPM=res[order(res$Study_ID),]
BPM$Study_ID<-as.character(BPM$Study_ID)
y=NULL
for(i in 1:length(unique(BPM$Study ID))){
y[[i]]=t(BPM$sbp[which(BPM$Study ID==unique(BPM$Study ID)[i])])
names(y)<-sprintf(unique(BPM$Study_ID),1:length(y))
y=t(sapply(y, '[', 1:max(sapply(y, length))))
n = dim(y)[1]
s = seq(1, 48, length = dim(y)[2])
## do FPCA on the observed functions
fpca.sys = fpca.sc(y)
fpca.sys
par(las=1) #y axis labels horizontal
par(mfrow = c(1,3))
par(mfrow = c(1,4))
par(mfrow = c(2,3))
par(mar=c(5,5,1.5,1.5)+0.1) #remove title space
```

plot FPCA effects for the first basis function

plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", bty="l", pch = 19, lwd = 2, xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", main = paste("1st PC for SBP (", 100*round(fpca.sys\$evalues[1]/sum(fpca.sys\$evalues),3), "%)", sep = ""), cex.axis = 1.5) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[1]) * fpca.sys\$efunctions[,1], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[1]) * fpca.sys\$efunctions[,1], pch = "-") ## plot FPCA effects for the second basis function plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", bty="l", pch = 19, lwd = 2, cex.axis = 1.5, main = paste("2nd PC for SBP (", 100*round(fpca.sys\$evalues[2]/sum(fpca.sys\$evalues),3), "%)", sep = "")) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[2]) * fpca.sys\$efunctions[,2], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[2]) * fpca.sys\$efunctions[,2], pch = "-") ## plot FPCA effects for the 3rd basis function plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", bty="l", pch = 19, lwd = 2, cex.axis = 1.5, main = paste("3rd PC for SBP (", 100*round(fpca.sys\$evalues[3]/sum(fpca.sys\$evalues),3), "%)", sep = "")) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[3]) * fpca.sys\$efunctions[,3], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[3]) * fpca.sys\$efunctions[,3], pch = "-") ## plot FPCA effects for the 4th basis function plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", bty="l", pch = 19, lwd = 2, cex.axis = 1.5, main = paste("4th PC for SBP (", 100*round(fpca.sys\$evalues[4]/sum(fpca.sys\$evalues),3), "%)", sep = "")) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[4]) * fpca.sys\$efunctions[,4], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[4]) * fpca.sys\$efunctions[,4], pch = "-") ## plot FPCA effects for the 5th basis function plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", bty="l", pch = 19, lwd = 2, cex.axis = 1.5, main = paste("5th PC for SBP (", 100*round(fpca.sys\$evalues[5]/sum(fpca.sys\$evalues),3), "%)", sep = "")) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[5]) * fpca.sys\$efunctions[,5], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[5]) * fpca.sys\$efunctions[,5], pch = "-") ## plot FPCA effects for the 6th basis function plot(s, fpca.sys\$mu, type = 'l', ylim = c(90, 160), ylab = "", xaxt = "n",cex.lab=1.5,xlab="Time (24-h clock)", bty="l", pch = 19, lwd = 2, cex.axis = 1.5, main = paste("6th PC for SBP (", 100*round(fpca.sys\$evalues[6]/sum(fpca.sys\$evalues),3), "%)", sep = "")) axis(1, at=c(0,12,23,35,47), labels=c("12.00", "18.00", "00.00", "06.00", "12.00"),cex.axis = 1.5) title(ylab = "SBP (mmHg)", cex.lab = 1.5, line = 3) points(s, fpca.sys\$mu - sqrt(fpca.sys\$evalues[6]) * fpca.sys\$efunctions[,6], pch = "+") points(s, fpca.sys\$mu + sqrt(fpca.sys\$evalues[6]) * fpca.sys\$efunctions[,6], pch = "-") #scree plot

par(mar=c(5,5,1.5,1.5)+0.1) #remove title space par(mfrow = c(1,1)) dd<-data.frame(y=fpca.sys\$evalues,x=1:8) plot(dd\$x, dd\$y, type = 'l', ylab = "", cex.lab=1.5,xlab="Principle Component", bty="l", pch = 21,cex=2,lwd = 2, cex.axis = 1.5) title(ylab = "Eigenvalue", cex.lab = 1.5, line = 3) points(dd\$x, dd\$y, pch = 21,cex=2,bg="white",lwd = 2)

```
## Curve reconstruction for different values of k
plot(y[800,], pch=19, ylab = "mmHg", xlab="Timepoint", main= "Curve reconstruction for subject 800 with K
expansions")
lines(fpca.sys$mu+(fpca.sys$scores[800,1]*fpca.sys$efunctions[,1]+fpca.sys$scores[800,2]), col="blue")
lines(fpca.sys$mu+(fpca.sys$scores[800,1]*fpca.sys$efunctions[,1]+fpca.sys$scores[800,2]*fpca.sys$efunctions
[,2]
```

+fpca.sys\$scores[800,3]*fpca.sys\$efunctions[,3]+fpca.sys\$scores[800,4]),col="green") lines(fpca.sys\$mu+(fpca.sys\$scores[800,1]*fpca.sys\$efunctions[,1]+fpca.sys\$scores[800,2]*fpca.sys\$efunctions [,2]

+fpca.sys\$scores[800,3]*fpca.sys\$efunctions[,3]+fpca.sys\$scores[800,4]*fpca.sys\$efunctions[,4]+fpca.sys\$scores[800,5]*fpca.sys\$efunctions[,5]+fpca.sys\$scores[800,6]*fpca.sys\$efunctions[,6]+fpca.sys\$scores[800,7]*fpca.sys\$efunctions[,7]+fpca.sys\$scores[800,8]*fpca.sys\$efunctions[,8]), col="red")

legend("topright",legend=(c("k=2","K=4","k=8")),col=c("blue","green","red"),lty=1,bty="n",xpd=NA, seg.len = 2)

plot_shiny(fpca.sys)

uy

```
#bring in cosinor 2-component values
a<-ranef(cos1qq)
colnames(a) = c("MESOR","Amplitude 1","Phase 1", "Amplitude 2", "Phase 2")
a<-cbind(Study_ID = rownames(a), a)
a
names(a)
aw<-merge(a,uy,by= "Study_ID")
names(aw)
aw<-select(aw,-Study_ID)</pre>
```

```
#correlation plot
pairs(aw)
panel.cor <- function(x, y, digits=1, prefix="", cex.cor, ...)
{
    usr <- par("usr"); on.exit(par(usr))
    par(usr = c(0, 1, 0, 1))
    r <- abs(cor(x, y))
    txt <- format(c(r, 0.123456789), digits=digits)[1]
    txt <- paste(prefix, txt, sep="")
    if(missing(cex.cor)) cex.cor <- 0.8/strwidth(txt)
    text(0.5, 0.5, txt, cex = 2)
        #, cex = cex.cor * r) #remove size proportional to correlation
}</pre>
```

Appendix I Research output, dissemination and training

| Year | Peer reviewed journal |
|---------|--|
| 2016 | Madden JM, Li X, Tilling K, Kearney PM, Fitzgerald AP. Exploring Diurnal |
| (Under | Variation Using Piecewise Linear Splines: An Example Using Blood |
| Review) | Pressure. (Under review - Emerging Themes in Epidemiology) |
| 2015 | Madden JM, O'Flynn AM, Fitzgerald AP, Kearney PM. Correlation |
| | between short-term blood pressure variability and left-ventricular |
| | mass index: a meta-analysis. Hypertens Res. 2015 39, 171–177; |
| | doi:10.1038/hr.126 |
| 2015 | Madden JM, O'Flynn AM, Dolan E, Fitzgerald AP, Kearney PM. Short- |
| | term blood pressure variability over 24h and target organ damage in |
| | middle-aged men and women. J Hum Hypertens 2015 doi: |
| | 10.1038/jhh.2015.18 |

Table 9-5 Peer reviewed publication from PhD

Table 9-6 Other research output during PhD

| Year | Peer reviewed journal |
|------|--|
| 2015 | O'Flynn AM, Madden JM, Russell AJ, Curtin RJ, Kearney PM. Isolated nocturnal hypertension and subclinical target organ damage: a systematic review of the literature. Hypertens Res. 2015 Apr 2. doi: 10.1038/hr.2015.43. |
| 2015 | O'Flynn AM, McHugh SM, Madden JM, Harrington JM, Perry IJ, Kearney PM. 'Applying the Ideal Cardiovascular Health Metrics to Couples: A Cross-Sectional Study in Primary Care'. Clinical Cardiology, 2015 |
| 2013 | Harrington, J., Fitzgerald, A. P., Kearney, P. M., McCarthy, V. J., Madden, J., Browne, G., Dolan, E. & Perry, I. J. DASH Diet Score and Distribution of Blood Pressure in Middle-Aged Men and Women. Am J Hypertens 2013 Aug 6. |
| 2013 | Buckley, C. M., Madden, J., Balanda, K., Barron, S., Fahy, L., Harrington, J., Perry, I. J. & Kearney, P. M. Pre-diabetes in adults 45 years and over in Ireland: the Survey of Lifestyle, Attitudes and Nutrition in Ireland 2007. Diabet Med. 2013 May 10. doi: 10.1111/dme.12226 |
| 2013 | Curtin, M., Madden, J., Staines, A. & Perry, I. J. Determinants of vulnerability in early childhood development in Ireland: a cross-sectional study BMJ Open 2013;3:e002387 |
| 2013 | Balanda KP, Buckley CM, Barron SJ, Fahy LE, Madden JM, <i>et al.</i> (2013) Prevalence of Diabetes in the Republic of Ireland: Results from the National Health Survey (SLAN) 2007. PLoS ONE 8(10): e78406. doi:10.1371/journal.pone.0078406 |

| 2013 | McKenna, G., Madden, J., Manton, S. & Cronin, M. Survey of Oral |
|------|---|
| | Health Behaviors of Patients Receiving Methadone Therapy at a Drug |
| | Rehabilitation Centre. Journal of Theory and Practice of Dental Public |
| | Health, 2013; Vol 1 No. 3. Pg 6-13 |
| 2013 | Hayes, M., Burke, F. M., McKenna, G., Madden, J. & Cronin, M. An analysis of the attitudes of dental patients attending general dental practice in Galway. Journal of The Irish Dental Association, 2013; 59 (4): 179-182. |

Table 9-7 Conference Presentations during PhD

| Year | Title | Conference |
|----------------------------------|------------------------------------|--|
| 2016 | Blood Pressure Variability over 24 | SPHeRE Network 2 nd Annual |
| hours using Mixed-effects Models | | Conference, RCSI, Dublin, Ireland, |
| | | February 29 th , 2016 |
| 2015 | Short-Term Blood Pressure | International Biometric Society, Eastern |
| | Variability over 24-h Using Mixed- | North America Region, Miami, USA |
| | effects Models | March 15-18, 2015 |
| 2015 | Short-term Blood Pressure | Conference on Applied Statistics in |
| | Variability over 24 hours and | Ireland, UCC, Cork, Ireland, May 11- |
| | Target Organ Damage in Middle- | 13 th , 2015 |
| | Aged Men and Women | |
| 2015 | Short-term Blood Pressure | SPHeRE Network 1 st Annual |
| | Variability over 24 hours and | Conference, RCSI, Dublin, Ireland, |
| | Target Organ Damage in Middle- | January 9 th , 2015 |
| | Aged Men and Women | |
| 2013 | Intensity of Physical Activity and | 3rd International Conference on |
| | Obesity in an Irish Cohort Using | Ambulatory Monitoring of Physical |
| | GENEActiv Accelerometers. | Activity and Movement Amherst, |
| | | Massachusetts, USA June 17-19, 2013 |

| Year | Course | |
|---|---|--|
| 2016 HRB-Trials Methodology Research Network: Statistical Con | | |
| | in Clinical Trial Design, UCC | |
| 2015 | European Society of Hypertension Summer School, Vienna | |
| 2014 | Newcastle R Course (1 week), Colin Gillespie | |
| 2014 | Advanced programming in R: Royal Statistics Society RSS Colin Gillespie | |
| 2014 | Multilevel Modeling using Stata, SPHeRE PhD programme, George Leckie | |
| 2014 | An Introduction to Cochrane Systematic Reviews, Cochrane Collaboration, UCC | |
| 2013 | Introduction to Propensity Score Methods with R, Predictive Modeling | |
| | with R and the caret Package, User R Conference, Albacete, Spain | |
| 2012 | Analysis of Repeated Measures, Centre for Multilevel Modeling, Bristol | |
| 2012 | PG7016 Systematic reviews for the health sciences, UCC | |

Table 9-8 Training & workshops attend during PhD

Table 9-9 Placements Completed

| Year | Supervisor | Award |
|--------|-------------------|---|
| Summer | Dr. John Newell | Biostatistics Unit in the HRB Clinical Research Facility, |
| 2013 | | NUI Galway |
| Summer | Prof Kate Tilling | School of Social and Community Medicine, University |
| 2015 | | of Bristol |

Table 9-10 Awards

| Year | Award |
|------|---|
| 2014 | Travel Bursary: College of Medicine and Health, UCC |

Appendix J Published Papers (PDFs)